UNITED STATES

SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549 FORM 10-O

(Mark One)

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended July 31, 2003

OR

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from to

Commission file number 000-23211

CASELLA WASTE SYSTEMS, INC.

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

03-0338873 (I.R.S. Employer Identification No.)

25 Greens Hill Lane, Rutland, Vermont

(Address of principal executive offices)

05701s (Zip Code)

Registrant's telephone number, including area code: (802) 775-0325

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square

Indicate by check mark whether the registrant is an accelerated filer (as defined in rule 126-2 of the Exchange Act) Yes 🗷 No 🗖

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of August 29, 2003:

Class A Common Stock Class B Common Stock 22,833,982 988,200

PART I. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

CASELLA WASTE SYSTEMS, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS (Unaudited) (in thousands)

	April 30, 2003			July 31, 2003
ASSETS				
CURRENT ASSETS:				
Cash and cash equivalents	\$	15,652	\$	5,737
Restricted cash		10,839		11,606
Accounts receivable - trade, net of allowance for doubtful accounts of \$895 and \$1,048		45,649		50,889
Notes receivable - officers/employees		1,105		1,105
Prepaid expenses		5,906		5,583
Inventory		1,740		1,581
Deferred income taxes		4,275		4,110
Other current assets		1,111		1,078
Total current assets		86,277		81,689
			-	
Property, plant and equipment, net of accumulated depreciation and amortization of \$201,681 and \$224,984		302,328		305,084
Goodwill, net		159,682		162,938
Other intangible assets, net		3,014		3,223
Investments in unconsolidated entities		34,740		35,048
Net assets under contractual obligation		3,844		5,948
Other non-current assets		12,756		13,269
		516,364		525,510
	\$	602,641	\$	607,199

The accompanying notes are an integral part of these consolidated financial statements

CASELLA WASTE SYSTEMS, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS (Unaudited) (in thousands, except for share and per share data)

		April 30, 2003		July 31, 2003
LIABILITIES AND STOCKHOLDERS' EQUITY				
CURRENT LIABILITIES:				
	\$	4.52.4	¢	4 4 4 1
Current maturities of long-term debt	\$	4,534	\$	4,441
Current maturities of capital lease obligations		1,287		1,028
Accounts payable		33,743		34,674
Accrued payroll and related expenses		7,383		5,991
Accrued interest		5,375		8,347
Accrued income taxes		4,526		3,832
Accrued closure and post-closure costs, current portion		2,962		1,750
Other accrued liabilities	. <u> </u>	15,662		16,288
Total current liabilities		75,472		76,351
		202 200		202.027
Long-term debt, less current maturities		302,389		302,037
Capital lease obligations, less current maturities		1,969		1,812
Accrued closure and post-closure costs, less current maturities		22,987		17,727
Deferred income taxes		5,473		8,514
Other long-term liabilities		11,375		11,344
Commitments and contingencies				
Series A redeemable, convertible preferred stock, 55,750 shares authorized, issued and outstanding as of April 30, 2003 and July 31, 2003, liquidation preference of \$1,000 per share plus accrued but unpaid dividends		63,824		64,622
STOCKHOLDERS' EQUITY:				
Class A common stock - Authorized - 100,000,000 shares, \$0.01 par value issued and outstanding - 22,769,000 and 22,797,000 shares		228		228
as of April 30, 2003 and July 31, 2003, respectively Class B Common Stock -		228		228
		10		10
Authorized - 1,000,000 shares, \$0.01 par value 10 votes per share, issued and outstanding - 988,000 shares		10		10
Accumulated other comprehensive income		542		733
Additional paid-in capital		270,068		269,323
Accumulated deficit		(151,696)		(145,502)
Total stockholders' equity		119,152		124,792
	\$	602,641	\$	607,199

The accompanying notes are an integral part of these consolidated financial statements

CASELLA WASTE SYSTEMS, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited) (in thousands)

	Three Months En		nded July 31,	
	 2002		2003	
Revenues	\$ 116,031	\$	113,888	
Operating expenses:				
Cost of operations	77,792		74,278	
General and administration	14,711		14,473	
Depreciation and amortization	 12,061		14,770	
	 104,564		103,521	
Operating income	 11,467		10,367	
Other (income)/expense, net:				
Interest income	(79)		(52)	
Interest expense	7,155		6,275	
Income from equity method investments	(201)		(35)	
Minority interest	(152)			
Other (income)/expense, net	28		(159)	
Other expense, net	 6,751		6,029	
Income from continuing operations before income taxes, discontinued operations and cumulative effect of change in accounting principle	4,716		4,338	
Provision for income taxes	 2,159		867	
Income from continuing operations before discontinued operations and cumulative effect of change in				
accounting principle	2,557		3,471	
Reclassification adjustment from discontinued operations				
(net of income taxes of \$19)	47			
Cumulative effect of change in accounting principle				
(net of income tax (provision) benefit of \$189 and (\$1,856))	 (63,916)		2,723	
Net income (loss)	(61,312)		6,194	
Preferred stock dividend	 759		798	
Net income (loss) available to common stockholders	\$ (62,071)	\$	5,396	

The accompanying notes are an integral part of these consolidated financial statements.

		Three Months E	nded July 31,
		2002	2003
Earnings Per Share:			
Basic:			
Net income before cumulative effect of change in accounting principle	\$	0.08	\$ 0.11
Reclassification adjustment from discontinued operations, net		_	—
Cumulative effect of change in accounting principle, net		(2.70)	0.11
	۴		¢ 0.00
Net income (loss) per common share	\$	(2.62)	\$ 0.22
Basic weighted average common shares outstanding		23,684	23,760
Diluted:			
Net income before cumulative effect of change in accounting principle	\$	0.07	\$ 0.11
Reclassification adjustment from discontinued operations, net		_	_
Cumulative effect of change in accounting principle, net		(2.64)	0.11
Net income (loss) per common share	\$	(2.57)	\$ 0.22
Diluted weighted average common shares outstanding		24,152	24,006

The accompanying notes are an integral part of these consolidated financial statements.

CASELLA WASTE SYSTEMS, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited) (in thousands)

	Three Months Ended		
	 2002	2003	
Cash Flows from Operating Activities:			
Net income (loss)	\$ (61,312)	\$	6,194
Adjustments to reconcile net income (loss) to net cash provided by operating activities -			
Depreciation and amortization	12,061		14,770
Reclassification adjustment from discontinued operations	(47)		
Cumulative effect of change in accounting principle, net	63,916		(2,72)
Income from equity method investments	(201)		(3:
(Gain) loss on sale of assets	4		(54
Minority interest	(152)		_
Deferred income taxes	1,308		1,35
Changes in assets and liabilities, net of effects of acquisitions and divestitures -			
Accounts receivable	(8,768)		(7,440
Accounts payable	7,419		2,252
Other assets and liabilities	(4,320)		67
	71,220		8,79
Net Cash Provided by Operating Activities	 9,908		14,98
Cash Flows from Investing Activities:			
Acquisitions, net of cash acquired			(6,02
Additions to property, plant and equipment	(11,336)	()	17,73
Proceeds from sale of equipment	110		5
(Advances to) distributions from unconsolidated entities	500		(69
Proceeds from assets under contractual obligation	—		30
Net Cash Used In Investing Activities	(10,726)	(2	24,09
Cash Flows from Financing Activities:			
Proceeds from long-term borrowings	21,550		33,40
Principal payments on long-term debt	(24,462)		34,26
Proceeds from exercise of stock options	427	(-	5
Net Cash Used In Financing Activities	 (2,485)		(80
Net decrease in cash and cash equivalents	(3,303)		(9,91)
Cash and cash equivalents, beginning of period	 4,298		15,652
Cash and cash equivalents, end of period	\$ 995	\$	5,73

The accompanying notes are an integral part of these consolidated financial statements.

	Three Months Ended July 3		
	 2001		2002
Supplemental Disclosures of Cash Flow Information:			
Cash paid (received) during the period for -			
Interest	\$ 6,598	\$	2,878
Income taxes, net of refunds	\$ 210	\$	341
Supplemental Disclosures of Non-Cash Investing and Financing Activities:			
Summary of entities acquired in purchase business combinations			
Fair market value of assets acquired	\$ _	\$	6,213
Cash paid, net	_		(6,027)
Liabilities assumed and receivables forgiven to seller	\$ _	\$	186

The accompanying notes are an integral part of these consolidated financial statements.

CASELLA WASTE SYSTEMS, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (Unaudited) (In thousands, except for per share data)

1. ORGANIZATION

The consolidated balance sheets of Casella Waste Systems, Inc. and Subsidiaries (the "Company" or the "Parent") as of April 30, 2003 and July 31, 2003, the consolidated statements of operations for the three months ended July 31, 2002 and 2003 and the consolidated statements of cash flows for the three months ended July 31, 2002 and 2003 are unaudited. In the opinion of management, such financial statements include all adjustments (which include normal recurring and nonrecurring adjustments) necessary for a fair presentation of the financial position, results of operations, and cash flows for the periods presented. The consolidated financial statements presented herein should be read in connection with the Company's audited consolidated financial statements as of and for the twelve months ended April 30, 2003. These were included as part of the Company's Annual Report on Form 10-K for the year ended April 30, 2003 (the "Annual Report"). The results of the three months ended July 31, 2003 may not be indicative of the results that may be expected for the fiscal year ending April 30, 2004.

2. RECLASSIFICATIONS

In the fourth quarter of fiscal 2003, the Company entered into negotiations with former employees for the transfer of our domestic brokerage operation and a commercial recycling business. The commercial recycling business had been accounted for as a discontinued operation since fiscal 2001. Due to the nature of the transaction, the Company could not retain historical discontinued accounting treatment for this operation. Therefore the commercial recycling business' operating results have been reclassified from discontinued to continuing operations for the quarter ended July 31, 2002. Also in connection with the discontinued accounting treatment recorded in fiscal 2001, estimated future losses from this operation were recorded and classified as losses from discontinued operations. This amount has been reclassified and offset against actual losses from operations for the quarter ended July, 31, 2002.

3. BUSINESS COMBINATIONS

During the three months ended July 31, 2003, the Company acquired three solid waste hauling operations in transactions accounted for as purchases. These transactions were in exchange for consideration of \$6,027 in cash to the sellers. The Company completed no such acquisitions during the three months ended July 31, 2002. The operating results of these businesses are included in the consolidated statements of operations from the dates of acquisition. The purchase prices have been allocated to the net assets acquired based on their fair values at the dates of acquisition with the residual amounts allocated to goodwill.

The following unaudited pro forma combined information shows the results of the Company's operations as though each of the acquisitions had been completed as of May 1, 2002.

	Three Mor July 31		 ree Months Ended July 31, 2003
Revenues	\$	117,471	\$ 114,508
Operating income	\$	11,790	\$ 10,510
Net income (loss) available to common stockholders	\$	(61,975)	\$ 5,466
Diluted net income (loss) per common share	\$	(2.57)	\$ 0.23
Diluted weighted average common shares outstanding		24,152	24,006

The foregoing pro forma results have been prepared for comparative purposes only and are not necessarily indicative of the actual results of operations had the acquisitions taken place as of May 1, 2002 or the results of future operations of the Company. Furthermore, such pro forma results do not give effect to all cost savings or incremental costs that may occur as a result of the integration and consolidation of the completed acquisitions.



4. ADOPTION OF NEW ACCOUNTING STANDARDS

Effective May 1, 2003, the Company adopted SFAS No. 143, *Accounting for Asset Retirement Obligations*. SFAS No. 143 does not change the basic accounting principles that the Company and the waste industry has historically followed for accounting for these types of obligations. In general, the Company has followed and will continue the practice of life cycle accounting which recognizes a liability on the balance sheet and related expense as airspace is consumed at the landfill to match operating costs with revenues.

The primary modification to the Company's methodology required by SFAS No. 143 is to require that capping, closure and post-closure costs be discounted to present value. The Company's estimates of future capping, closure and post-closure costs historically have not taken into account discounts for the present value of costs to be paid in the future. Under SFAS No. 143, the Company's estimates of costs to discharge asset retirement obligations for landfills are developed in today's dollars. These costs are then inflated by 2.6% to reflect a normal escalation of prices up to the year they are expected to be paid. These estimated costs are then discounted to their present value using a credit adjusted risk-free rate of 9.5%.

Under SFAS No. 143, the Company no longer accrues landfill retirement obligations through a charge to cost of operations, but rather by an increase to landfill assets. Under SFAS No. 143, the amortizable landfill assets include not only the landfill development costs incurred but also the recorded capping, closure and post-closure liabilities as well as the cost estimates for future capping, closure and post-closure costs. The landfill asset is amortized over the total capacity of the landfill, as airspace is consumed during the life of the landfill with one exception. The exception is for capping for which both the recognition of the liability and the amortization of these costs are based instead on the airspace consumed for the specific capping event.

Upon adoption, SFAS No. 143 required a cumulative change in accounting for landfill obligations retroactive to the date of the inception of the landfill. Inception of the asset retirement obligation is the date operations commenced or the date the asset was acquired. To do this, SFAS No. 143 required the creation of the related landfill asset, net of accumulated amortization and an adjustment to the capping, closure and post-closure liability for cumulative accretion.

At May 1, 2003, the Company recorded a cumulative effect of change in accounting principle of \$2,723 (net of taxes of \$1,856). In addition we recorded a decrease in our capping, closure and post-closure obligations of \$7,807, and a decrease in our net landfill assets of \$3,228. The following is a summary of the balance sheet changes for landfill assets and capping, closure and post-closure liabilities at May 1, 2003 (in thousands):

	nce at April 30, 2003	Change	Balance at May 1, 2003
Landfill assets	\$ 148,029	\$ 6,166	\$ 154,195
Accumulated amortization	(63,207)	(9,394)	(72,601)
Net landfill assets	\$ 84,822	\$ (3,228)	\$ 81,594
Capping, closure, and post-Closure liability	\$ 25,949	\$ (7,807)	\$ 18,142
	9		

The following table shows the activity and total balances related to accruals for capping, closure and post-closure from April 30, 2003 to July 31, 2003 (in thousands):

Balance at April 30, 2003	\$ 25,949
Obligations incurred	1,104
Accretion expense	596
Payments	(365)
Cumulative effect of change in accounting principle	 (7,807)
Balance at June 30, 2003	\$ 19,477

In April 2002, the FASB issued SFAS No. 145, *Rescission of FASB Statements No., 4, 44 and 64, Amendment of FASB Statement No. 13, and Technical Corrections.* SFAS No. 145, among other things, restricts the classification of gains and losses from extinguishment of debt as extraordinary such that most debt extinguishment gains and losses will no longer be classified as extraordinary. The Company adopted SFAS 145 effective May 1, 2003. Under SFAS 145, gains and losses on future debt extinguishment, if any, will be recorded in pre-tax income. In the third quarter of fiscal year 2003, the Company recorded an extraordinary loss of \$2,170 (net of income tax benefit of \$1,479) in connection with the write-off of deferred financing costs related to the Company's old term loan and old revolver. This item will be reclassified to continuing operations in the third quarter of fiscal 2004.

In July 2002, the FASB issued SFAS No. 146, Accounting for Costs Associated with Exit or Disposal Activities. SFAS No. 146 addresses costs such as restructuring, involuntary termination of employees and consolidating facilities but excludes from its scope exit and disposal activities that are in connection with a business combination and those activities to which SFAS No. 143 and No. 144 are applicable. SFAS No. 146 is effective for exit and disposal activities that are initiated after December 31, 2002. The Company has not engaged in or initiated any exit or disposal activities since December 31, 2002.

In November 2002, the FASB issued Interpretation No. 45, *Guarantor's Accounting and Disclosure Requirements for Guarantees, Including Indirect Guarantees of Indebtedness of Others* ("FIN 45"). FIN 45 clarifies the requirements of FASB No. 5, *Accounting for Contingencies*, relating to a guarantor's accounting for, and disclosure of, the issuance of certain types of guarantees. It requires a guarantor to recognize, at the inception of a guarantee, a liability for the fair value of the obligation undertaken in issuing the guarantee. The initial recognition and initial measurement provisions of FIN 45 are effective on a prospective basis to guarantees issued or modified after December 31, 2002. The Company will record the fair value of future material guarantees, if any.

In December, 2002, the FASB issued SFAS No. 148, Accounting for Stock-Based Compensation - Transition and Disclosure - an amendment of FAS 123. This statement amends FASB Statement No. 123, Accounting for Stock-Based Compensation, to provide alternative methods of transition for a voluntary change to the fair value based method of accounting for stock-based employee compensation. In addition, SFAS No. 148 amends the disclosure requirements of SFAS No. 123 to require prominent disclosures in both annual and interim financial statements about the method of accounting for stock-based employee compensation and the effect of the method used in reporting results. SFAS No. 148 is effective for fiscal years ending after December 15, 2002. The Company has included the required disclosures in these financial statements (Note 10).

In January 2003, the FASB issued Interpretation No. 46, *Consolidation of Variable Interest Entities, an Interpretation of APB No. 51* ("FIN 46"). FIN 46 requires that unconsolidated variable interest entities be consolidated by their primary beneficiary who absorbs a majority of the entities' expected losses or residual



benefits. FIN 46 consolidation requirements apply immediately to all variable interest entities created after January 31, 2003 and on June 15, 2003 for those entities already established. The Company has no unconsolidated subsidiaries or affiliates; therefore FIN 46 had no impact on the Company's consolidated financial statements.

5. LEGAL PROCEEDINGS

In the normal course of its business and as a result of the extensive governmental regulation of the waste industry, the Company may periodically become subject to various judicial and administrative proceedings involving Federal, state or local agencies. In these proceedings, an agency may seek to impose fines on the Company or to revoke, or to deny renewal of, an operating permit held by the Company. In addition, the Company may become party to various claims and suits for alleged damages to persons and property, alleged violation of certain laws and for alleged liabilities arising out of matters occurring during the normal operation of the waste management business.

In July 1996, Clinton County, New York entered into a privatization agreement with Casella for Casella to run the County's solid waste management system (the "System") as a private enterprise, including operations at both the existing unlined landfill, as well as newly constructed lined landfill areas. During the period of November 21, 1996 to October 9, 1997, we performed certain closure activities and installed a cut-off wall at the unlined portion of the landfill. On or about April 1999, the New York State Department of Labor alleged that we should have paid prevailing wages in connection with the labor associated with such activities related to the unlined landfill. The DOL is attempting to apply the prevailing wage provisions of Labor Law § 220 to Casella's construction activities at the unlined portion of the Clinton County landfill, to include (1) cap construction at the unlined landfill; (2) construction of the "Casella Barrier Wall," which the New York State Department of Environment Conservation (the "DEC") required as a precondition to permitting the Phase III expansion of the Lined Landfill; and (3) construction of the "County Barrier Wall," which the DEC required as a corrective measure to control the historical contamination. We have disputed the allegations and a hearing on only the liability issue was held on September 16, 2002. Since the hearing did not address damages, relevant payroll documents have not been fully reviewed by either party. Accordingly, neither side is in a position to estimate wage amounts that might be payable in the event the hearing officer finds that Casella is liable for the payment of such prevailing wages. In addition, any such estimate will differ depending on whether any liability ruling applies to some or all of the activities described above; and whether it would apply only to activities of Casella or to all subcontractors as well. In November 2002, both sides submitted proposed findings of fact and conclusions of law. The hearing officer is expected to make a recommendation to the Department of Labor commissioner during the summer or fall of 2003 on the liability issue. We continue to explore settlement possibilities with the State. We believe that we have meritorious defenses to these claims. Although a loss as a result of these claims is reasonably possible, we cannot estimate a range of reasonably possible losses at this time.

The Company is a defendant in certain other lawsuits alleging various claims, none of which, either individually or in the aggregate, the Company believes are material to its financial condition, results of operations or cash flows.

6. ENVIRONMENTAL LIABILITIES

The Company is subject to liability for any environmental damage, including personal injury and property damage, that its solid waste, recycling and power generation facilities may cause to neighboring property owners, particularly as a result of the contamination of drinking water sources or soil, possibly including damage resulting from conditions existing before the Company acquired the facilities. The Company may also be subject to liability for similar claims arising from off-site environmental contamination caused by pollutants or hazardous substances if the Company or its predecessors arrange to transport, treat or dispose of those materials. Any substantial liability incurred by the Company arising from environmental damage could have a material adverse effect on the Company's business, financial condition and results of operations. The Company is not presently aware of any situations that it expects would have a material adverse impact.



7. EARNINGS PER SHARE

The following table sets forth the numerator and denominator used in the computation of earnings per share from continuing operations before discontinued operations and cumulative effect of change in accounting principle on a basic and diluted basis for the three months ended July 31, 2002 and 2003.

		Three Mon July	 led
	20	02	2003
Numerator:			
Income from continuing operations before discontinued operations and cumulative			
effect of change in accounting principle	\$	4,716	\$ 4,338
Less: Preferred dividends		(759)	(798)
Income from continuing operations before discontinued operations and cumulative			
effect of change in accounting principle available to common stockholders	\$	3,957	\$ 3,540
Denominator:			
Number of shares outstanding, end of period:			
Class A common stock		22,722	22,796
Class B common stock		988	988
Effect of weighted average shares outstanding during period		(26)	(24)
Weighted average number of common shares used in basic EPS		23,684	23,760
Impact of potentially dilutive securities:			
Dilutive effect of options, warrants and contingent stock		468	246
Weighted average number of common shares used in diluted EPS		24,152	 24,006

For the three months ended July 31, 2002 and 2003, 7,039 and 7,163 common stock equivalents related to options, warrants, and redeemable convertible preferred stock, respectively, were excluded from the calculation of dilutive shares since the inclusion of such shares would be anti-dilutive.

8. COMPREHENSIVE INCOME (LOSS)

Comprehensive income (loss) represents the change in the Company's equity from transactions and other events and circumstances from non-owner sources and includes all changes in equity except those resulting from investments by owners and distributions to owners. Comprehensive income (loss) for the three months ended July 31, 2002 and 2003 is as follows:

	Three Mon July		led
	 2002	,	2003
Net income (loss)	\$ (61,312)	\$	6,194
Other comprehensive income	426		191
Comprehensive income (loss)	\$ (60,886)	\$	6,385

The components of other comprehensive income (loss) for the three months ended July 31, 2002 and 2003 are shown as follows:

				onths Ended 31, 2002		
	(Gross	Ta	x effect	N	et of Tax
Changes in fair value of marketable securities during the						
period, net	\$	(37)	\$		\$	(37)
Change in fair value of interest rate swaps and commodity						
hedges during period, net		779		316		463
	\$	742	\$	316	\$	426
				onths Ended 31, 2003		
	(Gross	Ta	x effect	N	et of Tax
Change in fair value of interest rate swaps and commodity						
hedges during period, net	\$	322	\$	131	\$	191
	\$	322	\$	131	\$	191

9. DERIVATIVE INSTRUMENTS AND HEDGING ACTIVITIES

The Company's strategy to hedge against fluctuations in the commodity prices of recycled paper is to enter into hedges to mitigate the variability in cash flows generated from the sales of recycled paper at floating prices, resulting in a fixed price being received from these sales. The Company is party to twelve commodity hedge contracts as of July 31, 2003. These contracts expire between August 2003 and November 2005. The Company has evaluated these hedges and believes that these instruments qualify for hedge accounting pursuant to SFAS No. 133, *Accounting for Derivative Instruments and Hedging Activities*. As of July 31, 2003 the fair value of these hedges was an obligation of \$363, with the net amount (net of taxes of \$147) recorded as an unrealized loss in accumulated other comprehensive income.

The Company is party to two interest swap agreements as of July 31, 2003 for an aggregate notional amount of \$53,000 expiring in February, 2004. The Company has evaluated these swaps and believes these instruments qualify for hedge accounting pursuant to SFAS No. 133. As of July 31, 2003, the fair value of these swaps was a receivable of \$39, with the net amount (net of taxes of \$16) recorded as an unrealized gain in other comprehensive income. The estimated net amount of the existing losses as of July 31, 2003 included in accumulated other comprehensive income expected to be reclassified into earnings as payments are either made or received under the terms of the interest rate swaps within the next 12 months is approximately \$15. The actual amounts reclassified into earnings are dependent on future movements in interest rates.

10. STOCK BASED COMPENSATION PLANS

The Company has elected to account for its stock-based compensation plans under APB Opinion No. 25, *Accounting for Stock Issued to Employees*, for which no compensation expense is recorded in the statements of operations for the estimated fair value of stock options issued with an exercise price equal to the fair value of the underlying common stock on the grant date.

SFAS No. 123, Accounting for Stock-Based Compensation, as amended by SFAS No. 148, Accounting for Stock-Based Compensation - Transition and Disclosure - an amendment of FAS 123, requires that entities electing to remain with the accounting in APB Opinion No. 25 disclose pro forma net income and earnings per share as if the fair value based method of accounting defined in SFAS No. 123 had been applied.

If the Company applied the recognition provisions of SFAS 123 using the Black-Scholes option pricing model, the resulting pro forma net income available to common stockholders per share would be as follows:

	 Three Months Ended July 31,						
	2002		2003				
Net income (loss) available to common stockholders, as reported	\$ (62,071)	\$	5,396				
Deduct: Total stock-based compensation expense determined under fair value based							
method, net	273		131				
Net income (loss) available to common stockholders, pro forma	\$ (62,344)	\$	5,265				
Basic income (loss) per common share:							
As reported	\$ (2.62)	\$	0.22				
Pro forma	\$ (2.63)	\$	0.22				
Diluted income (loss) per common share:							
As reported	\$ (2.57)	\$	0.22				
Pro forma	\$ (2.57)	\$	0.22				

In accordance with SFAS 123, the fair value of each option grant has been estimated as the date of grant using the Black-Scholes option pricing model with the following weighted average assumptions:

	Three Months Ended July 31, 2002
Risk free interest rate	3.72% - 4.07%
Expected dividend yield	N/A
Expected life	5 Years
Expected volatility	65.00%

The Company granted no stock options during the three months ended July 31, 2003. The Company has recorded no compensation expense for stock options granted to employees during the three months ended July 31, 2002 or 2003.

11. SEGMENT REPORTING

SFAS No. 131, *Disclosures about Segments of an Enterprise and Related Information*, establishes standards for reporting information about operating segments in financial statements. In general, SFAS No. 131 requires that business entities report selected information about operating segments in a manner consistent with that used for internal management reporting.

The Company classifies its operations into Eastern, Central, Western and FCR Recycling. The Company's revenues in the Eastern, Central and Western segments are derived mainly from one industry segment, which includes the collection, transfer, recycling and disposal of non-hazardous solid waste. The Eastern Region also includes Maine Energy, which generates electricity from non-hazardous solid waste. The Company's

revenues in the FCR Recycling segment are derived from integrated waste handling services, including processing and recycling of wood, paper, metals, aluminum, plastics and glass and brokerage of recycled materials. In September 2002, the Company transferred the export brokerage operation and in June 2003 the Company transferred its domestic brokerage operation and a commercial recycling business to two groups of employees who had managed those businesses. Included in Other are ancillary operations, mainly major customer accounts, earnings from equity method investees and in the quarter ended July 31, 2002, residue recycling operations.

		Eastern Region		Central Region		Western Region		Recycling		Other
Three Months Ended July 31, 2002										
Outside Revenues	\$	40,014	\$	24,472	\$	17,324	\$	30,477	\$	3,744
Inter-segment Revenues		10,143		11,979		3,894		7,035		
Income (loss) from continuing operations before discontinued operations and cumulative effect of change in accounting principle		(459)		5,515		1,072		(240)		(3,331)
Total Assets	\$	219,689	\$	109,366	\$	105,629	\$	70,438	\$	58,333
Total Assets	φ	219,009	φ		φ	105,027	φ	70,430	ψ	56,555
		Eliminations		Total						
Three Months Ended July 31, 2002										
Outside Revenues	\$	_	- \$	116,031						
Inter-segment Revenues		(33,05)	l)							
Income (loss) from continuing operations before discontinued operations and cumulative effect of				0.555						
change in accounting principle	٩		-	2,557						
Total Assets	\$	_	- \$	563,455						
		Eastern Region		Central Region		Western Region		Recycling		Other
Three Months Ended July 31, 2003								Recycling		Other
Three Months Ended July 31, 2003 Outside Revenues	\$	Region	\$		\$		\$	Recycling 19,497	\$	Other 4,161
• • •	\$	Region	\$	Region	\$	Region	\$		\$	
Outside Revenues Inter-segment Revenues Income (loss) from continuing operations before cumulative effect of change in accounting	\$	Region 43,656 13,232	\$	Region 25,970 12,551	\$	Region 20,604 3,370	\$	19,497 3,195	\$	4,161
Outside Revenues Inter-segment Revenues Income (loss) from continuing operations before cumulative effect of change in accounting principle		Region 43,656 13,232 (1,369)		Region 25,970 12,551 5,441		Region 20,604 3,370 981		19,497 3,195 400		4,161
Outside Revenues Inter-segment Revenues Income (loss) from continuing operations before cumulative effect of change in accounting	\$	Region 43,656 13,232	\$	Region 25,970 12,551	\$	Region 20,604 3,370	\$	19,497 3,195	\$	4,161
Outside Revenues Inter-segment Revenues Income (loss) from continuing operations before cumulative effect of change in accounting principle	\$	Region 43,656 13,232 (1,369)		Region 25,970 12,551 5,441		Region 20,604 3,370 981		19,497 3,195 400		4,161
Outside Revenues Inter-segment Revenues Income (loss) from continuing operations before cumulative effect of change in accounting principle	\$	Region 43,656 13,232 (1,369) 244,434		Region 25,970 12,551 5,441 114,453		Region 20,604 3,370 981		19,497 3,195 400		4,161
Outside Revenues Inter-segment Revenues Income (loss) from continuing operations before cumulative effect of change in accounting principle Total Assets Three Months Ended July 31, 2003	\$ E	Region 43,656 13,232 (1,369) 244,434 liminations	\$	Region 25,970 12,551 5,441 114,453 Total		Region 20,604 3,370 981		19,497 3,195 400		4,161
Outside Revenues Inter-segment Revenues Income (loss) from continuing operations before cumulative effect of change in accounting principle Total Assets Three Months Ended July 31, 2003 Outside Revenues	\$	Region 43,656 13,232 (1,369) 244,434 liminations		Region 25,970 12,551 5,441 114,453		Region 20,604 3,370 981		19,497 3,195 400		4,161
Outside Revenues Inter-segment Revenues Income (loss) from continuing operations before cumulative effect of change in accounting principle Total Assets Three Months Ended July 31, 2003 Outside Revenues Inter-segment Revenues	\$ E	Region 43,656 13,232 (1,369) 244,434 liminations	\$	Region 25,970 12,551 5,441 114,453 Total		Region 20,604 3,370 981		19,497 3,195 400		4,161
Outside Revenues Inter-segment Revenues Income (loss) from continuing operations before cumulative effect of change in accounting principle Total Assets Three Months Ended July 31, 2003 Outside Revenues Inter-segment Revenues Income (loss) from continuing operations before cumulative effect of change in accounting	\$ E	Region 43,656 13,232 (1,369) 244,434 liminations	\$	Region 25,970 12,551 5,441 114,453 Total 113,888 —		Region 20,604 3,370 981		19,497 3,195 400		4,161
Outside Revenues Inter-segment Revenues Income (loss) from continuing operations before cumulative effect of change in accounting principle Total Assets Three Months Ended July 31, 2003 Outside Revenues Inter-segment Revenues Income (loss) from continuing operations before	\$ E	Region 43,656 13,232 (1,369) 244,434 liminations	\$	Region 25,970 12,551 5,441 114,453 Total		Region 20,604 3,370 981		19,497 3,195 400		4,161

Amounts of the Company's total revenue attributable to services provided are as follows:

	Three Months Ended July 31,								
		2002		2003					
Collection	\$	51,267	\$	55,553					
Landfill / disposal facilities		15,607		17,827					
Transfer		13,142		14,247					
Recycling		17,501		22,967					
Brokerage		18,417		3,294					
Other		97							
Reported revenues	\$	116,031	\$	113,888					
			16						

12. NET ASSETS UNDER CONTRACTUAL OBLIGATION

Effective September 30, 2002, the Company transferred its export brokerage operations to former employees, who had been responsible for managing that business. Consideration for the transaction was in the form of two notes receivable amounting up to \$5,460. These notes are payable within five years of the anniversary date of the transaction to the extent of free cash flow generated from the operations.

Effective June 30, 2003, the Company entered into a similar transaction transferring its domestic brokerage operations as well as a commercial recycling business to former employees who had been responsible for managing those businesses. Consideration for the transaction was in the form of two notes receivable amounting up to \$6,925. These notes are payable within twelve years of the anniversary date of the transaction to the extent of free cash flow generated from the operations.

The Company has not accounted for either of these transactions as a sale based on an assessment that the risks and other incidents of ownership have not sufficiently transferred to the buyer. The net assets of the operations are disclosed in the balance sheet as "net assets under contractual obligation", and will be reduced as payments are made.

13. NEW ACCOUNTING PRONOUNCEMENTS

In May 2003, the FASB issued SFAS No. 150, Accounting for Certain Financial Instruments with Characteristics of Both Liability and Equity. The statement changes the accounting for certain financial instruments that, under previous guidance, issuers could account for as equity. The new statement requires that those instruments be classified as liabilities in statements of financial position. SFAS No. 150 is effective for all financial instruments entered into or modified after May 31, 2003, and otherwise is effective at the beginning of the first interim period beginning after June 15, 2003. Management believes that adoption will have no effect on the Company's results of operations and financial position as well as related disclosures.

14. CONDENSED CONSOLIDATING FINANCIAL INFORMATION

The senior subordinated notes are guaranteed jointly and severally, fully and unconditionally by the Company's significant wholly-owned subsidiaries. The Parent is the issuer and non-guarantor of the senior subordinated notes. The information which follows presents the condensed consolidating financial position as of April 30, 2003 and July 31, 2003; the condensed consolidating results of operations for the three months ended July 31, 2002 and 2003; and the condensed consolidating statements of cash flows for the three months ended July 31, 2002 and 2003 of (a) the parent company only ("the Parent"), (b) the combined guarantors ("the Guarantors"), each of which is 100% wholly-owned by the Parent, (c) the combined non-guarantors ("the Non-Guarantors"), (d) eliminating entries and (e) the Company on a consolidated basis.

CASELLA WASTE SYSTEMS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATING BALANCE SHEET AS OF APRIL 30, 2003 (Unaudited) (In thousands)

		Parent	G	uarantors		Non-Guarantors]	Elimination	Co	nsolidated
ASSETS										
CURRENT ASSETS:										
Cash and cash equivalents	\$	12,188	\$	2,686	\$	778	\$	—	\$	15,652
Accounts receivable - trade, net of allowance for										
doubtful accounts		485		44,155		1,009		_		45,649
Prepaid expenses		613		5,138		155		—		5,906
Inventory				1,740				—		1,740
Deferred taxes		3,504				771		—		4,275
Other current assets		1,237		1,103		10,715		_		13,055
Total current assets		18,027		54,822		13,428		—		86,277
Property, plant and equipment, net of accumulated										
depreciation and amortization		2,996		294,109		5,223		—		302,328
Intangible assets, net				162,696						162,696
Deferred income taxes		—		—		—		—		—
Investment in subsidiaries		(43,783)						43,783		_
Investments in unconsolidated entities		7,778		31,341				(4,379)		34,740
Assets under contractual obligation				3,844				—		3,844
Other non-current assets		11,046		1,238		472		_		12,756
		(21,963)		493,228		5,695		39,404		516,364
			_							
Terterer		507.920		(500.997)		(2.212)		4 2 7 0		
Intercompany receivable		507,820		(509,887)		(2,312)	-	4,379		
	\$	503,884	\$	38,163	\$	16,811	\$	43,783	\$	602,641
	-		_		-		-			
			17							

	Parent	Guarantors	Non - Guarantors	Elimination	Consolidated
LIABILITIES AND STOCKHOLDERS' EQUITY	1 arent	Guarantors	Guarantors		Consondated
CURRENT LIABILITIES:					
Current maturities of long term debt	\$ 1,500	\$ 1,777	\$ 1,257	\$ —	\$ 4,534
Accounts payable	1,350	32,285	108	_	33,743
Accrued payroll and related expenses	1,368	6,015	_	—	7,383
Accrued interest	5,373	2		—	5,375
Accrued closure and post-closure costs, current portion	_	2,286	676	_	2,962
Other current liabilities	7,203	5,617	8,655	_	21,475
	.,		- ,		,
Total current liabilities	16,794	47,982	10,696		75,472
Long-term debt, less current maturities	298,500	2,318	1,571		302,389
Capital lease obligations, less current maturities	298,500	1,828	1,371	_	1,969
Accrued closure and post closure costs, less current	141	1,020			1,909
portion		21,977	1,010		22,987
Minority interest			1,010		
Deferred income taxes	5,473				5,473
Other long-term liabilities		10,047	1.328		11,375
ç		,	-,-=+		,
COMMITMENTS AND CONTINGENCIES					
Series A redeemable, convertible preferred stock, 55,750 shares authorized, issued and outstanding, liquidation preference of \$1,000 per share plus accrued but unpaid dividends	63,824	_	_	_	63,824
STOCKHOLDERS' EQUITY:					
Class A common stock -					
Authorized - 100,000,000 shares, \$0.01 par value	228	101	100	(201)	229
issued and outstanding - 22,769,000 shares Class B common stock -	228	101	100	(201)	228
Authorized - 1,000,000 shares, \$0.01 par value 10 votes per share, issued and outstanding - 988,000					
shares	10				10
Accumulated other comprehensive income (loss)	542	1,190		(1,190)	542
Additional paid-in capital	270,068	47,885	2.825	(50,710)	270,068
Accumulated deficit	(151,696)	/)		(151,696)
	(101,000)	();;;;05)	((1))		(151,050)
Total stockholders' equity	119,152	(45,989)	2,206	43,783	119,152
	\$ 503,884	\$ 38,163	\$ 16,811	\$ 43,783	\$ 602,641
	\$ 200,004	\$ 50,105	÷ 10,011	¢ 15,765	¢ 002,041
		18			

CASELLA WASTE SYSTEMS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATING BALANCE SHEET AS OF July 31, 2003 (Unaudited) (In thousands)

		Parent		Guarantors	_	Non-Guarantors	_	Elimination	С	onsolidated
ASSETS CURRENT ASSETS:										
CORRENT ASSETS: Cash and cash equivalents	\$	3,175	\$	2,157	\$	405	\$		\$	5,737
Accounts receivable-trade, net of allowance for	φ	5,175	φ	2,137	φ	405	φ		φ	5,151
doubtful accounts		1,118		48,370		1,401				50,889
Other current assets		5,223		7,349		12,491				25,063
Total current assets		9,516	_	57,876		14,297				81,689
Property, plant and equipment, net of accumulated		2 (01		200.005		1.000				205.004
depreciation and amortization		2,691		298,095		4,298		_		305,084
Intangible assets, net Investment in subsidiaries		(37,161)		162,938				37,161		162,938
Assets under contractual obligation		(37,101)		5,948				57,101		5,948
Other non-current assets		18,937		36,486		496		(4,379)		51,540
		10,957		50,100		190		(1,577)		51,510
		(15,533)		503,467		4,794		32,782	_	525,510
Intercompany receivable		519,821		(522,608)	_	(1,592)		4,379		
	\$	513,804	\$	38,735	\$	17,499	\$	37,161	\$	607,199
						Non -				
		Parent		Guarantors		Guarantors		Elimination	C	onsolidated
LIABILITIES AND STOCKHOLDERS' EQUITY										
CURRENT LIABILITIES:	¢	1.70.6	¢	22.654	Φ.	102	0		¢	24.552
Accounts Payable Accrued interest	\$	1,796 8,345	\$	32,654 2	\$	103	\$	—	\$	34,553 8,347
Accrued income taxes		8,343 1,997		1,649		186				3,832
Accrued closure and post-closure costs, current		1,777		1,047		100				5,652
portion				1,001		870				1,871
Other current liabilities		5,135		11,928		10,685		_		27,748
Total current liabilities		17,273		47,234		11,844				76,351
Long-term debt, less current maturities		298,500		2,274		1,263		_		302,037
Deferred income taxes		8,514						_		8,514
Other long-term liabilities		103		28,498		2,282		—		30,883
COMMITMENTS AND CONTINGENCIES										
Saving A radio making convertible professional stack										
Series A redeemable, convertible preferred stock, 55,750 shares authorized, issued and outstanding,										
liquidation preference of \$1,000 per share plus										
accrued but unpaid dividends		64,622		_		_		_		64,622
		,								,
STOCKHOLDERS' EQUITY:										
Class A common stock -										
Authorized - 100,000,000 shares, \$0.01 par value		228		101		100		(201)		220
issued and outstanding - 22,769,000 shares Class B common stock -		228		101		100		(201)		228
Authorized - 1,000,000 shares, \$0.01 par value 10										
votes per share, issued and outstanding - 988,000										
shares		10		_		_		_		10
Accumulated other comprehensive income (loss)		733		770				(770)		733
Additional paid-in capital		269,323		46,932		2,825		(49,757)		269,323
Accumulated deficit		(145,502)		(87,074)		(815)		87,889		(145,502)
			_							
Total stockholders' equity		124,792	_	(39,271)	_	2,110	_	37,161		124,792
	\$	513,804	\$	38,735	\$	17,499	\$	37,161	\$	607,199
			1	9						

CASELLA WASTE SYSTEMS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATING STATEMENTS OF OPERATIONS THREE MONTHS ENDED JULY 31, 2002 (Unaudited) (In thousands)

		Parent	_(Guarantors		Non - Guarantors		Elimination	 Consolidated
Revenues	\$	—	\$	115,235	\$	2,320	\$	(1,524)	\$ 116,031
Operating expenses:									
Cost of operations		918		75,769		2,629		(1,524)	77,792
General and administration		(156)		14,696		171		_	14,711
Depreciation and amortization		428		11,244		389			12,061
•	-	1,190		101,709		3,189	_	(1,524)	104,564
Operating income (loss)		(1,190)		13,526		(869)	_		 11,467
Other expense/(income), net:									
Interest income		(7,036)		(354)		(48)		7,359	(79)
Interest expense		7,024		7,380		110		(7,359)	7,155
(Income) loss from equity method investments		57.954		(201)		110		(57,954)	(201)
Minority interest		57,954		(201)		(152)		(37,934)	(152)
Other expense/(income), net:		65		(38)		(152)			28
Other expense, net		58,007		6,787	_	(89)		(57,954)	 6,751
Sther expense, her		56,007		0,707	-	(0)		(37,734)	 0,751
Income (loss) from continuing operations before income taxes, discontinued operations, extraordinary item and cumulative effect of change in accounting principle		(59,197)		6.739		(780)		57,954	4,716
Provision for income taxes		2,115		1		43			2,159
									 ,
Income (loss) from continuing operations before discontinued operations and cumulative effect of		((1.212)		(72)		(122)		57.054	0.657
change in accounting principle		(61,312)		6,738		(823)		57,954	2,557
Reclassification adjustment from discontinued operations, net				47					47
Cumulative effect of change in accounting principle,				.,					.,
net		_		(63,916)		_		_	(63,916)
Net loss		(61,312)		(57,131)		(823)	_	57,954	 (61,312)
Preferred stock dividend		759				(759
Net loss available to common stockholders	\$	(62,071)	\$	(57,131)	\$	(823)	\$	57,954	\$ (62,071)
			20						

CASELLA WASTE SYSTEMS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATING STATEMENTS OF OPERATIONS THREE MONTHS ENDED JULY 31, 2003 (Unaudited) (In thousands)

	Parent	Guarantors	Non - Guarantors	Elimination	Consolidated
Revenues	\$ —	\$ 112,146	\$ 4,222	\$ (2,480)	\$ 113,888
Operating expenses:					
Cost of operations	204	73,740	2,814	(2,480)	74,278
General and administration	(118)	14,394	197	—	14,473
Depreciation and amortization	449	12,830	1,491	—	14,770
	535	100,964	4,502	(2,480)	103,521
Operating income (loss)	(535)	11,182	(280)		10,367
Other expense/(income), net:					
Interest income	(5,927)	(337)	(26)	6,238	(52)
Interest expense	6,337	6,107	69	(6,238)	6,275
(Income) loss from equity method investments	(7,995)	(35)	_	7,995	(35)
Other expense/(income), net:	(4)	(126)	(29)		(159)
Other expense, net	(7,589)	5,609	14	7,995	6,029
Income (loss) from continuing operations before income taxes, discontinued operations, extraordinary item and cumulative effect of change in accounting principle	7.054	5.573	(294)	(7,995)	4,338
Provision for income taxes	860	5,575	(2)4)	(1,555)	867
			/		
Income (loss) from continuing operations before discontinued operations, extraordinary item and cumulative effect of change in accounting					
principle	6,194	5,573	(301)	(7,995)	3,471
Cumulative effect of change in accounting principle, net		2,518	205		2,723
Net loss	6,194	8,091	(96)	(7,995)	6,194
Preferred stock dividend	798	0,091	(90)	(7,993)	798
Net loss available to common stockholders	\$ 5,396	\$ 8,091	\$ (96)	\$ (7,995)	\$ 5,396
net loss available to common stockholders	φ <i>5</i> ,590	φ <u>0,091</u>	<u>а (90)</u>	<u>а (7,993</u>)	φ <u>5,590</u>
		21			

CASELLA WASTE SYSTEMS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATING STATEMENTS OF CASH FLOWS THREE MONTHS ENDED JULY 31, 2002 (Unaudited) (In thousands)

	Parent		Guarantors	Non-Guarantors	Elimination		Consolidated
Net Cash Provided by (Used in) Operating Activities	\$ (3,490)	\$	13,388	\$ 10	\$ 	\$	9,908
Cash Flows from Investing Activities:							
Additions to property, plant and equipment	(38)		(11,288)	(10)	—		(11,336)
Other			610		 		610
Net Cash Used In Investing Activities	 (38)		(10,678)	(10)	 		(10,726)
Cash Flows from Financing Activities:							
Proceeds from long-term borrowings	21,550		—	—	—		21,550
Principal payments on long-term debt	(23,421)		(871)	(170)	—		(24,462)
Proceeds from exercise of stock options	427		—	—	—		427
Intercompany borrowings	 2,735		(2,626)	(109)	 		_
Net Cash Provided by (Used In) Financing Activities	1,291		(3,497)	(279)	—		(2,485)
Net (decrease) increase in cash and cash		_				_	
equivalents	(2,237)		(787)	(279)	—		(3,303)
Cash and cash equivalents, beginning of period	4,362		(2,377)	2,313	—		4,298
Cash and cash equivalents, end of period	\$ 2,125	\$	(3,164)	\$ 2,034	\$ —	\$	995
- · ·							
		22					

CASELLA WASTE SYSTEMS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATING STATEMENTS OF CASH FLOWS THREE MONTHS ENDED JULY 31, 2003 (Unaudited) (In thousands)

	 Parent		Guarantors	Non-Guarantors		Elimination		Consolidated
Net Cash Provided by (Used in) Operating Activities	\$ 2,010	\$	11,515	\$ 1,043	\$	420	\$	14,988
Cash Flows from Investing Activities:								
Acquisitions, net of cash acquired	_		(6,027)	—		—		(6,027)
Additions to property, plant and equipment	(144)		(17,028)	(566)		—		(17,738)
Other	(693)		363					(330)
Net Cash Used In Investing Activities	(837)		(22,692)	(566)		_		(24,095)
Cash Flows from Financing Activities:								
Proceeds from long-term borrowings	33,400							33,400
Principal payments on long-term debt	(33,689)		(262)	(310)		—		(34,261)
Proceeds from exercise of stock options	53		_	_		—		53
Intercompany borrowings	(9,950)		10,910	(540)		(420)		
Net Cash Provided by (Used In) Financing Activities	(10,186)		10,648	(850)	_	(420)		(808)
Net (decrease) increase in cash and cash								
equivalents	(9,013)		(529)	(373)		—		(9,915)
Cash and cash equivalents, beginning of period	12,188		2,686	778				15,652
Cash and cash equivalents, end of period	\$ 3,175	\$	2,157	\$ 405	\$		\$	5,737
		23					_	
		23						

15. SUBSEQUENT EVENTS

On August 26, 2003, the Company amended the terms of its Senior Secured Term "B" Loan, lowering the borrowing rate and modifying the prepayment provisions to include a prepayment premium for the first two years following the date of the amendment.

On August 28, 2003, the Company announced that the McKean County, Pennsylvania Solid Waste Authority voted unanimously to negotiate a twenty-year service agreement with the Company to operate and develop the county's Subtitle D landfill. The McKean County landfill is permitted to accept 1,000 tons of municipal solid waste per day and currently accepts 200 tons per day.

On September 5, 2003, the Company announced that the Solid Waste Committee of the Ontario County, New York Board of Supervisors voted to recommend that the county Board of Supervisors select the Company to be the county's twenty-five year partner to operate and develop the county's Subtitle D landfill. The Ontario County landfill is currently permitted to accept 2,000 tons of municipal solid waste per day.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Casella Waste Systems, Inc. and Subsidiaries (the "Company") is a vertically integrated regional solid waste services company that provides collection, transfer, disposal and recycling services to residential, industrial and commercial customers, primarily throughout the eastern region of the United States. As of August 29, 2003, the Company owned and/or operated five Subtitle D landfills, two landfill permitted to accept construction and demolition materials, 37 solid waste collection operations, 33 transfer stations, 36 recycling facilities and one waste-to-energy facility, as well as a 50% interest in a joint venture that manufactures, markets and sells cellulose insulation made from recycled fiber. In June 2003 the Company entered into a service agreement with the Town of Templeton, Massachusetts to construct and operate the town's sanitary landfill. The landfill is expected to be permitted within a year to accept 500 tons a day of municipal solid waste and operations will likely commence in the middle of calendar 2004.

The Company's revenues decreased from \$116.0 million for the three months ended July 31, 2002 to \$113.9 million for the three months ended July 31, 2003. From May 1, 2002 through April 30, 2003, the Company acquired eight solid waste collection, transfer, disposal and recycling operations all of which were accounted for under the purchase method of accounting for business combinations. Between May 1, 2003 and July 31, 2003 the Company acquired three such businesses. Under the rules of purchase accounting, the acquired companies' revenues and results of operations have been included together with those of the Company from the actual dates of the acquisitions and affect the period-to-period comparisons of the Company's historical results of operations. Effective September 30, 2002, the Company transferred its export brokerage operations to former employees, who had been responsible for managing that business. The domestic brokerage operations, and a recycling business, constituting the remainder of the Company's brokerage revenues, were transferred effective June 30, 2003 to the employees of that unit. For the three months ended July 31, 2003 and 2002, the brokerage and recycling businesses accounted for \$3.3 million and \$18.3 million, respectively, of the Company's revenues.

Forward Looking Statements

This Form 10-Q and other reports, proxy statements, and other communications to stockholders, as well as oral statements by the Company's officers or its agents, may contain forward-looking statements within the meaning of Section 27A of the Securities Act and Section 21E of the Securities Exchange Act, with respect to, among other things, the Company's future revenues, operating income, or earnings per share. Without limiting the foregoing, any statements contained in this Form 10-Q that are not statements of historical fact may be deemed to be forward-looking statements, and the words "believes", "anticipates", "plans", "expects", and similar expressions are intended to identify forward-looking statements. There are a number of important factors of which the Company is aware that may cause the Company's actual results to vary materially from those forecasted or projected in any such forward-looking statement, certain of which are beyond the Company's control. These factors include, without limitation, those outlined below in the section entitled "Certain Factors That May Affect Future Results". The Company's failure to successfully address any of these factors could have a material adverse effect on the Company's results of operations.

General

Revenues

The Company's revenues in the Eastern, Central and Western regions are attributable primarily to fees charged to customers for solid waste disposal and collection, landfill, waste-to-energy, transfer and recycling services. The Company derives a substantial portion of its collection revenues from commercial, industrial and municipal services that are generally performed under service agreements or pursuant to contracts with municipalities. The majority of residential collection services are performed on a subscription basis with individual households. Landfill, waste-to-energy facility and transfer customers are charged a tipping fee on a per ton basis for disposing of their solid waste at disposal facilities and transfer stations. The majority of the



Company's disposal and transfer customers are under one to ten year disposal contracts, with most having clauses for annual cost of living increases. Recycling revenues, which are included in FCR Recycling and in the Eastern, Central and Western regions, consist of revenues from the sale of recyclable commodities and operations and maintenance contracts of recycling facilities for municipal customers. FCR Recycling revenues include revenues from commercial brokerage and recycling operations through June 30, 2003, when those operations were sold.

Effective August 1, 2000, the Company contributed its cellulose insulation assets to a joint venture with Louisiana-Pacific, and accordingly, since that date has recognized half of the joint venture's net income/(loss) on the equity method in its results of operations. Also in the "Other" segment, the Company has ancillary revenues including major customer accounts, earnings from equity method investors and in the quarter ended July 31, 2002, revenues from residue recycling.

The Company's revenues are shown net of inter-company eliminations. The Company typically establishes its inter-company transfer pricing based upon prevailing market rates. The table below shows, for the periods indicated, the percentage of the Company's revenues attributable to services provided. For the three months ended July 31, 2003 the percentages of revenues shown below reflect revenues from the commercial brokerage and recycling operations through June 30, 2003 and no revenues from the export brokerage business, sold in September 2002. The reduction in these revenues caused the percentages of the remaining operations to increase. Collection revenues increased as a percentage of total revenues for the quarter ended July 31, 2003 compared to the prior year comparable period due to the effects of volume and price increases. The increase in landfill/disposal facilities revenue in the quarter ended July 31, 2003 is mainly due to an increase in transfer volumes. The increase in recycling revenues as a percentage of total revenues in the quarter ended July 31, 2003 increased as a percentage of revenues as a percentage of total revenues in the quarter ended July 31, 2003 increased as a percentage of revenues as a percentage of total revenues in the quarter ended July 31, 2003 increased as a percentage of revenues as a percentage of revenues in the quarter ended July 31, 2003 increased as a percentage of revenues as a percentage of total revenues in the quarter ended July 31, 2003 compared to the prior year commodity prices. The decrease in brokerage revenues as percentage of total revenues in the quarter ended July 31, 2003 compared to the prior year comparable period is due to lower commodity prices and volumes as well as the transfer of the export business to the employees of that unit in September 2002. The domestic brokerage operations, constituting the remainder of the Company's brokerage revenues were transferred effective June 30, 2003 to the employees of that unit. The decrease in the Company's other re

	Three Months July 31,	
	2002	2003
Collection	44.2%	48.8%
Landfill/disposal facilities	13.5	15.7
Transfer	11.3	12.5
Recycling	15.1	20.2
Brokerage	15.8	2.8
Other	0.1	0.0
Total revenues	100.0%	100.0%

Operating Expenses

Cost of operations includes labor, tipping fees paid to third party disposal facilities, fuel, maintenance and repair of vehicles and equipment, worker's compensation and vehicle insurance, the cost of purchasing materials to be recycled, third party transportation expense, district and state taxes, host community fees and royalties. Cost of operations also includes closure and post-closure amortization and accretion expense, leachate treatment and disposal costs.

General and administration expenses include management, clerical and administrative compensation and overhead, professional services and costs associated with marketing, sales force and community relations efforts.

Depreciation and amortization expense includes depreciation of fixed assets over the estimated useful life of the assets using the straight-line method, amortization of landfill airspace assets under the units-of-production method, and the amortization of intangible assets (other than goodwill) using the straight-line method. Effective May 1, 2003, the Company adopted SFAS No. 143, *Accounting for Asset Retirement Obligations*. Under SFAS No. 143, except for accretion expense, the Company no longer accrues landfill retirement obligations through a charge to cost of operations, but rather as an increase to landfill assets which are amortized using a straight-line rate per ton as landfill airspace is utilized. The amount of landfill amortization expense related to airspace consumption can vary materially from landfill to landfill depending upon the purchase price and landfill site and cell development costs. The Company depreciates all fixed and intangible assets to a zero net book value, and does not apply a salvage value to any fixed assets.

The Company capitalizes certain direct landfill development costs, such as engineering, permitting, legal, construction and other costs associated directly with the expansion of existing landfills. Additionally, the Company also capitalizes certain third party expenditures related to pending acquisitions, such as legal and engineering costs. The Company routinely evaluates all such capitalized costs, and expenses those costs related to projects not likely to be successful. Internal and indirect landfill development and acquisition costs, such as executive and corporate overhead, public relations and other corporate services, are expensed as incurred. The Company will have material financial obligations relating to closure and post-closure costs of its existing landfills and any disposal facilities which it may own or operate in the future. The Company has provided and will in the future provide accruals for these future financial obligations (generally for a term of 30 years after final closure) based on engineering estimates of consumption of permitted landfill airspace over the useful life of any such landfill. There can be no assurance that the Company's financial obligations for closure or post-closure costs will not exceed the amount accrued and reserved or amounts otherwise receivable pursuant to trust funds.

Results of Operations

The following table sets forth for the periods indicated the percentage relationship that certain items from the Company's Consolidated Financial Statements bear in relation to revenues.

	Three Months Ended July 31,	
	2002	2003
Revenues	100.0%	100.0%
Cost of operations	67.0	65.2
General and administration	12.7	12.7
Depreciation and amortization	10.4	13.0
Operating income	9.9	9.1
Interest expense, net	6.1	5.5
Income from equity method investments,	(0.2)	(0.1)
Other expense/(income), net	(0.1)	(0.1)
Provision for income taxes	1.9	0.8
Income from continuing operations before discontinued operations and		
cumulative effect of change in accounting principle	2.2%	3.0%



Revenues. Revenues decreased \$2.1 million, or (1.8)% to \$113.9 million in the quarter ended July 31, 2003 from \$116.0 million in the quarter ended July 31, 2002. The revenue decrease in the quarter is mainly attributable to a decrease in revenues from businesses divested amounting to \$13.6 million, partially offset by an increase in core solid waste revenues of \$6.0 million, primarily due to volume and price increases. Higher recycling volumes, offset partially by lower commodity prices, resulted in a net increase in revenues of \$1.4 million. Revenues from the rollover effect of acquired businesses accounted for \$4.1 million in the quarter.

Cost of operations. Cost of operations decreased \$3.5 million or (4.5)% to \$74.3 million in the quarter ended July 31, 2003 from \$77.8 million in the quarter ended July 31, 2002. Cost of operations as a percentage of revenues decreased to 65.2% in the quarter ended July 31, 2003 from 67.0% in the prior year. This decrease mainly arose from lower commodity purchases resulting from the divestiture of the export and domestic brokerage businesses.

General and administration. General and administration expenses decreased \$0.2 million, or (1.4)% to \$14.5 million in the quarter ended July 31, 2003 from \$14.7 million in the quarter ended July 31, 2002, and remained constant as a percentage of revenues at 12.7% in the quarters ended July 31, 2003 and July 31, 2002.

Depreciation and amortization. Depreciation and amortization expense increased \$2.7 million, or 22.3%, to \$14.8 million in the quarter ended July 31, 2003 from \$12.1 million in the quarter ended July 31, 2002. While depreciation expense remained relatively constant between periods, the increase was primarily attributable to the increase in landfill amortization as a result of adopting SFAS 143. Depreciation and amortization expense as a percentage of revenue increased to 13.0% in the quarter ended July 31, 2003 from 10.4% in the quarter ended July 31, 2002 which resulted from the increase in landfill amortization expense and lower levels of revenue.

Interest expense, net. Net interest expense decreased \$0.9 million, or (12.7)% to \$6.2 million in the quarter ended July 31, 2003 from \$7.1 million in the quarter ended July 31, 2002. This decrease is primarily attributable to lower average debt balances and lower interest rates on variable debt in the current fiscal quarter, versus last year. Interest expense, as a percentage of revenues decreased to 5.5% in the quarter ended July 31, 2003 from 6.1% in the quarter ended July 31, 2002.

Income from equity method investments. Income in the quarters ended July 31, 2003 and 2002 was from income recorded at U.S. GreenFiber LLC ("GreenFiber"), the Company's 50% owned joint venture. The decline in the quarter is due to higher raw material and marketing costs.

Minority interest. For the quarter ended July 31, 2002, this amount represented the minority owners' interest in the Company's majority owned subsidiary American Ash Recycling of Tennessee, Ltd ("AART") which ceased operations in the third quarter of fiscal 2003.

Other (income)/expenses. Other income increased to \$0.2 million in the quarter ended July 31, 2003 mainly due to a gain on sale of equipment.

Provision for income taxes. Provision for income taxes decreased \$1.3 million to \$0.9 million for the quarter ended July 31, 2003 from \$2.2 million for the quarter ended July 31, 2002. The effective tax rate decreased to 20.0% in the quarter ended July 31, 2003 from 45.8% in the quarter ended July 31, 2002 primarily due to a decrease in the valuation allowance for loss carryforwards.

Reclassification adjustment from discontinued operations, net. In the fourth quarter of fiscal 2003, the Company entered into negotiations with former employees for the transfer of our domestic brokerage operation and a commercial recycling business. The commercial recycling business had been accounted for as a discontinued operation

since fiscal 2001. Due to the nature of the transaction, the Company could not retain historical discontinued accounting treatment for this operation. Therefore the commercial recycling business' operating results have been reclassified from discontinued to continuing operations for the quarter ended July 31, 2002. Also in connection with the discontinued accounting treatment recorded in fiscal 2001, estimated future losses from this operations were recorded and classified as losses from discontinued operations. This amount has been reclassified and offset against actual losses from operations for the quarter ended July, 31, 2002.

Cumulative effect of change in accounting principle, net. Effective May 1, 2003, the Company adopted SFAS No. 143, *Accounting for Asset Retirement Obligations.* The primary modification to the Company's methodology required by SFAS No. 143 is to require that capping, closure and post-closure costs be discounted to present value. Upon adoption of SFAS No. 143 the Company recorded a cumulative effect of change in accounting principle of \$2.7 million (net of taxes of \$1.9 million) in order to reflect the cumulative change in accounting for landfill obligations retroactive to the date of the inception of the landfill.

Effective May 1, 2002, the Company adopted SFAS No. 142, *Goodwill and Other Intangible Assets*, which, among other things, eliminates the amortization of goodwill and requires an annual assessment of goodwill impairment by applying a fair value based test. Goodwill was determined to be impaired and the amount of \$63.9 million (net of tax benefit of \$0.2 million) was charged to earnings in the quarter ended July 31, 2002 as a cumulative effect of change in accounting principle. The goodwill impairment charge was related to our waste-to-energy operation, Maine Energy, and the Brokerage business of the FCR Recycling segment, both of which were acquired as part of our acquisition of KTI. At the time of acquisition, we recorded the fair value of these businesses using an independent third party valuation. The underlying assumptions used to establish the value of these businesses, including earnings projections, commodity pricing assumptions and industry valuation multiples for recycling products, were not realized. Accordingly, goodwill impairment charges were recorded as the net book value of these businesses exceeded their fair value.

Liquidity and Capital Resources

The Company's business is capital intensive. The Company's capital requirements include acquisitions, fixed asset purchases and capital expenditures for landfill development and cell construction, as well as site and cell closure. The Company had a net working capital deficit of \$0.4 million at July 31, 2003 compared to a net working capital deficit of \$4.8 million at April 30, 2003. Net working capital comprises current assets, excluding cash and cash equivalents, minus current liabilities. The main factors accounting for the increase were higher trade receivable balances and lower current closure/post-closure costs and accruals.

The Company has a \$325.0 million credit facility with a group of banks for which Fleet Bank, N.A. is acting as agent. This credit facility consists of a \$175.0 million Senior Secured Revolving Credit Facility ("Revolver") and a Senior Secured Term "B" Loan, which had an outstanding balance of \$150.0 million at July 31, 2003 ("Term Loan"). The Company has the right to increase the amount of the revolver and/or the term loan by an aggregate amount of up to \$50 million at the Company's discretion, provided that the Company is not in default at the time of the increase, subject to the receipt of commitments from lenders for such additional amount. On August 26, 2003, the Company amended the terms of its Senior Secured Term "B" Loan, lowering the borrowing rate and modifying the prepayment provisions to include a prepayment premium for the first two years following the date of the amendment.

The new term loan and revolving credit facility agreement contains covenants that may limit the Company's activities, including covenants that restrict dividends and stock repurchases, limit capital expenditures, and set minimum net worth and profitability requirements and interest coverage and leverage ratios. As of July 31, 2003, the Company considered the profitability covenant, which requires the Company's cumulative adjusted net income for any two consecutive quarters to be positive, to be the most restrictive. As of July 31, 2003, the Company was in compliance with this covenant as the Company reported consolidated adjusted net income of \$4.2 million for the six months ended July 31, 2003. Consolidated adjusted net income is defined by the credit facility agreement. In accordance with such definition, consolidated net income, determined in accordance with generally accepted



accounting principles, is adjusted for elimination of certain nonrecurring charges, extraordinary gains, income from discontinued operations and non-cash income attributable to equity investments.

As of July 31, 2003, the Company had available borrowing capacity under its new \$175.0 million revolving credit facility of up to \$141.7 million, subject to the Company's ability to meet certain borrowing conditions. The available amount is net of outstanding irrevocable letters of credit. The Company intends to use the additional availability under its new senior secured credit facilities to support the Company's acquisition program. This credit facility is secured by all of the Company's assets, including its interest in the equity securities of its subsidiaries. The Revolver matures in January 2008 and the Term Loan matures in January 2010.

The Company has outstanding \$150.0 million of 9.75% senior subordinated notes (the "notes"). The notes mature in January 2013. The senior subordinated note agreement contains covenants that restrict dividends, stock repurchases and other payments, and limits the incurrence of debt and issuance of preferred stock subject to the Company meeting a minimum consolidated fixed charge ratio. The notes are guaranteed jointly and severally, fully and unconditionally by the Company's significant wholly-owned subsidiaries. Pursuant to the terms of the agreements under which the notes were issued, the Company, on August 18, 2003 commenced an exchange offer pursuant to which holders of the notes have the right to exchange the notes for substantially similar registered notes. The exchange offer is scheduled to be completed on September 26, 2003.

Net cash provided by operating activities amounted to \$15.0 million for the three months ended July 31, 2003 compared to \$9.9 million for the same period of the prior fiscal year. The increase was due to changes in the Company's working capital and an increase in depreciation and amortization.

Net cash used in investing activities was \$24.1 million for the three months ended July 31, 2003 compared to \$10.7 million used in investing activities in the same period of the prior fiscal year. The increase in cash used in investing activities was due to an increase in capital expenditures and acquisitions.

Net cash used in financing activities was \$0.8 million for the three months ended July 31, 2003 compared to \$2.5 million used by financing activities in the same period of the prior fiscal year. The decreased in cash used in investing activities is primarily due to lower net borrowings in the quarter ended July, 31, 2003 compared to the prior year comparable period.

Inflation and Prevailing Economic Conditions

To date, inflation has not had a significant impact on the Company's operations. Consistent with industry practice, most of the Company's contracts provide for a pass-through of certain costs, including increases in landfill tipping fees and, in some cases, fuel costs. The Company therefore believes it should be able to implement price increases sufficient to offset most cost increases resulting from inflation. However, competitive factors may require the Company to absorb at least a portion of these cost increases, particularly during periods of high inflation.

The Company's business is located mainly in the eastern United States. Therefore, the Company's business, financial condition and results of operations are susceptible to downturns in the general economy in this geographic region and other factors affecting the region, such as state regulations and severe weather conditions. The Company is unable to forecast or determine the timing and /or the future impact of a sustained economic slowdown.

New Accounting Pronouncements

In May 2003, the FASB issued SFAS No. 150, Accounting for Certain Financial Instruments with Characteristics of Both Liability and Equity. The statement changes the accounting of certain financial instruments that, under previous guidance, issuers could account for as equity. The new statement requires that those instruments be classified as liabilities in statements of financial position. SFAS No. 150 is effective for all financial instruments entered into or modified after May 31, 2003, and otherwise is effective at the beginning of the first interim period beginning after June 15, 2003. Management is evaluating the effect of the statement on our results of operations and financial positions as well as related disclosures.

Certain Factors That May Affect Future Results

The following important factors, among others, could cause actual results to differ materially from those indicated by forward-looking statements made in this Form 10-Q and presented elsewhere by management from time to time.

The Company's increased leverage may restrict its future operations and impact its ability to make future acquisitions.

As a result of the acquisition of KTI, the Company's indebtedness increased substantially. In addition, the Company's indebtedness has increased further as a result of the Company's new credit facility and the completion of the offering of the senior subordinated notes. The payment of interest and principal due under this indebtedness has reduced, and may continue to reduce, funds available for other business purposes, including capital expenditures and acquisitions. In addition, the aggregate amount of indebtedness has limited and may continue to limit the Company's ability to incur additional indebtedness, and thereby may limit its acquisition program.

The Company may not be successful in making acquisitions of solid waste assets, including developing additional disposal capacity, or in integrating acquired businesses or assets, which could limit the Company's future growth.

The Company's strategy envisions that a substantial part of the Company's future growth will come from making acquisitions of traditional solid waste assets or operations and acquiring or developing additional disposal capacity. These acquisitions may include "tuck-in" acquisitions within the Company's existing markets, assets that are adjacent to or outside the Company's existing markets, or larger, more strategic acquisitions. In addition, from time to time the Company may acquire businesses that are complementary to the Company's core business strategy. The Company may not be able to identify suitable acquisition candidates. If the Company identifies suitable acquisition candidates, the Company may be unable to negotiate successfully their acquisition at a price or on terms and conditions favorable to the Company. Furthermore, the Company may be unable to obtain the necessary regulatory approval to complete potential acquisitions.

The Company's ability to achieve the benefits the Company anticipates from acquisitions, including cost savings and operating efficiencies, depends in part on the Company's ability to successfully integrate the operations of such acquired businesses with the Company's operations. The integration of acquired businesses and other assets may require significant management time and company resources that would otherwise be available for the ongoing management of the Company's existing operations.

In addition, the process of acquiring or developing additional disposal capacity is lengthy, expensive and uncertain. For example, the Company is currently involved in litigation with the Town of Bethlehem, New Hampshire relating to the expansion of a landfill owned by the Company's wholly owned subsidiary, North Country Environmental Services, Inc. Moreover, the disposal capacity at the Company's existing landfills is limited by the remaining available volume at the Company's landfills and annual and/or daily disposal limits imposed by the various governmental authorities with jurisdiction over the Company's landfills. The Company typically reaches or approximates the Company's daily and annual maximum permitted disposal capacity at all of the Company's landfills. If the Company is unable to develop or acquire additional disposal capacity, the Company's ability to achieve economies from the internalization of the Company's waste stream will be limited and the Company may be required to increase the Company's utilization of disposal facilities owned by third parties, which could reduce the Company's revenues and/or the Company's operating margins.

The Company's ability to make acquisitions is dependent on the availability of adequate cash and the attractiveness of the Company's stock price.

The Company anticipates that any future business acquisitions will be financed through cash from operations, borrowings under the Company's new senior secured credit facilities, the issuance of shares of the Company's Class A common stock and/or seller financing. The Company may not have sufficient existing capital resources and may be unable to raise sufficient additional capital resources on terms satisfactory to the Company, if at all, in order to meet the Company's capital requirements for such acquisitions.



The Company also believes that a significant factor in the Company's ability to close acquisitions will be the attractiveness to the Company and to persons selling businesses to the Company of the Company's Class A common stock as consideration for potential acquisition candidates. This attractiveness may, in large part, be dependent upon the relative market price and capital appreciation prospects of the Company's Class A common stock compared to the equity securities of the Company's competitors. The trading price of the Company's Class A common stock on the Nasdaq National Market has limited the Company's willingness to use the Company's equity as consideration and the willingness of sellers to accept the Company's shares and as a result has limited, and could continue to limit, the size and scope of the Company's acquisition program.

Environmental regulations and litigation could subject the Company to fines, penalties, judgments and limitations on the Company's ability to expand.

The Company is subject to potential liability and restrictions under environmental laws, including those relating to transport, recycling, treatment, storage and disposal of wastes, discharges to air and water, and the remediation of contaminated soil, surface water and groundwater. The waste management industry has been and will continue to be subject to regulation, including permitting and related financial assurance requirements, as well as to attempts to further regulate the industry through new legislation. The Company's waste-to-energy facility is subject to regulations limiting discharges of pollution into the air and water, and the Company's solid waste operations are subject to a wide range of federal, state and, in some cases, local environmental, odor and noise and land use restrictions. For example, the Company's waste-to-energy facility in Biddeford, Maine is affected by zoning restrictions and air emissions limitations in its efforts to implement a new odor control system. If the Company is not able to comply with the requirements that apply to a particular facility or if the Company operates without necessary approvals, the Company could be subject to civil, and possibly criminal, fines and penalties, and the Company may be required to spend substantial capital to bring an operation into compliance or to temporarily or permanently discontinue, and/or take corrective actions, possibly including removal of landfilled materials, regarding an operation that is not permitted under the law. The Company may be useful to Company's environmental liabilities. Those costs or actions could be significant to the Company may and impact the Company's environmental liabilities.

Environmental and land use laws also impact the Company's ability to expand and, in the case of the Company's solid waste operations, may dictate those geographic areas from which the Company must, or, from which the Company may not, accept waste. Those laws and regulations may limit the overall size and daily waste volume that may be accepted by a solid waste operation. If the Company is not able to expand or otherwise operate one or more of the Company's facilities because of limits imposed under environmental laws, the Company may be required to increase the Company's utilization of disposal facilities owned by third parties, which could reduce the Company's revenues and/or operating margins.

The Company has historically grown and intends to continue to grow through acquisitions, and the Company has tried and will continue to try to evaluate and address environmental risks and liabilities presented by newly acquired businesses as the Company has identified them. It is possible that some liabilities, including ones that may exist only because of the past operations of an acquired business, may prove to be more difficult or costly to address than the Company anticipates. It is also possible that government officials responsible for enforcing environmental laws may believe an issue is more serious than the Company would expect, or that the Company will fail to identify or fully appreciate an existing liability before the Company becomes legally responsible to address it. Some of the legal sanctions to which the Company could become subject could cause the Company to lose a needed permit, or prevent the Company from or delay the Company in obtaining or renewing permits to operate the Company's facilities or harm the Company's reputation.

The Company's operating program depends on the Company's ability to operate and expand the landfills the Company owns and leases and to develop new landfill sites. Localities where the Company operates generally seek to regulate some or all landfill operations, including siting and expansion of operations. The laws adopted by municipalities in which the Company's landfills are located may limit or prohibit the



expansion of the landfill as well as the amount of waste that the Company can accept at the landfill on a daily or annual basis and any effort to acquire or expand landfills typically involves a significant amount of time and expense. For example, expansion at the Company's North County Environmental Services, Inc. landfill will require the New Hampshire Supreme Court to overturn a lower court ruling which interpreted a local ordinance to prohibit expansion of the landfill with respect to 1.3 million tons of capacity, and expansion of the Company's Hyland landfill is subject to the passage of a town-wide referendum. The Company may not be successful in obtaining new landfill sites or expanding the permitted capacity of any of the Company's current landfills once their remaining disposal capacity has been consumed. If the Company is unable to develop additional disposal capacity, the Company's ability to achieve economies from the internalization of the Company's waste stream will be limited and the Company will be required to utilize the disposal facilities of the Company's competitors.

In addition to the costs of complying with environmental laws and regulations, the Company incurs costs defending against environmental litigation brought by governmental agencies and private parties. The Company is, and also may be in the future, a defendant in lawsuits brought by parties alleging environmental damage, personal injury, and/or property damage. For example, the Company is one of over twenty defendants named in a toxic tort lawsuit filed on July 2, 2001 by residents surrounding three sites in Cheektowaga, New York alleging, among other things, that the Company has liability as a result of the Company's airspace agreement at the Schultz construction and demolition debris landfill. In addition, the Company is also a defendant in a lawsuit filed on January 10, 2002 by the City of Biddeford, Maine alleging, among other things, that the Company's subsidiary, Maine Energy, processed amounts of waste above contractual limits without notice to the city. A significant judgment against the Company could harm the Company's business, the Company's prospects and the Company's reputation. See "Legal Proceedings."

The Company's operations would be adversely affected if the Company does not have access to sufficient capital.

The Company's ability to remain competitive and sustain the Company's operations depends in part on cash flow from operations and the Company's access to capital. The Company intends to fund the Company's cash needs primarily through cash from operations and borrowings under the Company's new senior secured credit facilities. However, the Company may require additional equity and/or debt financing for debt repayment obligations and to fund the Company's growth and operations. In addition, if the Company undertakes more acquisitions or further expands the Company's operations, the Company's capital requirements may increase. The Company may not have access to the amount of capital that the Company requires from time to time, on favorable terms or at all.

The Company's results of operations could continue to be affected by changing prices or market requirements for recyclable materials.

The Company's results of operations have been and may continue to be affected by changing purchase or resale prices or market requirements for recyclable materials. The Company's recycling business involves the purchase and sale of recyclable materials, some of which are priced on a commodity basis. The resale and purchase prices of, and market demand for, recyclable materials, particularly waste paper, plastic and ferrous and aluminum metals, can be volatile due to numerous factors beyond the Company's control. Although the Company seeks to limit the Company's exposure to fluctuating commodity prices through the use of hedging agreements and long-term supply contracts with customers, these changes have in the past contributed, and may continue to contribute, to significant variability in the Company's period-to-period results of operations.

The Company's business is geographically concentrated and is therefore subject to regional economic downturns.

The Company's operations and customers are principally located in the eastern United States. Therefore, the Company's business, financial condition and results of operations are susceptible to regional economic downturns and other regional factors, including state regulations and budget constraints and severe

weather conditions. In addition, as the Company expands in the Company's existing markets, opportunities for growth within these regions will become more limited and the geographic concentration of the Company's business will increase.

Maine Energy may be required to make a payment in connection with the payoff of certain obligations and limited partner loans earlier than the Company had anticipated and which may exceed the amount of the liability the Company recorded in connection with the KTI acquisition.

Under the terms of waste handling agreements among the Biddeford-Saco Waste Handling Committee, the cities of Biddeford and Saco, Maine, 13 other municipalities and the Company's subsidiary Maine Energy, Maine Energy will be required, following the date on which the bonds that financed Maine Energy and certain limited partner loans to Maine Energy are paid in full, to pay a residual cancellation payment to the respective municipalities party to those agreements equal to an aggregate of 18% of the fair market value of the equity of the partners in Maine Energy. In connection with the Company's merger with KTI, the Company estimated the fair market value of Maine Energy as of the date the limited partner loans are anticipated to be paid in full, and recorded a liability equal to 18% of such amount. The Company's estimate of the fair market value of Maine Energy may not prove to be accurate, and in the event the Company has underestimated the value of Maine Energy, the Company could be required to recognize unanticipated charges, in which case the Company's operating results could be harmed.

In connection with these waste handling agreements, the cities of Biddeford and Saco and the additional 13 municipalities that were parties to the agreements have filed lawsuits in the State of Maine seeking the residual cancellation payments and alleging, among other things, the Company's breach of the waste handling agreement for the Company's failure to pay the residual cancellation payments in connection with the KTI merger, failure to pay off limited partner loans in accordance with the terms of the agreement and processing amounts of waste above contractual limits without issuance of proper notice. The complaint seeks damages for breach of contract and a court order requiring the Company to provide an accounting of all relevant transactions since May 3, 1996. If the plaintiffs are successful in their claims against the Company and damages are awarded the Company's operating income in the period in which such a claim is paid would be impacted. See "Legal Proceedings."

The Company may not be able to effectively compete in the highly competitive solid waste services industry.

The solid waste services industry is highly competitive, has undergone a period of rapid consolidation and requires substantial labor and capital resources. Some of the markets in which the Company competes or will likely compete are served by one or more of the large national or multinational solid waste companies, as well as numerous regional and local solid waste companies. Intense competition exists not only to provide services to customers, but also to acquire other businesses within each market. Some of the Company's competitors have significantly greater financial and other resources than the Company. From time to time, competitors may reduce the price of their services in an effort to expand market share or to win a competitively bid contract. These practices may either require the Company to reduce the pricing of the Company's services or result in the Company's loss of business.

As is generally the case in the industry, some municipal contracts are subject to periodic competitive bidding. The Company may not be the successful bidder to obtain or retain these contracts. If the Company is unable to compete with larger and better capitalized companies, or to replace municipal contracts lost through the competitive bidding process with comparable contracts or other revenue sources within a reasonable time period the Company's revenues would decrease and the Company's operating results would be harmed.

In the Company's solid waste disposal markets the Company also competes with operators of alternative disposal and recycling facilities and with counties, municipalities and solid waste districts that maintain their own waste collection, recycling and disposal operations. These entities may have financial advantages

because user fees or similar charges, tax revenues and tax-exempt financing may be more available to them than to the Company.

The Company's GreenFiber insulation manufacturing joint venture with Louisiana-Pacific Corporation competes with other parties, principally national manufacturers of fiberglass insulation, which have substantially greater resources than GreenFiber does, which they could use for product development, marketing or other purposes to the Company's detriment.

The Company's results of operations and financial condition may be negatively affected if the Company inadequately accrues for closure and postclosure costs.

The Company has material financial obligations relating to closure and post-closure costs of the Company's existing landfills and will have material financial obligations with respect to any disposal facilities which the Company may own or operate in the future. Once the permitted capacity of a particular landfill is reached and additional capacity is not authorized, the landfill must be closed and capped, and post-closure maintenance started. The Company establishes reserves for the estimated costs associated with such closure and post-closure obligations over the anticipated useful life of each landfill on a per ton basis. In addition to the landfills the Company currently operates, the Company owns four unlined landfills, which are not currently in operation. The Company has provided and will in the future provide accruals for financial obligations relating to closure and post-closure costs of the Company's owned or operated landfills, generally for a term of 30 years after final closure of a landfill. The Company's financial obligations for closure or post-closure costs could exceed the amount accrued and reserved or amounts otherwise receivable pursuant to trust funds established for this purpose. Such a circumstance could result in significant unanticipated charges.

Fluctuations in fuel costs could affect the Company's operating expenses and results.

The price and supply of fuel is unpredictable and fluctuates based on events beyond the Company's control, including among others, geopolitical developments, supply and demand for oil and gas, actions by OPEC and other oil and gas producers, war and unrest in oil producing countries and regional production patterns. Because fuel is needed to run the Company's fleet of trucks, price escalations for fuel increase the Company's operating expenses. During fiscal 2003, the Company used approximately 6.5 million gallons of diesel fuel in the Company's solid waste operations. Although many of the Company's customer contracts permit the Company to pass on some or all fuel increases to the Company's customers, the Company may choose not to do so for competitive reasons.

The Company could be precluded from entering into contracts or obtaining permits if the Company is unable to obtain third party financial assurance to secure the Company's contractual obligations.

Municipal solid waste collection and recycling contracts, obligations associated with landfill closure and the operation and closure of waste-to-energy facilities may require performance or surety bonds, letters of credit or other means of financial assurance to secure the Company's contractual performance. If the Company is unable to obtain the necessary financial assurance in sufficient amounts or at acceptable rates, the Company could be precluded from entering into additional municipal solid waste collection contracts or from obtaining or retaining landfill operating permits. Any future difficulty in obtaining insurance could also impair the Company's ability to secure future contracts conditioned upon the contractor having adequate insurance coverage

The Company may be required to write-off capitalized charges or intangible assets in the future, which could harm the Company's earnings.

Any write-off of capitalized costs or intangible assets reduces the Company's earnings and, consequently, could affect the market price of the Company's Class A common stock. In accordance with generally accepted accounting principles, the Company capitalizes certain expenditures and advances relating to the Company's acquisitions, pending acquisitions, landfills and development projects. From time to time in future periods, the Company may be required to incur a charge against earnings in an amount equal to any



unamortized capitalized expenditures and advances, net of any portion thereof that the Company estimate will be recoverable, through sale or otherwise, relating to (1) any operation that is permanently shut down or has not generated or is not expected to generate sufficient cash flow, (2) any pending acquisition that is not consummated, (3) any landfill or development project that is not expected to be successfully completed, and (4) any goodwill or other intangible assets that are determined to be impaired. The Company has incurred such charges in the past.

The Company's revenues and the Company's operating income experience seasonal fluctuations.

The Company's transfer and disposal revenues have historically been lower during the months of November through March. This seasonality reflects the lower volume of waste during the late fall, winter and early spring months primarily because:

• the volume of waste relating to construction and demolition activities decreases substantially during the winter months in the northeastern United States; and

• decreased tourism in Vermont, Maine and eastern New York during the winter months tends to lower the volume of waste generated by commercial and restaurant customers, which is partially offset by increased volume from the winter ski industry.

Since certain of the Company's operating and fixed costs remain constant throughout the fiscal year, operating income is therefore impacted by a similar seasonality. In addition, particularly harsh weather conditions typically result in increased operating costs to the Company's operations.

The Company's recycling business experiences increased volumes of newspaper in November and December due to increased newspaper advertising and retail activity during the holiday season. The Company's cellulose insulation joint venture experiences lower sales in November and December because of lower production of manufactured housing due to holiday plant shutdowns.

Efforts by labor unions to organize the Company's employees could divert management attention and increase the Company's operating expenses.

Labor unions regularly make attempts to organize the Company's employees, and these efforts will likely continue in the future. Certain groups of the Company's employees have chosen to be represented by unions, and the Company has negotiated collective bargaining agreements with these groups. Additional groups of employees may seek union representation in the future, and the negotiation of collective bargaining agreements could divert management attention and result in increased operating expenses and lower net income. If the Company is unable to negotiate acceptable collective bargaining agreements, the Company might have to wait through "cooling off" periods, which are often followed by union-initiated work stoppages, including strikes. Depending on the type and duration of any labor disruptions, the Company's revenues could decrease and the Company's operating expenses could increase, which could adversely affect the Company's financial condition, results of operations and cash flows. As of July 1, 2003, approximately 5.1% of the Company's employees involved in collection, transfer, disposal, recycling or other operations, including the Company's employees at the Company's Maine Energy waste-to-energy facility, were represented by unions.

The Company's Class B common stock has ten votes per share and is held exclusively by John W. Casella and Douglas R. Casella.

The holders of the Company's Class B common stock are entitled to ten votes per share and the holders of the Company's Class A common stock are entitled to one vote per share. At August 29, 2003, an aggregate of 988,200 shares of the Company's Class B common stock, representing 9,882,000 votes, were outstanding, all of which were beneficially owned by John W. Casella, the Company's Chairman and Chief Executive Officer, or by his brother, Douglas R. Casella, a member of the Company's Board of Directors. Based on the number of shares of common stock and Series A redeemable convertible preferred stock outstanding on August 29, 2003, the shares of the Company's Class A common stock and Class B common



stock beneficially owned by John W. Casella and Douglas R. Casella represent approximately 30.5% of the aggregate voting power of the Company's stockholders. Consequently, John W. Casella and Douglas R. Casella are able to substantially influence all matters for stockholder consideration.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Interest rate volatility

The interest rate on \$53.0 million of long term debt has been fixed through two interest rate swaps. The Company has interest rate risk relating to approximately \$97.0 million of long-term debt at July 31, 2003. The interest rate on the variable rate portion of long-term debt was approximately 4.375% at July 31, 2003. Should the average interest rate on the variable rate portion of long-term debt change by 100 basis points, it would have an approximate interest expense change of \$0.2 million for the quarter reported.

The remainder of the Company's long-term debt is at fixed rates and not subject to interest rate risk.

Commodity price volatility

The Company is subject to commodity price fluctuations related to the portion of its sales of recyclable commodities that are not under floor or flat pricing arrangements. To minimize the Company's commodity exposure, the Company has entered into twelve commodity hedging agreements that have been authorized pursuant to the Company's policies and procedures. The Company does not use financial instruments for trading purposes and is not a party to any leveraged derivatives. If commodity prices were to change by 10%, the impact on the Company's operating margin is estimated at \$1.0 million for the quarter reported.

ITEM 4. CONTROLS AND PROCEDURES

- a) Evaluation of disclosure controls and procedures. Based on their evaluation of the Company's disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934) as of the end of the period covered by this Quarterly Report on Form 10-Q, the Company's chief executive officer and chief financial officer have concluded that the Company's disclosure controls and procedures are designed to ensure that information required to be disclosed by the Company in the reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms and are operating in an effective manner.
- b) Changes in internal controls. During the period covered by this Quarterly Report on Form 10-Q, there were no changes in the Company's internal control over financial reporting (as defined in Rule 13a-15(f)) that have materially affected or are reasonably likely to materially affect, the Company's internal controls over financial reporting.

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

Pursuant to the applicable rules of the Securities and Exchange Commission, information as to material legal proceedings is presented in the Company's Annual Report on Form 10-K, and information as to such legal proceedings is only included in this Quarterly Report on Form 10-Q and in any other Quarterly Report on Form 10-Q to the extent there have been developments with respect to such proceedings.

The Company's wholly owned subsidiary, North Country Environmental Services, Inc. ("NCES"), is a party to an appeal against the Town of Bethlehem, New Hampshire ("Town") before the New Hampshire Supreme Court. The appeal arose from cross actions for declaratory and injunctive relief filed by NCES and the Town to determine the permitted extent of NCES's landfill in the Town. The New Hampshire Superior Court in Grafton ruled on February 1, 1999 that the Town could not enforce an ordinance purportedly prohibiting expansion of the landfill, at least with respect to 51 acres of NCES's 87 acre parcel, based upon certain existing land-use approvals. As a result, NCES was able to construct and operate "Stage II, Phase II" of the landfill. In May 2001, the Supreme Court denied the Town's appeal. Notwithstanding the Supreme Court's ruling, the Town continued to assert jurisdiction to conduct unqualified site plan review with respect to Stage III and has further stated that the Town's height ordinance and building permit process may apply to Stage III. On September 12, 2001, the Company filed a petition for, among other things, declaratory relief. On December 4, 2001, the Town filed an answer to the Company's petition asserting counterclaims seeking, among other things, authorization to assert site plan review over Stage III, which commenced operation in December 2000, as well as the methane gas utilization/leachate handling facility operating in Stage III, and also an order declaring that an ordinance prohibiting landfills applies to Stage IV expansion. The trial related to the Town's jurisdiction was held in December 2002 and on April 24, 2003, the Grafton Superior Court issued its ruling, upholding the Town's 1992 ordinance preventing the location or expansion of any landfill, ruling that the ordinance may be applied to any part of Stage IV that goes beyond the 51 acres; ruling that the Town's height ordinance is valid within the 51 acres; upholding the Town's right to require Site Plan Review, except that there are certain areas within the Town's Site Plan Review regulation that are preempted; ruling that the methane gas utilization/leachate handling facility is not subject to the Town's ordinance forbidding incinerators. On May 27, 2003, NCES appealed the Court ruling to the New Hampshire Supreme Court, which agreed to hear the case, except for the Company's appeal of the Superior Court's ruling denying attorneys fees. The Supreme Court scheduled briefing deadlines through October 2003, at which time oral argument will be scheduled. If upheld on appeal, the Superior Court's rulings would have the effect of preventing the development of Stage IV and limiting the further development of Stage III to the extent of the height restriction. If the Company does not prevail, the Company may be unable to continue, or to expand, current operations in accordance with the Company's plans.

On or about December 11, 2001, the Company was served with a bill in equity in aid of discovery filed in the Strafford Superior Court in New Hampshire by Nancy Hager. The bill in equity seeks an accounting related to non-compete tip fee payments from the Company to Ms. Hager pursuant to a 1993 release and settlement agreement. The bill in equity is a request for pre-litigation discovery for the purpose of investigating a potential claim for failure to pay appropriate non-compete tip fee amounts. In light of an arbitration clause in the 1993 release and settlement agreement, the Company filed a motion to stay the proceedings under the bill in equity pending completion of the arbitration process. On March 18, 2002, the court granted the Company's motion to stay. On August 5, 2002, the court extended the stay pending the arbitration process. On October 17, 2002, Ms. Hager voluntarily withdrew her bill in equity without prejudice. On January 15, 2003, Ms. Hager filed a written request for arbitration with the American Arbitration Association. The arbitration hearing is in process and will continue through mid-September. On June 5, 2003, Mrs. Hager submitted a disclosure letter to the arbitration panel alleging that she is owed between \$480,000 and \$560,000. On July 7, 2003, Ms. Hager revised her claim to allege that she is owed between \$637,000 and \$1,000,000. The Company believes it has meritorious defenses to these claims and that no loss as a result of these claims is probable or reasonably possible.

The Company offers no prediction of the outcome of any of the proceedings described above. The Company is vigorously defending each of these lawsuits. However, the Company may not prevail and any judgments against the Company, if sustained on appeal, may result in the incurrence of significant costs or the restriction of the Company's operations.

ITEM 2. CHANGES IN SECURITIES

None

ITEM 3. DEFAULTS ON SENIOR SECURITIES

None.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

None.

ITEM 5. OTHER INFORMATION

None.

ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K

(a) Exhibits:

The Exhibits that are filed as part of this Quarterly Report on Form 10-Q or that are incorporated by reference herein are set forth in the Exhibit Index hereto.

(b) Reports on Form 8-K:

The Company filed a report on Form 8-K on July 24, 2003 reporting under item 12 thereof its results for the fiscal year ended April 30, 2003

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Casella Waste Systems, Inc.

Date: September 12, 2003

By: <u>/s/ Richard A. Norris</u> Richard A. Norris Chief Financial Officer (Principal Financial and Accounting Officer and Duly Authorized Officer)

Exhibit Index

- 10.1 Second Amended and Restated Revolving Credit and Term Loan Agreement, dated January 24, 2003, by and among Casella Waste Systems, Inc. and its Subsidiaries (other than Excluded Subsidiaries), the lending institutions party thereto and Fleet National Bank, individually and as administrative agent, and Bank of America, N.A., individually and as syndication agent, with Fleet Securities, Inc. and Banc of American Securities LLC acting as Co-Arrangers.
- 10.2 Amendment No. 2 to Second Amended and Restated Revolving Credit and Term Loan Agreement.
- 31.1 Section 302 Certification of John W. Casella, Chairman of the Board of Directors and Chief Executive Officer.
- 31.2 Section 302 Certification of Richard A. Norris, Senior Vice President and Chief Financial Officer.
- 32.1 Certification pursuant to 18 U.S.C. S 1350 of John W. Casella, Chairman of the Board of Directors and Chief Executive Officer .
- 32.2 Certification pursuant to 18 U.S.C. S 1350 of Richard A. Norris, Senior Vice President and Chief Financial Officer.

SECOND AMENDED AND RESTATED REVOLVING CREDIT AND TERM LOAN AGREEMENT dated as of January 24, 2003

by and among

CASELLA WASTE SYSTEMS, INC.

and its Subsidiaries (other than Excluded Subsidiaries), as Borrowers

THE LENDING INSTITUTIONS PARTY HERETO

and

FLEET NATIONAL BANK, as Administrative Agent

and

BANK OF AMERICA, N.A., as Syndication Agent

with

FLEET SECURITIES, INC.

and

BANC OF AMERICA SECURITIES LLC

acting as Co-Arrangers

TABLE OF CONTENTS

§1. DEFINITIONS AND RULES OF INTERPRETATION.

<u>§1.1. Definitions.</u> §1.2. Rules of Interpretation.

§2. THE REVOLVING CREDIT LOANS.

§2.1. Commitment to Lend.

§2.2. Reduction of Total Commitment.

§2.3. The Revolving Credit Notes.

§2.4. Interest on Revolving Credit Loans; Maturity of the Revolving Credit Loans.

§2.5. Mandatory Repayments of the Revolving Credit Loans.

§2.6. Requests for Revolving Credit Loans.

§2.7. Funds for Revolving Credit Loans.

§2.8. Swing Line Loans; Settlements.

§2.9. Optional Prepayments or Repayments of Revolving Credit Loans.

§3. LETTERS OF CREDIT.

§3.1. Letter of Credit Commitments.

§3.2. Reimbursement Obligations of the Borrowers.

§3.3. Letter of Credit Payments.

§3.4. Obligations Absolute.

§3.5. Reliance by Issuing Lender.

§4. THE TERM LOAN.

§4.1. Commitment to Lend.

§4.2. The Term Notes.

§4.3. Scheduled Installment Payments of Principal of Term Loan.

§4.4. Mandatory Prepayments of Term Loan.

§4.4.1. Mandatory Prepayments.

§4.4.2. Application of Payments.

§4.5. Optional Prepayment of Term Loan.

§4.6. Interest on Term Loan.

§4.6.1. Interest Rates.

§4.6.2. Notification by Borrowers.

§4.6.3. Amounts, etc.

\$5. FEES; PAYMENTS; COMPUTATIONS; JOINT AND SEVERAL LIABILITY; CERTAIN GENERAL PROVISIONS.

<u>§5.1. Fees.</u>

§5.2. Payments.

§5.3. Computations.

§5.4. Capital Adequacy.

§5.5. Certificate.

§5.6. Interest After Default.

§5.7. Interest Limitation.

§5.8. Additional Costs, Etc.

§5.9. Concerning Joint and Several Liability of the Borrowers.

§5.10. Currency of Account. All of the Loans and Letters of Credit hereunder shall be denominated and payable in Dollars.

- §5.11. Election of Eurodollar Rate; Notice of Election; Interest Periods; Minimum Amounts.
- §5.12. Eurodollar Indemnity.

§5.13. Illegality; Inability to Determine Eurodollar Rate.

§6. REPRESENTATIONS AND WARRANTIES.

- §6.1. Corporate Authority.
- §6.2. Governmental Approvals.
- §6.3. Title to Properties; Leases.
- §6.4. Financial Statements; Solvency.
- §6.5. No Material Changes, Etc. §6.6. Permits, Franchises, Patents, Copyrights, Etc.
- §6.7. Litigation.
- §6.8. No Materially Adverse Contracts, Etc.
- §6.9. Compliance With Other Instruments, Laws, Etc.
- §6.10. Tax Status.
- §6.11. No Event of Default.
- §6.12. Holding Company and Investment Company Acts.
- §6.13. Absence of Financing Statements, Etc.
- §6.14. Employee Benefit Plans.
- §6.15. Use of Proceeds.
- <u>§6.16. Environmental Compliance.</u> <u>§6.17. Perfection of Security Interests.</u>
- §6.18. Certain Transactions.
- §6.19. Subsidiaries.
- §6.20. Capitalization.
- §6.21. True Copies of Charter and Other Documents.
- §6.22. Disclosure.
- §6.23. Guarantees of Excluded Subsidiaries.
- <u>§6.24.</u>
- §7. AFFIRMATIVE COVENANTS OF THE BORROWERS.
 - §7.1. Punctual Payment.
 - §7.2. Maintenance of Office.
 - §7.3. Records and Accounts.
 - §7.4. Financial Statements, Certificates and Information.
 - §7.5. Legal Existence and Conduct of Business.
 - §7.6. Maintenance of Properties.
 - §7.7. Insurance.
 - §7.8. Taxes.
 - §7.9. Inspection of Properties, Books, and Contracts.
 - §7.10. Compliance with Laws, Contracts, Licenses and Permits; Maintenance of Material Licenses and Permits.
 - §7.11. Environmental Indemnification.
 - §7.12. Further Assurances.
 - §7.13. Notice of Potential Claims or Litigation.
 - §7.14. Notice of Certain Events Concerning Insurance and Environmental Claims.
 - ii

<u>§7.15. Notice of Default.</u>

- §7.16. Closure and Post Closure Liabilities.
- §7.17. Subsidiaries.
- §7.18. Interest Rate Protection.
- §7.19. Additional Borrowers.

§8. CERTAIN NEGATIVE COVENANTS OF THE BORROWERS.

- §8.1. Restrictions on Indebtedness.
- §8.2. Restrictions on Liens.
- §8.3. Restrictions on Investments.
- §8.4. Mergers, Consolidations, Sales.
 - §8.4.1. Mergers and Acquisitions.
- §8.4.2. Dispositions of Assets.
- §8.5. Sale and Leaseback.
- §8.6. Restricted Payments.
- §8.7. Employee Benefit Plans.
- §8.8. Prepayments of Certain Obligations; Modifications of Subordinated Debt.
- <u>§8.9. Negative Pledges and Upstream Limitations.</u> <u>§8.10. Transactions with Affiliates.</u>
- §8.11. Business Activities.
- <u>§8.12. No Other Senior Debt.</u>
- §8.13. Actions Otherwise Prohibited by Subordinated Debt.

§9. FINANCIAL COVENANTS.

- §9.1. Interest Coverage Ratio.
- §9.2. Profitable Operations.
- §9.3. Consolidated Total Funded Debt to Consolidated EBITDA.
- §9.4. Consolidated Senior Funded Debt to Consolidated EBITDA.
- §9.5. Consolidated Net Worth.
- §9.6. Capital Expenditures.

§10. CLOSING CONDITIONS.

- §10.1. Corporate Action.
 §10.2. Loan Documents; Senior Subordinated Debt Documents.
 §10.3. Officer's Certificate; Certified Copies of Charter Documents.
 §10.4. Incumbency Certificate.
 §10.5. Validity of Liens.
 §10.6. Certificates of Insurance.
 §10.7. Opinion of Counsel.
 §10.8. Payment of Fees.
 §10.9. Payoff.
 §10.10. Financial Statements.
 §10.11. Financial Condition.
 §10.12. Perfection Certificates and UCC Search Results.
 §10.13. Issuance of Senior Subordinated Debt.

 §11. CONDITIONS OF ALL LOANS.

 §11.1. Representations True; No Event of Default.
 - §11.2. Performance; No Event of Default.

iii

§11.3. No Legal Impediment. §11.4. Proceedings and Documents.

§12. COLLATERAL SECURITY.

§13. EVENTS OF DEFAULT; ACCELERATION; TERMINATION OF COMMITMENT. §13.1. Events of Default and Acceleration. §13.2. Termination of Commitments.

§13.3. Remedies.

§13.4. Distribution of Collateral Proceeds.

§14. SETOFF.

§15. THE AGENTS.

<u>§15.1. Appointment, Powers and Immunities.</u>

§15.2. Actions By Agents.

§15.3. Indemnification of Agents.

§15.4. Reimbursement for Advances Made by Administrative Agent.

§15.5. Closing Documentation, etc. §15.6. Non-Reliance on Agents and Other Lenders.

§15.7. Resignation.

§15.8. Action by the Lenders, Consents, Amendments, Waivers, Etc.

§16. EXPENSES.

§17. INDEMNIFICATION.

§18. SURVIVAL OF COVENANTS, ETC.

§19. ASSIGNMENTS AND PARTICIPATION.

§20. PARTIES IN INTEREST.

§21. NOTICES, ETC.

§22. MISCELLANEOUS.

§23. ENTIRE AGREEMENT, ETC.

§24. WAIVER OF JURY TRIAL.

§25. GOVERNING LAW.

§26. SEVERABILITY.

§27. PARI PASSU TREATMENT.

§28. EXISTING CREDIT AGREEMENT SUPERSEDED.

§29. TREATMENT OF CERTAIN CONFIDENTIAL INFORMATION.

iv

<u>§29.1. Confidentiality.</u> <u>§29.2. Prior Notification.</u> <u>§29.3. Other.</u>

<u>Exhibits</u>

Exhibit A-1	_	Form of Revolving Credit Note
Exhibit A-2	_	Form of Swing Line Note
Exhibit A-3	_	Form of Term Note
Exhibit B	_	Form of Loan and Letter of Credit Request
Exhibit C	_	Form of Compliance Certificate
Exhibit D	_	Form of Environmental Compliance Certificate
Exhibit E		Form of Subordination Agreement
Exhibit F	_	Form of Joinder Agreement
Exhibit G	_	Form of Assignment and Acceptance
Exhibit H	_	Form of Instrument of Accession
		Schedules
Schedule 1 -		Subsidiaries of the Parent
Schedule 2 -		Lenders; Commitment Percentages
Schedule 6.7 -		Litigation
Schedule 6.16 -		Environmental Compliance
Schedule 6.18 -		Certain Transactions
Schedule 6.20(a) -		Series A Holders
Schedule 6.20(b) -		Options, Etc.
Schedule 8.1(c) -		Existing Indebtedness
Schedule 8.2(f) -		Existing Liens
Schedule 8.3(f) -		Existing Investments

SECOND AMENDED AND RESTATED REVOLVING CREDIT AND TERM LOAN AGREEMENT

This **SECOND AMENDED AND RESTATED REVOLVING CREDIT AND TERM LOAN AGREEMENT** (this "<u>Credit Agreement</u>") is made as of the 24th day of January, 2003 by and among (a) **CASELLA WASTE SYSTEMS, INC.**, a Delaware corporation (the "<u>Parent</u>"), its Subsidiaries (other than the Excluded Subsidiaries) listed on <u>Schedule 1</u> hereto (the Parent and such Subsidiaries herein collectively referred to as the "<u>Borrowers</u>"), (b) **FLEET NATIONAL BANK**, ("<u>Fleet</u>"), individually and as administrative agent (in such capacity, the "<u>Administrative Agent</u>"), (c) **BANK OF AMERICA, N.A.** ("<u>BOA</u>"), individually and as syndication agent (in such capacity, the "<u>Syndication Agent</u>" and, together with the Administrative Agent, collectively referred to herein from time to time as the "<u>Agents</u>") and (d) the lending institutions from time to time parties hereto (the "<u>Lenders</u>").

WHEREAS, the Borrowers, the Agents and certain lending institutions (the "Existing Lenders") are parties to that certain Amended and Restated Revolving Credit and Term Loan Agreement, dated as of December 14, 1999 (the "Existing Credit Agreement"), pursuant to which the Existing Lenders have made loans and other extensions of credit (the "Existing Loans") to the Borrowers;

WHEREAS, the Lenders that are not Existing Lenders (the "New Lenders") wish to become parties to this Credit Agreement;

WHEREAS, the Existing Lenders are willing to amend and restate the Original Credit Agreement, the New Lenders are willing to become parties to this Credit Agreement and the Lenders are willing to make loans and other extensions of credit to the Borrowers only on the terms and conditions set forth herein;

NOW, THEREFORE, in consideration of the foregoing, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged (these recitals being an integral part of this Credit Agreement), the Borrowers, Agents and the Lenders hereby agree that, as of the Effective Date (as defined below), the Existing Credit Agreement shall be amended and restated in its entirety and shall remain in full force and effect only as set forth herein:

§1. DEFINITIONS AND RULES OF INTERPRETATION.

<u>§1.1.</u> Definitions. The following terms shall have the meanings set forth in this §1 or elsewhere in the provisions of this Credit Agreement referred to below:

Acceding Lender. See §19(g).

Accountants. See §6.4(a).

Acquired Business. A business acquired by any Borrower, whether through asset or stock purchases, merger, consolidation or otherwise, during the period reported in the most recent financial statements delivered to the Lenders pursuant to §7.4.

Administrative Agent. Fleet acting as administrative agent for the Lenders.

Administrative Agent's Office. The Administrative Agent's office located at 100 Federal Street, Boston, Massachusetts 02110, or at such other location as the Administrative Agent may designate from time to time.

<u>Affiliate</u>. Any Person which, directly or indirectly, controls, is controlled by or is under common control with a Borrower. "Control" of a Borrower means the power, directly or indirectly, (a) to vote ten percent (10%) or more of the capital stock or other equity interests (on a fully diluted basis) of a Borrower having ordinary voting power for the election of directors, managing members or general partners (as applicable); or (b) to direct or cause the direction of the management and policies of a Borrower (whether by contract or otherwise).

Applicable Canadian Pension Legislation. At any time, any pension or retirement benefits legislation (be it federal, provincial, territorial, or otherwise) then applicable to any of the Borrowers or Excluded Subsidiaries, including the Pension Benefits Act (Ontario), the Income Tax Act (Canada), and all regulations thereunder.

Applicable Laws. See §7.10.

<u>Applicable Rate</u>. The applicable rate per annum set forth in the following table:

		Revolving Credit Loans:		Term Loans:		
Level	Pricing Ratio	Applicable Rate for Base Rate Loans	Applicable Rate for Eurodollar Rate Loans	Applicable Rate for Base Rate Loans	Applicable Rate for Eurodollar Rate Loans	Commitment Fee
I	less than 2.75:1.0	Base Rate <u>plus</u> 0.25% per annum	Eurodollar Rate <u>plus</u> 2.25% per annum	Base Rate <u>plus</u> 1.00% per annum	Eurodollar Rate <u>plus</u> 3.00% per annum	0.375%
Π	greater than or equal to 2.75:1.0and less than 3.25:1.0	Base Rate <u>plus</u> 0.50% per annum	Eurodollar Rate <u>plus</u> 2.50% per annum	Base Rate <u>plus</u> 1.00% per annum	Eurodollar Rate <u>plus</u> 3.00% per annum	0.500%
			2			

Ш	greater than or equal to 3.25:1.0 and less than 3.75:1.0	Base Rate <u>plus</u> 0.75% per annum	Eurodollar Rate <u>plus</u> 2.75% per annum	Base Rate <u>plus</u> 1.25% per annum	Eurodollar Rate <u>plus</u> 3.25% per annum	0.500%
IV	greater than or equal to 3.75:1.0 and less than 4.25:1.0	Base Rate <u>plus</u> 1.00% per annum	Eurodollar Rate <u>plus</u> 3.00% per annum	Base Rate <u>plus</u> 1.25% per annum	Eurodollar Rate <u>plus</u> 3.25% per annum	0.500%
V	greater than or equal to 4.25:1.0	Base Rate <u>plus</u> 1.25% per annum	Eurodollar Rate <u>plus</u> 3.25% per annum	Base Rate <u>plus</u> 1.25% per annum	Eurodollar Rate <u>plus</u> 3.25% per annum	0.500%

Each Applicable Rate shall become effective on the first day after receipt by the Lenders of financial statements delivered pursuant to \$7.4(a) or (b) hereof which indicate a change in the Pricing Ratio and in the Applicable Rate in accordance with the above table; <u>provided</u> that, for the period commencing on the Effective Date and ending on the date that the Compliance Certificate is delivered with respect to the second full fiscal quarter of the Borrowers ending after the Effective Date, the Applicable Rate shall be no lower than the rate set forth for Level IV. If at any time the financial statements required to be delivered pursuant to \$7.4(a) or (b) hereof are not delivered within ten (10) days after the time periods specified in such subsections, the Applicable Rate shall be the rate set forth for Level V, subject to prospective adjustment upon actual receipt of such financial statements.

Approved Fund. Any Fund that is administered or managed by (a) a Lender, (b) an affiliate of a Lender or (c) an entity or an affiliate of an entity that administers or manages a Lender.

Assignment and Acceptance. See §19.

Auto-Renewal Letter of Credit. See §3.1(f).

Balance Sheet Date. April 30, 2002.

Base Rate. The higher of (a) the variable annual rate of interest so designated from time to time by Fleet as its "prime rate", such rate being a reference rate and not necessarily representing the lowest or best rate being charged to any customer, and (b) one-half of one percent (0.50%) above the Federal Funds Effective Rate. For the purposes of this definition, "Federal Funds Effective Rate" shall mean for any day, the rate per annum equal to the weighted average of the rates on overnight federal funds transactions with members of the Federal Reserve System arranged by federal funds brokers, as published for such day (or, if such day is not a Business Day, for the next preceding Business Day) by the Federal Reserve Bank of New York, or, if such rate is not so published for any day that is a Business Day, the average of the quotations for such day on such transactions received by the Administrative Agent from three funds brokers of recognized standing selected by the Administrative Agent. Changes in the Base Rate resulting from any changes in Fleet's "prime rate" shall take place immediately without notice or demand of any kind.

Base Rate Loans. Loans bearing interest calculated by reference to the Base Rate.

Benefit Amount. See §5.9(f).

BOA. See preamble.

Borrowers. The Parent and each of its Subsidiaries (other than Excluded Subsidiaries) listed on <u>Schedule 1</u> hereto as "Borrowers", which conduct all or substantially all of their business in the United States or Canada or which are incorporated or otherwise formed under the laws of the United States or a jurisdiction thereof or Canada and which have executed this Credit Agreement as of the Effective Date or have become a party hereto thereafter by executing a Joinder Agreement.

Business Day. Any day on which banking institutions in Boston, Massachusetts are open for the transaction of banking business, and, in the case of Eurodollar Rate Loans, also a day which is a Eurodollar Business Day,

<u>Capital Assets</u>. Fixed assets, both tangible (such as land, buildings, fixtures, machinery and equipment) and intangible (such as patents, copyrights, trademarks, franchises and goodwill); <u>provided</u> that Capital Assets shall not include any item customarily charged directly to expense or depreciated over a useful life of twelve (12) months or less in accordance with GAAP.

<u>Capital Expenditures</u>. Amounts paid or Indebtedness incurred by any Person in connection with (a) the purchase or lease by such Person of Capital Assets that would be required to be capitalized and shown on the balance sheet of such Person in accordance with GAAP or (b) the lease of any assets by such Person as lessee under any Synthetic Lease to the extent that such assets would have been Capital Assets had the Synthetic Lease been treated for accounting purposes as a Capitalized Lease.

<u>Capitalized Leases</u>. Leases under which any Borrower is the lessee or obligor, the discounted future rental payment obligations under which are required to be capitalized on the balance sheet of the lessee or obligor in accordance with GAAP.

Capital Stock. Any and all shares, interests, participations or other equivalents (however designated) of capital stock of a corporation, any and all equivalent ownership interests in a

Person (other than a corporation) and any and all warrants, rights or options to purchase any of the foregoing.

Cellulose Joint Venture. The joint venture between U.S. Fiber and Greenstone Industries, Inc. with respect to the cellulose fibers business, including the manufacturing, marketing and selling of insulation and other cellulose-based products.

<u>Certified</u>. With respect to the financial statements of any Person, such statements as audited by a firm of independent auditors, whose report expresses the opinion, without qualification (including, without limitation, as to the scope of such auditors' review or any going concern qualification), that such financial statements present fairly the financial position of such Person in accordance with GAAP.

CFO. See §7.4(b).

Co-Arrangers. Fleet Securities, Inc. and Banc of America Securities LLC, acting as Co-Arrangers.

Code. The Internal Revenue Code of 1986, as amended and in effect from time to time.

Collateral. All of the property, rights and interests of the Borrowers that are or are intended to be subject to the security interests and mortgages created by the Security Documents.

<u>Commitment</u>. With respect to each Revolving Credit Lender, the amount determined by multiplying such Lender's Commitment Percentage by the Total Commitment, as the same may be increased pursuant to §19(g) or reduced or reallocated from time to time pursuant to the provisions hereof, or if such commitment is terminated pursuant to the provisions hereof, zero.

Commitment Fee. See §5.1(a).

<u>Commitment Percentage</u>. With respect to each Revolving Credit Lender, the percentage set forth beside its name on <u>Schedule 2</u> hereto as the amount of such Revolving Credit Lender's percentage of the Total Commitment (subject to adjustment upon any assignment or accession pursuant to §19).

Compliance Certificate. See §7.4(c).

<u>Consolidated</u> or <u>consolidated</u>. With reference to any term defined herein, shall mean that term as applied to the accounts of the Parent and its Subsidiaries consolidated in accordance with GAAP.

<u>Consolidated Adjusted Net Income</u>. For any period, the Consolidated Net Income (or Loss) of the Parent and its Subsidiaries determined in accordance with GAAP, <u>plus</u>, to the extent deducted and without duplication, (a) adjustments for non-cash write-offs attributable to the use of a fair value methodology for recognition and measurement of impairment of goodwill not identified with impaired assets in accordance with Financial Accounting Standards Board Statement No. 142 up to an aggregate amount of \$62,825,000, (b) charges incurred by the Borrowers in connection with the early termination of interest rate hedging contracts up to an aggregate amount of \$4,000,000, (c) adjustments for non-cash, non-recurring charges related to losses from asset impairment charges resulting from the sale of the Specified Entities or their



assets up to an aggregate amount of \$15,000,000, and (d) the non-recurring, non-cash write-off of debt issuance expenses related to the refinancing of Indebtedness under the Existing Credit Agreement, such write-off not to exceed \$4,000,000.

Consolidated EBITDA. For any period, the Consolidated Adjusted Net Income of the Parent and its Subsidiaries determined in accordance with GAAP, <u>plus</u>, to the extent that such charge was deducted in determining Consolidated Adjusted Net Income in the relevant period and without duplication, (a) interest expense for such period; (b) income taxes for such period; (c) amortization expense for such period; (d) depreciation expense for such period; and, solely for the purpose of determining the Pricing Ratio and calculating the financial covenants set forth in §§9.3 and 9.4, (e) EBITDA of each Acquired Business and New Subsidiary, which in each case shall be included in the calculation of Consolidated EBITDA of the Parent and its Subsidiaries as if such Acquired Business or New Subsidiary was a Subsidiary as of the first day of such period only if (i)(A) the financial statements of such Acquired Business or such New Subsidiary, as the case may be, have been audited for the most recent fiscal year ended of such Acquired Business or such New Subsidiary, a portion of which fiscal year is sought to be included in the calculation of Consolidated EBITDA of the Borrowers, or, (B) if audited financial statements are not available, the Administrative Agent consents to such inclusion after being furnished with other acceptable financial statements, and (ii) a Compliance Certificate and other reasonably appropriate documentation, in form and substance reasonably satisfactory to the Administrative Agent, with respect to the historical operating results, adjustments and balance sheet of such Acquired Business or such New Subsidiary, as the case may be, (which information to the knowledge of the CFO is correct in all material respects) are provided to the Administrative Agent; and (f) non-recurring acquisition-related expenses of an Acquired Business (including compensation payable to former owner(s) of such Acquired Business) in such amounts as are approved by the Administrative Agent.

<u>Consolidated Net Income (or Loss</u>). The consolidated net income (or loss) of the Parent and its Subsidiaries after deduction of all expenses, taxes, and other proper charges determined in accordance with GAAP, <u>less</u>, to the extent included therein, (i) gains from extraordinary items, (ii) any income from discontinued operations, and (iii) income attributable to any minority equity or other Investment in any non-Borrower; <u>provided</u>, <u>however</u>, that consolidated net income shall not be reduced pursuant to this clause (iii) by the aggregate amount of actual cash received by the Borrowers with respect to the Cellulose Joint Venture and the New Heights Investment in the form of cash dividends or cash partnership or limited liability company distributions during the applicable period to the extent that such amount exceeds the aggregate amount of Investments made by the Borrowers in the Cellulose Joint Venture and the New Heights Investment during such period.

<u>Consolidated Net Worth</u>. The excess of Consolidated Total Assets over Consolidated Total Liabilities <u>plus</u>, without duplication, the Liquidation Value (as defined in the Series A Certificate) of the issued and outstanding Series A Preferred Stock <u>less</u>, to the extent otherwise includable in the computations of Consolidated Net Worth, any subscriptions receivable.

<u>Consolidated Senior Funded Debt</u>. At any time of determination, (a) Consolidated Total Funded Debt <u>minus</u> (b) Subordinated Debt outstanding as of such date <u>plus</u> (c) any and all scheduled principal payments in respect of Seller Subordinated Debt that will become due and payable during the next successive period of four fiscal quarters.

Consolidated Total Assets. The sum of all assets ("consolidated balance sheet assets") of the Parent and its Subsidiaries determined on a consolidated basis in accordance with GAAP.

<u>Consolidated Total Funded Debt</u>. At any time of determination with respect to the Parent and its Subsidiaries, collectively, without duplication, whether classified as Indebtedness or otherwise on the consolidated balance sheet of the Parent and its Subsidiaries, (a) the aggregate amount of Indebtedness for (i) borrowed money or credit obtained or other similar monetary obligations, direct or indirect, (including any unpaid reimbursement obligations with respect to letters of credit, but excluding any contingent obligations with respect to letters of credit outstanding), (ii) all obligations evidenced by notes, bonds, debentures or other similar debt instruments (other than Performance Bonds), (iii) the deferred purchase price of assets (other than trade payables incurred in the ordinary course of business), and (iv) all obligations, liabilities and Indebtedness under Capitalized Leases and Synthetic Leases which correspond to principal, <u>plus</u> (b) Indebtedness of the type referred to in clause (a) of another Person guaranteed by the Parent or any of its Subsidiaries.

<u>Consolidated Total Interest Expense</u>. For any period, the aggregate amount of interest expense required to be paid or accrued by the Parent and its Subsidiaries during such period on all Indebtedness of the Parent and its Subsidiaries outstanding during all or any part of such period, whether such interest was or is required to be reflected as an item of expense or capitalized, including payments consisting of interest in respect of any Capitalized Lease or any Synthetic Lease, and including commitment fees, agency fees, balance deficiency fees and similar fees or expenses for such period in connection with the borrowing of money, but excluding therefrom the non-cash amortization of debt issuance costs.

Consolidated Total Liabilities. All liabilities of the Parent and its Subsidiaries determined on a consolidated basis in accordance with GAAP.

Consulting Engineer. An environmental consulting firm acceptable to the Administrative Agent.

<u>Conversion Request</u>. A notice given by the Parent on behalf of the Borrowers to the Administrative Agent of such Borrowers' election to convert or continue a Loan in accordance with §5.11.

Credit Agreement. This Second Amended and Restated Revolving Credit and Term Loan Agreement, including the Schedules and Exhibits hereto.

Default. See §13.

<u>Delinquent Lender</u>. Any Lender that fails (a) to make available to the Administrative Agent its <u>pro rata</u> share of any Loan or to purchase any Letter of Credit participation or (b) to comply with the provisions of $\S14$ with respect to making dispositions and arrangements with the other Lenders, where such Lender's share of any payment received, whether by setoff or otherwise, is in excess of its <u>pro rata</u> share of such payments due and payable to all of the Lenders, in each case as, when and to the full extent required by the provisions of this Credit Agreement, shall be deemed a Delinquent Lender until such time as such delinquency is satisfied.

De Minimis Subsidiaries. Any Subsidiary of the Parent whose assets, liabilities and annual gross revenues do not, in each case, exceed \$1,000,000; provided that the aggregate assets,

liabilities and annual gross revenues of all such Subsidiaries taken as a whole shall not exceed \$2,000,000.

Disposal/Disposed. See the definition of "Release".

Distribution. The declaration or payment of any dividend on or in respect of any shares of any class of Capital Stock of any Person, other than dividends payable solely in shares of common stock of such Person; the purchase, redemption, defeasance, retirement or other acquisition of any shares of any class of Capital Stock of such Person, directly or indirectly through a Subsidiary of such Person or otherwise and whether in the form of increases in the liquidation value of such shares or otherwise (including the setting apart of assets for a sinking or other analogous fund to be used for such purpose); the return of capital by any Person to its shareholders as such; or any other distribution on or in respect of any shares of any class of Capital Stock of such Person.

Dollars or §. Dollars in lawful currency of the United States of America.

Domestic Subsidiary. All Subsidiaries of the Parent which are incorporated or otherwise formed under the laws of the United States or a jurisdiction thereof.

<u>Drawdown Date</u>. The date on which any Loan is made or is to be made, and the date on which any Loan is converted or continued in accordance with §5.11, or the date that any draft or other form of demand for payment is honored with respect to a Letter of Credit.

Effective Date. The first date on which all of the conditions precedent set forth in §10 are satisfied.

Employee Benefit Plan. Any employee benefit plan within the meaning of §3(3) of ERISA maintained or contributed to by any Borrower or any ERISA Affiliate, other than a Guaranteed Pension Plan or a Multiemployer Plan.

Environmental Laws. See §6.16(a).

Equity Offering. The sale or issuance by the Parent of any of its Capital Stock.

ERISA. The Employee Retirement Income Security Act of 1974, as amended and in effect from time to time.

ERISA Affiliate. Any Person which is treated as a single employer with any Borrower under §414 of the Code.

ERISA Reportable Event. A reportable event with respect to a Guaranteed Pension Plan within the meaning of §4043 of ERISA and the regulations promulgated thereunder as to which the requirement of notice has not been waived.

Eurocurrency Reserve Rate. For any day with respect to a Eurodollar Rate Loan, the maximum rate (expressed as a decimal) at which any bank subject thereto would be required to maintain reserves under Regulation D of the Board of Governors of the Federal Reserve System (or any successor or similar regulations relating to such reserve requirements) against "Eurocurrency Liabilities" (as that term is used in Regulation D), if such liabilities were

outstanding. The Eurocurrency Reserve Rate shall be adjusted automatically on and as of the effective date of any change in the Eurocurrency Reserve Rate.

Eurodollar Business Day. Any day on which commercial banks are open for international business (including dealings in Dollar deposits) in London or such other eurodollar interbank market as may be selected by the Administrative Agent in its sole discretion acting in good faith.

Eurodollar Rate. For any Interest Period with respect to a Eurodollar Rate Loan, the rate of interest equal to (a) the arithmetic rate per annum (rounded upwards to the nearest 1/16 of one percent) at which Dollar deposits are offered to the Administrative Agent by prime banks in whatever eurodollar market may be selected by the Administrative Agent in its sole discretion, acting in good faith at or about 10:00 a.m. local time in such interbank market two (2) Eurodollar Business Days prior to the beginning of such Interest Period, for delivery on the first day of such Interest Period for the number of days comprised therein and in an amount comparable to the amount of the Eurodollar Rate Loan to which such Interest Period applies, <u>divided by</u> (b) a number equal to 1.00 minus the Eurocurrency Reserve Rate, if applicable.

Eurodollar Rate Loans. Loans bearing interest calculated by reference to the Eurodollar Rate.

Event of Default. See §13.

Excess Operating Cash Flow. For any period, an amount equal to (a) Consolidated EBITDA for such period <u>plus</u> any increases, or <u>minus</u> any decreases, as the case may be, resulting from Net Working Capital Changes for such period, <u>minus</u> extraordinary cash items of income during such period and <u>plus</u> extraordinary cash items of loss during such period, <u>minus</u> (b) the sum of (i) Capital Expenditures paid in cash to the extent not already deducted in the determination of Consolidated EBITDA, <u>plus</u> (ii) interest expense paid in cash for such period, <u>plus</u> (iii) taxes paid in cash during such period, <u>plus</u> (iv) scheduled principal payments of Indebtedness made during such period, <u>plus</u> (v) cash consideration paid during such period for Permitted Acquisitions. For the purposes of this definition, Excess Operating Cash Flow with respect to the fiscal year ending April 30, 2003, shall be measured from the February 1, 2003 through April 30, 2003, and for each fiscal year thereafter, shall be measured for such fiscal year.

Excluded Subsidiaries. The Insurance Subsidiary and each of the De Minimis Subsidiaries listed on Schedule 1 hereto under the heading "Excluded Subsidiaries".

FCR. FCR, Inc., a Delaware corporation and a wholly-owned Subsidiary of KTI.

<u>Financial Affiliate</u>. A Subsidiary of the bank holding company controlling any Lender, which Subsidiary is engaging in any of the activities permitted by §4(e) of the Bank Holding Company Act of 1956 (12 U.S.C. §1843).

<u>Financial L/C(s)</u>. Letter(s) of Credit where the event which triggers payment is financial, such as the failure to pay money, and not performance related, such as failure to ship a product or provide a service, as set forth in greater detail in the letter dated March 30, 1995 from the Board of Governors of the Federal Reserve System or in any applicable directive or letter ruling of the Board of Governors of the Federal Reserve System issued subsequent thereto.</u>

Financial L/C Fee. See §5.1(b).

Fleet. See preamble.

<u>Foreign Subsidiary</u>. Each Subsidiary of any Borrower (whether direct or indirect, existing on the date hereof or acquired or formed hereafter in accordance with the provisions hereof) which is incorporated under the laws of a jurisdiction other than a State or other jurisdiction of the United States of America.

Fuel Derivatives Obligations. See §8.1(g).

<u>Fund</u>. Any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business.

<u>Generally Accepted Accounting Principles or GAAP</u>. When used in general, Generally Accepted Accounting Principles means principles that are consistent with the principles promulgated or adopted by the Financial Accounting Standards Board and its predecessors, in effect for the fiscal year ended on the Balance Sheet Date, as shall be concurred in by independent certified public accountants of recognized standing whose report expresses an unqualified opinion (other than a qualification regarding changes in Generally Accepted Accounting Principles) as to financial statements in which such principles have been applied; and when used with reference to the Borrowers, such principles shall include (to the extent consistent with such principles) the accounting practices reflected in the consolidated financial statements for the year ended on the Balance Sheet Date.

Greenfiber. U.S. GreenFiber LLC, a Delaware limited liability company in which U.S. Fiber owns a 50% equity interest and through which the Cellulose Joint Venture is conducted.

<u>Guaranteed Pension Plan</u>. Any employee pension benefit plan within the meaning of §3(2) of ERISA maintained or contributed to by any Borrower or any ERISA Affiliate the benefits of which are guaranteed on termination in full or in part by the PBGC pursuant to Title IV of ERISA, other than a Multiemployer Plan.

<u>Hazardous Substances</u>. Any hazardous waste, as defined by 42 U.S.C. §6903(5), any hazardous substances as defined by 42 U.S.C. §9601(14), any pollutant or contaminant as defined by 42 U.S.C. §9601(33) and any waste, hazardous waste, dangerous goods, contaminants, pollutants, toxic substance, oil or hazardous materials or other chemicals or substances regulated by any Environmental Laws.

Indebtedness. As to any Person and whether recourse is secured by or is otherwise available against all or only a portion of the assets of such Person and whether or not contingent, but without duplication:

(a) every obligation of such Person for money borrowed,

(b) every obligation of such Person evidenced by bonds, debentures, notes or other similar instruments, including obligations incurred in connection with the acquisition of property, assets or businesses,

(c) every reimbursement obligation of such Person with respect to letters of credit, bankers' acceptances or similar facilities issued for the account of such Person,

(d) every obligation of such Person issued or assumed as the deferred purchase price of property or services (including securities repurchase agreements but excluding (x) trade accounts payable or accrued liabilities arising in the ordinary course of business which are not overdue in accordance with their terms or the Borrowers' normal or ordinary business practices or which are being contested in good faith and (y) contingent royalty payments made in connection with the purchase or operation of landfills and other types of disposal facilities),

(e) every obligation of such Person under any Capitalized Lease,

(f) every obligation of such Person under any Synthetic Lease,

(g) all sales by such Person of (i) accounts or general intangibles for money due or to become due, (ii) chattel paper, instruments or documents creating or evidencing a right to payment of money or (iii) other receivables (collectively "<u>receivables</u>"), whether pursuant to a purchase facility or otherwise, other than in connection with the disposition of the business operations of such Person relating thereto or a disposition of defaulted receivables for collection and not as a financing arrangement, and together with any obligation of such Person to pay any discount, interest, fees, indemnities, penalties, recourse, expenses or other amounts in connection therewith,

(h) every obligation of such Person (an "<u>equity related purchase obligation</u>") to purchase, redeem, retire or otherwise acquire for value any shares of capital stock of any class issued by such Person, any warrants, options or other rights to acquire any such shares, or any rights measured by the value of such shares, warrants, options or other rights,

(i) every obligation of such Person under any forward contract, futures contract, swap, option or other financing agreement or arrangement (including, without limitation, caps, floors, collars and similar agreements), the value of which is dependent upon interest rates, currency exchange rates, commodities or other indices (a "derivative contract"),

(j) every obligation in respect of Indebtedness of any other entity (including any partnership in which such Person is a general partner) to the extent that such Person is liable therefor as a result of such Person's ownership interest in or other relationship with such entity, except to the extent that the terms of such Indebtedness provide that such Person is not liable therefor and such terms are enforceable under applicable law,

(k) every obligation, contingent or otherwise, of such Person guaranteeing, or having the economic effect of guarantying or otherwise acting as surety for, any obligation of a type described in any of clauses (a) through (j) (the "primary obligation") of another Person (the "<u>primary</u> <u>obligor</u>"), in any manner, whether directly or indirectly, and including, without limitation, any obligation of such Person (i) to purchase or pay (or advance or supply funds for the purchase of) any security for the payment of such primary obligation, (ii) to purchase property, securities or services for the purpose of assuring the payment of such primary obligation, or (iii) to maintain working capital,

equity capital or other financial statement condition or liquidity of the primary obligor so as to enable the primary obligor to pay such primary obligation.

The "amount" or "principal amount" of any Indebtedness at any time of determination represented by (t) any Indebtedness, issued at a price that is less than the principal amount at maturity thereof, shall be the amount of the liability in respect thereof determined in accordance with GAAP, (u) any Capitalized Lease shall be the principal component of the aggregate of the rentals obligation under such Capitalized Lease payable over the term thereof that is not subject to termination by the lessee, (v) any sale of receivables shall be the amount of unrecovered capital or principal investment of the purchaser (other than the Borrower or any of its wholly-owned Subsidiaries) thereof, excluding amounts representative of yield or interest earned on such investment, (w) any Synthetic Lease shall be the stipulated loss value, termination value or other equivalent amount, (x) any derivative contract shall be the maximum amount of any termination or loss payment required to be paid by such Person if such derivative contract were, at the time of determination, to be terminated by reason of any event of default or early termination event thereunder, whether or not such event of default or early termination event than in faced redemption or purchase price thereof that is payable upon a mandatory redemption or purchase of such equity inclusive of any accrued and unpaid dividends to be comprised in such redemption or purchase price and (z) any guaranty or other contingent liability referred to in clause (k) shall be an amount equal to the stated or determinable amount of the primary obligation in respect of which such guaranty or other contingent obligation is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof (assuming such Person is required to perform thereunder) as determinable, the maximum reasonably anticipated liability in this paragraph.

Indenture. The Indenture dated as of January 24, 2003 among the Parent, certain of its Subsidiaries as guarantors and U.S. Bank National Association as trustee, with respect to the Senior Subordinated Notes.

Insurance Subsidiary. Casella Insurance Company, a Vermont corporation and a wholly-owned Subsidiary of the Parent.

Interest Payment Date. (a) As to any Base Rate Loan, the last Business Day of each calendar quarter with respect to interest accrued during such calendar quarter, including, without limitation, the calendar quarter which includes the Drawdown Date of such Base Rate Loan; and (b) as to any Eurodollar Rate Loan in respect of which the Interest Period is (i) 3 months or less, the last day of such Interest Period and (ii) more than 3 months, the date that is 3 months from the first day of such Interest Period and, in addition, the last day of such Interest Period.

Interest Period. With respect to each Revolving Credit Loan or all or any relevant portion of the Term Loan, (a) initially, the period commencing on the Drawdown Date of such Loan and ending on the last day of one of the periods set forth below, as selected by the Borrowers in a Loan and Letter of Credit Request or as otherwise required by the terms of this Credit Agreement (i) for any Base Rate Loan, the last day of the calendar quarter; and (ii) for any Eurodollar Rate Loan, 1, 2, 3, or 6 months; and (b) thereafter, each period commencing on the last day of the next preceding applicable Interest Period, and ending on the last day of one of the periods set forth above in clause (i) or (ii), as selected by the Borrowers in a Conversion Request or as otherwise required pursuant to the provisions of this Credit Agreement; provided that the foregoing provisions relating to Interest Periods are subject to the following:

(A) if any Interest Period with respect to any Eurodollar Rate Loan would otherwise end on a day that is not a Eurodollar Business Day, that Interest Period shall be extended to the next succeeding Eurodollar Business Day unless the result of such extension would be to carry such Interest Period into another calendar month, in which event such Interest Period shall end on the immediately preceding Eurodollar Business Day;

(B) if the Borrowers shall fail to give notice as provided in \$5.11, the Borrowers shall be deemed to have requested a conversion of the affected Eurodollar Rate Loan to a Base Rate Loan and the continuance of all Base Rate Loans as Base Rate Loans on the last day of the then current Interest Period with respect thereto;

(C) any Interest Period relating to any Eurodollar Rate Loan that begins on the last Eurodollar Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Eurodollar Business Day of a calendar month; and

(D) any Interest Period that would otherwise extend beyond the Revolving Credit Maturity Date (if comprising a Revolving Credit Loan) or the Term Loan Maturity Date (if comprising the Term Loan or a portion thereof) shall end on the Revolving Credit Maturity Date or (as the case may be) the Term Loan Maturity Date.

Investments. All expenditures made and all liabilities incurred (contingently or otherwise) for the acquisition of stock or Indebtedness of, or the amount of loans, advances, capital contributions or transfers of property to, or in respect of any guarantees (or other commitments as described under Indebtedness), or obligations of, any Person. In determining the aggregate amount of Investments outstanding at any particular time: (a) the amount of any Investment represented by a guaranty shall be taken at not less than the principal amount of the obligations guaranteed and still outstanding; (b) there shall be included as an Investment all interest accrued with respect to Indebtedness constituting an Investment unless and until such interest is paid; (c) there shall be deducted in respect of each such Investment any amount received as a return of capital (but only by repurchase, redemption, retirement, repayment, liquidating dividend or liquidating distribution); (d) there shall not be deducted in respect of any Investment any amounts received as earnings on such Investment, whether as dividends, interest or otherwise, except that accrued interest included as provided in the foregoing clause (b) may be deducted when paid; and (e) there shall not be deducted or (as the case may be) added from the aggregate amount of Investments any decrease or increase in the value thereof.

Issuance Fee. See §5.1(b).

Issuing Lender. Fleet.

Joinder Agreement. See §7.19.

KTI. KTI, Inc., a New Jersey corporation and a wholly-owned Subsidiary of the Parent.

Lender Affiliate. With respect to any Lender, (a) an affiliate of such Lender or (b) any Approved Fund.

Lenders. The lending institutions listed on <u>Schedule 2</u> hereto and any other Person who becomes an assignee of any rights and obligations of a Lender or becomes a Lender pursuant to §19.

Letters of Credit. See §3.1(a).

Letter of Credit Applications. Letter of Credit Applications in such form as may be agreed upon by any Borrower and the Issuing Lender from time to time which are entered into pursuant to §3 hereof as such Letter of Credit Applications are amended, varied or supplemented from time to time.

Letter of Credit Participation. See §3.1(b).

Letter of Credit Percentage. The percentage per annum equal to the margin above the Eurodollar Rate charged on Revolving Credit Loans that are Eurodollar Rate Loans, as in effect from time to time, as set forth in the column "Applicable Rate for Eurodollar Rate Loans" in the table set forth in the definition of "Applicable Rate" above.

Loan and Letter of Credit Request. See §2.6.

Loan Documents. This Credit Agreement, the Notes, the Letter of Credit Applications, the Letters of Credit, the Security Documents, the Subordination Agreements, and any documents, instruments or agreements executed in connection with any of the foregoing, each as amended, modified, supplemented, or replaced from time to time.

Loans. The Revolving Credit Loans, the Swing Line Loans and the Term Loan.

Loan Percentage. With respect to each Lender as of a particular date, such Lender's portion of, and participating interests in, (calculated as a percentage) the sum of (a) the outstanding principal amount of the Revolving Credit Loans on such date, (b) the outstanding principal amount of the Term Loan on such date, (c) the outstanding principal amount of the Swing Line Loans on such date, (d) the Maximum Drawing Amount of Letters of Credit and any Unpaid Reimbursement Obligations outstanding on such date and, (e) with respect to the definition of Required Lenders and §15.3 only, the unused Commitments on such date.

Material Acquisition. See §8.4.1(f).

Maximum Drawing Amount. The maximum aggregate amount that beneficiaries may at any time draw under outstanding Letters of Credit, as such aggregate amount may be reduced from time to time pursuant to the terms of such Letters of Credit.

Moody's. Moody's Investors Services, Inc.

Multiemployer Plan. Any multiemployer plan within the meaning of §3(37) of ERISA maintained or contributed to by any Borrower or any ERISA Affiliate.

<u>Net Cash Proceeds</u>. With respect to (a) any sale of any assets of the Borrowers or the Excluded Subsidiaries, the gross consideration received by any of the Borrowers or any of the Excluded Subsidiaries (in cash) from such sale, net of commissions, direct sales costs, normal closing adjustments, the amount used to repay any Indebtedness secured by such assets, income taxes attributable to such sale and professional fees and expenses incurred directly in connection

therewith, to the extent the foregoing are actually paid in connection with such sale and (b) any permitted debt offering of the Borrowers or the Excluded Subsidiaries, the gross consideration received by any of the Borrowers or any of the Excluded Subsidiaries (in cash) from such debt offering, net of reasonable and customary transaction expenses and fees actually incurred in connection with such debt offering.

<u>Net Equity Proceeds</u>. With respect to any Equity Offering, the excess of the gross cash proceeds received by such Person from such Equity Offering after deduction of reasonable and customary transaction expenses (including without limitation, underwriting discounts and commissions and reasonable legal fees) actually incurred in connection with such Equity Offering.

Net Working Capital Changes. With respect to the Parent and its Subsidiaries, for any fiscal period and without duplication, the difference (expressed as a positive or a negative number) of (a) the sum of (i) billed accounts receivable, <u>plus</u> (ii) inventory and other current assets considered part of working capital in accordance with GAAP, <u>minus</u> (iii) current accounts payable, <u>minus</u> (iv) current accruals and accretions (exclusive of interest accruals and accretions), in each case, as of the last day of such fiscal period, <u>minus</u> (b) the sum of (i) billed accounts payable, <u>minus</u> (iv) current accruals and accretions (exclusive of interest accruals and cordinate period, <u>minus</u> (b) the sum of (i) billed accounts payable, <u>minus</u> (iv) current accruals and accretions (exclusive of interest accruals and accretions), in each case, as of the last day of such fiscal period, <u>minus</u> (iii) current accounts payable, <u>minus</u> (iv) current accruals and accretions (exclusive of interest accruals and accretions), in each case, as of the last day of immediately preceding fiscal period.

New Heights. New Heights Investor Co., LLC, a Delaware limited liability company in which Casella NH Power Co., LLC owns 100% of the Class B common stock and Casella NH Investors Co., LLC owns 19.9% of the Class A common stock, and each of its direct and indirect Subsidiaries.

New Heights Investment. The Investments made by the Borrowers in New Heights.

<u>New Subsidiary</u>. A Subsidiary acquired or formed by any Borrower or any of its Subsidiaries during the period reported in the most recent financial statements delivered to the Lenders pursuant to §7.4.

Nonrenewal Notice Date. See §3.1(f).

Non-U.S. Lender. See §5.2(c).

Notes. Collectively, the Revolving Credit Notes, the Term Notes, and the Swing Line Note.

Obligations. All Indebtedness, obligations and liabilities of the Borrowers to any of the Lenders, the Agents, and the Issuing Lender, individually or collectively, existing on the date of this Credit Agreement or arising thereafter, direct or indirect, joint or several, absolute or contingent, matured or unmatured, liquidated or unliquidated, secured or unsecured, arising by contract, operation of law or otherwise, in each case arising or incurred under this Credit Agreement or any of the tother Loan Documents or in respect of any of the Loans made or Reimbursement Obligations incurred or the Letters of Credit, the Notes, Swap Contracts, Fuel Derivatives Obligations and similar agreements or arrangements provided by any of the Lenders or any other instrument at any time evidencing any thereof.

Parent. See Preamble.

Participant. See §19(b).

PBGC. The Pension Benefit Guaranty Corporation created by §4002 of ERISA and any successor entity or entities having similar responsibilities.

Performance Bonds. See §8.1(d).

<u>Performance L/C</u>. A Letter of Credit which is not a Financial L/C.

Performance L/C Fee. See §5.1(b).

Permitted Acquisition. See §8.4.1.

Permitted Liens. See §8.2.

<u>Person</u>. Any individual, corporation, limited liability company, partnership, limited liability partnership, trust, unincorporated association, business, or other legal entity, and any government or any governmental agency or political subdivision thereof.

Pledge Agreement. The Pledge Agreement, dated as of the Effective Date, among certain of the Borrowers and the Administrative Agent.

Post-Closing Increase. See §19(g).

<u>Pricing Ratio</u>. At the end of any fiscal quarter of the Borrowers, the ratio of (a) Consolidated Total Funded Debt as of the end of such fiscal quarter to (b) Consolidated EBITDA for the period of four (4) consecutive fiscal quarters then ended, as properly calculated on the Compliance Certificate delivered by the Borrowers pursuant to \$7.4(c).

Real Property. All real property heretofore, now, or hereafter owned or leased by the Borrowers.

Recovery Technology Group. RTG Holdings Corporation, a corporation, in which Casella RTG Investors Co., LLC owns a 19.9% equity interest, and each of its direct and indirect Subsidiaries.

<u>Reimbursement Obligations</u>. The Borrowers' joint and several obligations to reimburse the Issuing Lender and the Revolving Credit Lenders on account of any drawing under any Letter of Credit as provided in §3.2.

<u>Release</u>. Shall mean the broader of (i) the meaning specified for the term "Release" (or "Released") in the Comprehensive Environmental Response, Compensation and Liability Act of 1980, 42 U.S.C. §§9601 <u>et seq</u>. ("CERCLA") and (ii) the meaning specified for the term "<u>Disposal</u>" (or "<u>Disposed</u>") in the Resource Conservation and Recovery Act of 1976, 42 U.S.C. §§6901 <u>et seq</u>. ("RCRA") and regulations promulgated thereunder; provided, that in the event either CERCLA or RCRA is amended so as to broaden the meaning of any term defined thereby, such broader meaning shall apply as of the effective date of such amendment and provided further, to the extent that the laws of a state or province (or the federal laws of Canada applicable therein) wherein the property lies establishes a meaning for "Release" or "Disposal" or any

analogous term which is broader than specified in either CERCLA or RCRA, such broader meaning shall apply.

Required Lenders. As of any date, the Lenders whose aggregate percentages constitute at least fifty-one percent (51%) of the Loan Percentages, provided that for purposes of this definition "Lender" shall not include any Delinquent Lender.

<u>Restricted Payment</u>. In relation to the Borrowers and the Excluded Subsidiaries, any (a) Distribution, (b) payment or prepayment by any Borrower or any Subsidiary to (i) such Borrowers' or such Subsidiaries shareholders (or other equity holders), in each case, other than to another Borrower, or (ii) to any Affiliate of such Borrower or such Subsidiary or any Affiliate of such Borrower's or such Subsidiary's shareholders (or other equity holders), in each case, other than to another Borrower or (c) derivatives or other transactions with any financial institution, commodities or stock exchange or clearinghouse (a "<u>Derivatives Counterparty</u>") obligating such Borrower or such Subsidiary to make payments to such Derivatives Counterparty as a result of any change in market value of any capital stock of such Borrower or such Subsidiary.

<u>Revolving Credit Lenders</u>. The Lenders set forth on <u>Schedule 2</u> as Revolving Credit Lenders, acting in their role as makers of Revolving Credit Loans or as participants with respect to Letters of Credit.

Revolving Credit Loans. Revolving Credit Loans made or to be made by the Revolving Credit Lenders to the Borrowers pursuant to §2.

<u>Revolving Credit Maturity Date</u>. January 24, 2008; <u>provided</u> that, if, on or before May 1, 2007, the Parent has not either (i) converted a portion of the shares of Series A Preferred Stock into shares of common stock of the Parent so that no more than \$20,000,000 in principal amount of Series A Preferred Stock remains outstanding or (ii) duly effected an amendment, in form and substance satisfactory to the Administrative Agent, to the Series A Certificate, extending the mandatory redemption date of the Series A Preferred Stock to April 24, 2010 or later (with all other terms of the Series A Preferred Stock remaining substantially the same), in either case of clause (i) or clause (ii), as evidenced in a written statement certified by an authorized financial officer of the Parent and delivered to the Agents and the Lenders, then the Revolving Credit Maturity Date shall be May 11, 2007.

Security Agreement. The Second Amended and Restated Security Agreement, dated as of the Effective Date, among the Borrowers and the Administrative Agent.

<u>Security Documents</u>. The Security Agreement, the Pledge Agreement each as amended and in effect from time to time, and any additional documents evidencing or perfecting the Administrative Agent's lien on the assets of the applicable Borrowers for the benefit of the applicable Lenders (including Uniform Commercial Code financing statements).

<u>Seller Subordinated Debt</u>. Indebtedness of the Borrowers (other than the Senior Subordinated Debt) which has been subordinated and made junior to the payment and performance in full in cash of the Obligations, and evidenced as such by a subordination agreement containing subordination provisions substantially in the form of <u>Exhibit E</u> (the "<u>Subordination Agreement</u>") hereto; <u>provided</u> that (a) at the time such Seller Subordinated Debt is incurred, no Default or Event of Default has occurred or would occur as a result of such incurrence, and (b) the documentation evidencing such Seller Subordinated Debt shall have been

delivered to the Administrative Agent and shall contain all of the following characteristics: (i) it shall be unsecured, (ii) it shall bear interest at a rate not to exceed the market rate, (iii) it shall have a final maturity of at least three (3) years, (iv) it shall not require unscheduled principal repayments thereof prior to the maturity date of such debt, (v) if it has any covenants, such covenants (including covenants relating to incurrence of indebtedness) shall be meaningfully less restrictive than those set forth herein, (vi) it shall have no restrictions on the Borrower's ability to grant liens securing indebtedness ranking senior to such Seller Subordinated Debt, (vii) it shall permit the incurrence of senior indebtedness under this Credit Agreement, (viii) it may be cross-accelerated with the Obligations and other senior indebtedness of the Borrowers (but shall not be cross-defaulted except for payment defaults which the senior lenders have not waived) and may be accelerated upon bankruptcy, (ix) it shall provide for the complete, automatic and unconditional release of any and all guarantees of such Seller Subordinated Debt granted by any Borrower in the event of the sale by any Person of such Borrower or the sale by any Person of all or substantially all of such Borrower's assets (including in the case of a foreclosure), (x) it shall provide that (A) upon any payment or distribution of the assets of the Borrowers (including after the commencement of a bankruptcy proceeding) of any kind or character, all of the Obligations (including interest accruing after the commencement of any bankruptcy proceeding at the rate specified for the applicable Obligation, whether or not such interest is an allowable claim in any such proceeding) shall be paid in full in cash prior to any payment being received by the holders of the Seller Subordinated Debt and (B) until all of the Obligations (including the interest described in subclause (A) above) are paid in full in cash, any payment or distribution to which the holders of the Seller Subordinated Debt would be entitled but for the subordination provisions of the type described in clauses (xi) and (xii) hereof shall be made to the holders of the Obligations, (xi) it shall provide that in the event of a payment default under \$13.1(a) or (b) hereof, the Borrowers shall not be required to pay the principal of, or any interest, fees and all other amounts payable with respect to the Seller Subordinated Debt until the Obligations have been paid in full in cash, (xii) it shall provide that in the event of any other Event of Default, the Lenders shall be permitted to block with respect to the Seller Subordinated Debt for a period of 180 days (A) payments of principal, interest, fees and all other amounts payable, and (B) enforcement of remedies for Seller Subordinated Debt in excess of \$1,000,000, and (xiii) it shall acknowledge that none of the provisions outlined in part (b) of this definition can be amended, modified or otherwise altered without the prior written consent of the Required Lenders.

Senior Subordinated Debt. The senior subordinated Indebtedness of the Borrowers evidenced by the Senior Subordinated Debt Documents in the original principal amount of at least \$150,000,000.

Senior Subordinated Debt Documents. The Indenture, the Senior Subordinated Notes and all other documents, instruments and agreements entered into or executed in connection therewith, in each case, subject to terms and conditions satisfactory to the Administrative Agent.

Senior Subordinated Notes. The 9.75% Senior Subordinated Notes due 2013 issued by the Parent pursuant to the Indenture.

Series A Certificate. That certain Certificate of Designation of Series A Convertible Preferred Stock, dated as of August 8, 2000, which sets forth the rights and obligations of the Series A Holders and the Parent with respect to the Series A Preferred Stock.

Series A Holders. The holders of the Series A Preferred Stock listed on Schedule 6.20(a) hereto.

Series <u>A Preferred Stock</u>. The Series A Preferred Stock issued by the Parent to the Series A Holders pursuant to the Series A Certificate in an aggregate principal amount not to exceed \$55,750,000 plus dividends as provided for in the Series A Certificate.

Settlement. With respect to Swing Line Loans, the making or receiving of payments, in immediately available funds, by the Revolving Credit Lenders to or from the Administrative Agent in accordance with §2.8 hereof to the extent necessary to cause each such Lender's actual share of the outstanding amount of the Revolving Credit Loans to be equal to such Lender's Commitment Percentage of the outstanding amount of such Revolving Credit Loans, in any case when, prior to such action, the actual share is not so equal.

Settlement Amount. See §2.8(b).

Settlement Date. See §2.8(b).

Settling Lender. See §2.8(b).

S&P. Standard & Poor's Ratings Services, a division of The McGraw-Hill Companies, Inc., or any successor thereto.

Specified Entities. (a) K-C International, Ltd., (b) the brokerage business of KTI Recycling of New England, Inc., (c) the brokerage business of Pine Tree Waste, Inc., (d) Greenfiber, (e) KTI New Jersey Fibers, Inc., (f) Atlantic Coast Fibers, Inc., (g) Casella NH Investors Co., LLC, (h) Casella NH Power Co., LLC, (i) Casella RTG Investors Co., LLC and the Recovery Technology Group and (j) the companies and assets comprising the FCR operating segment, or the successors of any of the foregoing only with respect to the businesses conducted by the foregoing on the Effective Date.

Spot Rate. With respect to any "first currency" (as defined in §5.10), at any date of determination thereof, the spot rate of exchange in London that appears on the display page applicable to such first currency on the Reuters System (or such other page as may replace such page on such service for the purpose of displaying the spot rate of exchange in London) for the conversion of such first currency into the "second currency" (as defined in §5.10); provided, however, that if there shall at any time no longer exist such a page on such service, the Spot Rate shall be determined by reference to another similar rate publishing service selected by the Administrative Agent.

Subsidiary. Any corporation, limited liability company, partnership, association, trust, or other business entity of which the designated parent shall at any time own directly or indirectly through a Subsidiary or Subsidiaries at least a majority of the outstanding Capital Stock or other interest entitled to vote generally.

Subordinated Debt. Collectively, the Senior Subordinated Debt and the Seller Subordinated Debt.

Subordination Agreements. See definition of "Seller Subordinated Debt".

Swap Contracts. Any agreement (including any master agreement and any agreement, whether or not in writing, relating to any single transaction) that is an interest rate swap agreement, basis swap, forward rate agreement, commodity swap, commodity option, equity or

equity index swap or option, bond option, interest rate option, forward foreign exchange agreement, rate cap, collar or floor agreement, currency swap agreement, cross-currency rate swap agreement, swaption, currency option or other similar agreement (including any option to enter into any of the foregoing).

Swing Line Loans. See §2.8(a).

Swing Line Note. See §2.8(a).

Synthetic Lease. Any lease of goods or other property, whether real or personal, which is treated as an operating lease under GAAP and as a loan or financing for U.S. income tax purposes.

<u>Term Loan</u>. The term loan made or to be made by the Term Loan Lenders to the Borrowers pursuant to \$4 in the original principal amount of \$150,000,000, as the same may be increased pursuant to \$19(g) or reduced or reallocated from time to time pursuant to the provisions hereof.

Term Loan Lenders. The Lenders holding a portion of the Term Loan as set forth on <u>Schedule 2</u> hereto together with any other Person who becomes an assignee of any rights and obligations of a Term Loan Lender pursuant to §19.

Term Loan Maturity Date. January 24, 2010; provided that, if, on or before May 1, 2007, the Parent has not either (i) converted a portion of the shares of Series A Preferred Stock into shares of common stock of the Parent so that no more than \$20,000,000 in principal amount of Series A Preferred Stock remains outstanding or (ii) duly effected an amendment, in form and substance satisfactory to the Administrative Agent, to the Series A Certificate, extending the mandatory redemption date of the Series A Preferred Stock to April 24, 2010 or later (with all other terms of the Series A Preferred Stock remaining substantially the same), in either case of clause (i) or clause (ii), as evidenced in a written statement certified by an authorized financial officer of the Parent and delivered to the Agents and the Lenders, then the Term Loan Maturity Date shall be May 11, 2007.

Term Loan Percentage. With respect to each Term Loan Lender, the percentage set forth on <u>Schedule 2</u> (subject to adjustment in accordance with §19 hereof) as such Lender's percentage of the Term Loan.

Term Notes. See §4.2.

Term Note Record. A record with respect to a Term Note.

Total Commitment. The sum of the Commitments of the Lenders, as in effect from time to time, which amount shall initially equal \$175,000,000, as such amount may be reduced or increased pursuant to the terms hereof.

Type. As to any Loan, its nature as a Base Rate Loan or a Eurodollar Rate Loan.

Unpaid Reimbursement Obligation. Any Reimbursement Obligation for which the Borrowers have not reimbursed the Issuing Lender and the Lenders on the date specified in, and

in accordance with, §3.2, which has not been automatically converted into a Revolving Credit Loan pursuant to such section.

U.S. Fiber. U.S. Fiber, Inc., a North Carolina corporation.

§1.2. Rules of Interpretation.

(a) A reference to any document or agreement shall include such document or agreement as amended, modified or supplemented from time to time in accordance with its terms and the terms of this Credit Agreement.

- (b) The singular includes the plural and the plural includes the singular.
- (c) A reference to any law includes any amendment or modification to such law.
- (d) A reference to any Person includes its permitted successors and permitted assigns.

(e) Accounting terms not otherwise defined herein have the meanings assigned to them by GAAP applied on a consistent basis by the accounting entity to which they refer.

(f) The words "include", "includes" and "including" are not limiting.

(g) All terms not specifically defined herein or by GAAP, which terms are defined in the Uniform Commercial Code as in effect in the Commonwealth of Massachusetts have the meanings assigned to them therein, with the term "<u>instrument</u>" being that defined under Article 9 of the Uniform Commercial Code.

(h) Reference to a particular "§" refers to that section of this Credit Agreement unless otherwise indicated.

(i) The words "herein", "hereof", "hereunder" and words of like import shall refer to this Credit Agreement as a whole and not to any particular section or subdivision of this Credit Agreement.

(j) Unless otherwise expressly indicated, in the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including," the words "to" and "until" each mean "to but excluding," and the word "through" means "to and including."

(k) This Credit Agreement and the other Loan Documents may use several different limitations, tests or measurements to regulate the same or similar matters. All such limitations, tests and measurements are, however, cumulative and are to be performed in accordance with the terms thereof.

(1) This Credit Agreement and the other Loan Documents are the result of negotiation among, and have been reviewed by counsel to, among others, the Agents and the Borrower and are the product of discussions and negotiations among all parties. Accordingly, this Credit Agreement and the other Loan Documents are not intended to be

construed against the Agents or any of the Lenders merely on account of either Agent's or any Lender's involvement in the preparation of such documents.

§2. THE REVOLVING CREDIT LOANS.

§2.1. Commitment to Lend. Subject to the terms and conditions set forth in this Credit Agreement, each of the Revolving Credit Lenders severally agrees to lend to the Borrowers, and the Borrowers may borrow, repay, and reborrow from time to time between the Effective Date and the Revolving Credit Maturity Date, upon notice to the Administrative Agent given in accordance with §2.6, such Lender's Commitment Percentage of such sums as are requested by the Borrowers in the minimum aggregate amount of \$500,000 or an integral multiple thereof; provided that, except as otherwise provided herein, the sum of the outstanding amount of Revolving Credit Loans (including the Swing Line Loans and after giving effect to all amounts requested) plus the Maximum Drawing Amount and all Unpaid Reimbursement Obligations shall not at any time exceed the Total Commitment. Revolving Credit Loans made hereunder shall be made *pro rata* in accordance with each Revolving Credit Lender's Commitment Percentage. Each request for a Loan or Letter of Credit hereunder shall constitute a representation and warranty by the Borrowers that the conditions set forth in §10 and §11, as the case may be, have been satisfied on the date of such request.

§2.2. Reduction of Total Commitment. The Borrowers shall have the right at any time and from time to time upon five (5) Business Days' prior written notice to the Administrative Agent to reduce by \$5,000,000 or an integral multiple of \$1,000,000 in excess thereof or terminate entirely the Total Commitment, whereupon the Commitments of the Revolving Credit Lenders shall be reduced *pro rata* in accordance with their respective Commitment Percentages of the amount specified in such notice or, as the case may be, terminated; provided that at no time may the Total Commitment be reduced to an amount less than the sum of (a) the Maximum Drawing Amount and all Unpaid Reimbursement Obligations plus (b) all Revolving Credit Loans (including Swing Line Loans) then outstanding. No reduction or termination of the Total Commitment or the may not be revoked; the portion of the Total Commitment reduced or terminated may not be reinstated; and amounts in respect of such reduced or terminated portion may not be reborrowed. The Administrative Agent will notify the Revolving Credit Lenders promptly after receiving any notice of the Borrowers delivered pursuant to this §2.2. and will distribute to each such Lender a revised schedule of Commitments and Commitment Percentages.

§2.3. The Revolving Credit Notes. The Revolving Credit Loans shall be evidenced by promissory notes of the Borrowers in substantially the form of Exhibit A-1 hereto (each a "Revolving Credit Note"), dated as of the Effective Date (or such later date as a Revolving Credit Lender becomes a party hereto pursuant to §19) and completed with appropriate insertions. One Revolving Credit Note shall be payable to the order of each Revolving Credit Lender in a principal amount equal to such Revolving Credit Lender's Commitment or, if less, the outstanding amount of all Revolving Credit Loans made by such Revolving Credit Lender, plus interest accrued thereon, as set forth below. The Borrowers irrevocably authorize the Revolving Credit Lenders to make, or cause to be made, in connection with a Drawdown Date of any Revolving Credit Loan at the time of receipt of any payment of principal on any such Revolving Credit Note, an appropriate notation on such Lender's records or on the schedule attached to such

Lender's Revolving Credit Note or a continuation of such schedule attached thereto reflecting the making of such Loan, or the receipt of such payment (as the case may be) and may, prior to any transfer of its Revolving Credit Note, endorse on the reverse side thereof the outstanding principal amount of such Revolving Credit Loans evidenced thereby. The outstanding amount of the Revolving Credit Loans set forth on such Lender's record shall be <u>prima facie</u> evidence of the principal amount thereof owing and unpaid to such Lender, but the failure to record, or any error in so recording, any such amount shall not limit or otherwise affect the obligations of the applicable Borrowers hereunder or under such Revolving Credit Notes to make payments of principal of or interest on any such Revolving Credit Notes when due.

§2.4. Interest on Revolving Credit Loans; Maturity of the Revolving Credit Loans.

(a) The Borrowers jointly and severally promise to pay interest on each Revolving Credit Loan in arrears on each Interest Payment Date with respect thereto at the following rates, except as otherwise provided in §5.6:

- (i) Each Revolving Credit Loan which is a Base Rate Loan shall bear interest for the period commencing with the Drawdown Date thereof and ending on the last day of the Interest Period with respect thereto at the Applicable Rate for Revolving Credit Loans that are Base Rate Loans as in effect from time to time.
- (ii) Each Revolving Credit Loan which is a Eurodollar Rate Loan shall bear interest for the period commencing with the Drawdown Date thereof and ending on the last day of the Interest Period with respect thereto at the Applicable Rate for Revolving Credit Loans that are Eurodollar Rate Loans as in effect from time to time.

(b) The Borrowers jointly and severally promise to pay on the Revolving Credit Maturity Date all Revolving Credit Loans outstanding, Unpaid Reimbursement Obligations with respect to Letters of Credit and any and all unpaid interest accrued thereon. The Revolving Credit Loans shall become absolutely due and payable on the Revolving Credit Maturity Date, as set forth above.

§2.5. Mandatory Repayments of the Revolving Credit Loans. If at any time the sum of the outstanding amount of the Revolving Credit Loans (including Swing Line Loans) <u>plus</u> the Maximum Drawing Amount and any Unpaid Reimbursement Obligations exceeds the Total Commitment, whether by reduction of the Total Commitment or otherwise, then the Borrowers, jointly and severally, shall immediately pay the amount of such excess to the Administrative Agent for application: <u>first</u>, to any Unpaid Reimbursement Obligations; <u>second</u>, to the Revolving Credit Loans; and <u>third</u>, to provide to the Administrative Agent cash collateral for Reimbursement Obligations as contemplated by §3.2(b) and (c); <u>provided</u>, <u>however</u>, that if the amount of cash collateral held by the Administrative Agent pursuant to this §2.5 exceeds the amount required to be cash collateralized from time to time, the Administrative Agent shall return such excess to the Borrowers. Each payment of Unpaid Reimbursement Obligations or prepayment of Revolving Credit Loans shall be allocated among the Lenders, in proportion, as nearly as practicable, to each Unpaid Reimbursement Obligation or (as the case may be) the

respective unpaid principal amount of each Lender's Revolving Credit Note, with adjustments to the extent practicable to equalize any prior payments or repayments not exactly in proportion.

§2.6. Requests for Revolving Credit Loans. The Borrowers shall give to the Administrative Agent written notice in the form of Exhibit B hereto (or telephonic notice confirmed by telecopy on the same Business Day in the form of Exhibit B hereto) of each Revolving Credit Loan requested hereunder (a "Loan and Letter of Credit Request") not later than 11:00 a.m. Boston time (a) no less than one (1) Business Day prior to the proposed Drawdown Date of any Base Rate Loan and (b) no less than three (3) Eurodollar Business Days prior to the proposed Drawdown Date of any Eurodollar Rate Loan. Each such notice shall specify (i) the amount of such Revolving Credit Loan, (ii) the proposed Drawdown Date of such Revolving Credit Loan, (iii) the Type of such Revolving Credit Loan, (iv) the Interest Period for such Revolving Credit Loan (if a Eurodollar Rate Loan), and (v) the aggregate outstanding amount of all Revolving Credit Loans (including Swing Line Loans) after giving affect to all amounts requested and the aggregate Maximum Drawing Amount of all outstanding Letters of Credit. Each Revolving Credit Loan requested shall be in a minimum amount of \$5,000,000, or in \$1,000,000 additional increments thereof. Revolving Credit Loan requested from the Revolving Credit Leaner son the proposed Drawdown Date.

§2.7. Funds for Revolving Credit Loans.

(a) Not later than 2:00 p.m. (Boston time) on the proposed Drawdown Date of any Revolving Credit Loan, each of the Revolving Credit Lenders will make available to the Administrative Agent, at the Administrative Agent's Office, in immediately available funds, the amount of such Lender's Commitment Percentage of the amount of the requested Revolving Credit Loans. Upon receipt from each Lender of such amount, and upon receipt of the documents required by §§10 and 11 and the satisfaction of the other conditions set forth therein, to the extent applicable, the Administrative Agent will make available to the Borrowers in immediately available funds the aggregate amount of such Revolving Credit Loans made available to the Administrative Agent by the Revolving Credit Lenders. The failure or refusal of any Revolving Credit Lender to make available to the Administrative Agent at the aforesaid time and place on any Drawdown Date the amount of its Commitment Percentage of the requested Revolving Credit Lender from its several obligation hereunder to make available to the Administrative Agent the amount of such other Revolving Credit Lender's Commitment Percentage of any requested Revolving Credit Loans.

(b) The Administrative Agent may, unless notified to the contrary by any Revolving Credit Lender prior to a Drawdown Date, assume that such Lender has made available to the Administrative Agent on such Drawdown Date the amount of such Lender's Commitment Percentage of the Revolving Credit Loans to be made on such Drawdown Date, and the Administrative Agent may (but shall not be required to), in reliance upon such assumption, make available to the Borrowers a corresponding amount. If any Revolving Credit Lender makes available to the Administrative Agent such amount on a date after such Drawdown Date, such Lender shall pay to the Administrative

Agent on demand an amount equal to the product of (i) the average computed for the period referred to in clause (iii) below, of the weighted average interest rate paid by the Administrative Agent for federal funds acquired by the Administrative Agent during each day included in such period, times (ii) the amount of such Lender's Commitment Percentage of such Loans, times (iii) a fraction, the numerator of which is the number of days that elapse from and including such Drawdown Date until the date on which the amount of such Lender's Commitment Percentage of such Loans, the demoninator of which is 365. A statement of the Administrative Agent submitted to such Lender with respect to any amounts owing under this paragraph shall be <u>prima facie</u> evidence, absent manifest error, of the amount due and owing to the Administrative Agent by such Lender. If the amount of such Lender's Commitment Percentage of such Loans is not made available to the Administrative Agent within three (3) Business Days following such Drawdown Date, the Administrative Agent shall be entitled to recover such amount from the Borrowers on demand, with interest thereon at the Applicable Rate for Revolving Credit Loans of the Type made on such Drawdown Date.

§2.8. Swing Line Loans; Settlements.

(a) Solely for ease of administration of the Revolving Credit Loans, Fleet may, upon receipt of a Loan and Letter of Credit Request requesting a Swing Line Loan no later than 2:30 p.m. (Boston time) on the proposed date of funding, but shall not be required to, fund Base Rate Loans made in accordance with the provisions of this Credit Agreement (including, without limitation, satisfaction of the conditions set forth in §§10 and 11) for periods not to exceed seven (7) days in any one case, bearing interest at the rate set forth in §2.4(a)(i) for Revolving Credit Loans that are Base Rate Loans (<u>"Swing Line Loans</u>"). The Swing Line Loans shall be evidenced by a promissory note of the Borrowers in substantially the form of <u>Exhibit A-2</u> hereto (the <u>"Swing Line Note</u>") dated as of the Effective Date, and shall each be in a minimum amount of \$100,000 or greater, <u>provided that</u> the outstanding amount of Swing Line Loans advanced by Fleet hereunder shall not exceed \$10,000,000 at any time. Each Revolving Credit Lender shall remain severally and unconditionally liable to fund its *pro rata* share (based upon such Lender's Commitment Percentage of the Swing Line Loans requested, subject to satisfaction of the provisions hereof (including, without limitation, satisfaction of the conditions set forth in §§10 and 11) relating to the making of Swing Line Loans. Prior to each Settlement, all payments or repayments of the principal of, and interest on, Swing Line Loans shall be credited to the account of Fleet.

(b) The Revolving Credit Lenders shall effect Settlements on (i) the Business Day immediately following any day which Fleet gives written notice to the Administrative Agent to effect a Settlement, (ii) the Business Day immediately following the Administrative Agent's becoming aware of the existence of any Default or Event of Default, (iii) the Revolving Credit Maturity Date, (iv) any date on which the Borrowers wish to convert a Swing Line Loan into a Eurodollar Rate Loan, and (v) in any event, the seventh day on which any Swing Line Loan remains outstanding (each such date, a "<u>Settlement Date</u>"). One (1) Business Day prior to each such Settlement Date, the Administrative Agent shall give telephonic notice to the Revolving Credit Lenders of (A) the respective outstanding amount of Revolving Credit Loans made by each Revolving

Credit Lender as at the close of business on the prior day, and (B) the amount that any Revolving Credit Lender, as applicable (a "<u>Settling Lender</u>"), shall pay to effect a Settlement (a "<u>Settlement Amount</u>"). A statement of the Administrative Agent submitted to the Revolving Credit Lenders with respect to any amounts owing hereunder shall be <u>prima facie</u> evidence of the amount due and owing. Each Settling Lender shall, not later than 1:00 p.m. (Boston time) on each Settlement Date, effect a wire transfer of immediately available funds to the Administrative Agent at the Administrative Agent's Office in the amount of such Lender's Settlement Amount. All funds advanced by any Revolving Credit Lender as a Settling Lender pursuant to this §2.8 shall for all purposes be treated as a Base Rate Loan to the Borrowers.

(c) The Administrative Agent may (unless notified to the contrary by any Settling Lender by 12:00 noon (Boston time) one (1) Business Day prior to the Settlement Date) assume that each Settling Lender has made available (or will make available by the time specified in §2.8(b)) to the Administrative Agent its Settlement Amount, and the Administrative Agent may (but shall not be required to), in reliance upon such assumption, effect Settlements. If the Settlement Amount of such Settling Lender is made available to the Administrative Agent on a date after such Settlement Date, such Settling Lender shall pay the Administrative Agent on demand an amount equal to the product of (i) the average, computed for the period referred to in clause (iii) below, of the weighted average annual interest rate paid by the Administrative Agent for federal funds acquired by the Administrative Agent during each day included in such period times (ii) such Settlement Amount times (iii) a fraction, the numerator of which is the number of days that elapse from and including such Settlement Date to but not including the date on which such Settling Lender shall be deemed to have delivered its Settlement Amount on the Settlement Date and shall become entitled to interest payable by the Borrowers with respect to such Settling Lender's Settlement Amount as if such share were delivered on the Settlement Date. If such Settlement Amount is not in fact made available to the Administrative Agent by such Settling Lender within five (5) Business Days of such Settlement Date, the Administrative Agent shall be entitled to recover such amount from the Borrowers, with interest thereon at the Applicable Rate for Revolving Credit Loans that are Base Rate Loans.

(d) After any Settlement Date, any payment by the Borrowers of Swing Line Loans hereunder shall be allocated *pro rata* among the Revolving Credit Lenders, in accordance with such Lenders' Commitment Percentages.

(e) If, prior to the making of a Revolving Credit Loan pursuant to paragraph (b) of this §2.8, a Default or Event of Default has occurred and is continuing, each Revolving Credit Lender will, on the date such Revolving Credit Loan was to have been made, purchase an undivided participating interest in the outstanding Swing Line Loans in an amount equal to its Commitment Percentage of such Swing Line Loans. Each Revolving Credit Lender will immediately transfer to the Administrative Agent, in immediately available funds, the amount of its participation and upon receipt thereof the Administrative Agent will deliver to such Revolving Credit Lender a Swing Line participation certificate dated the date of receipt of such funds and in such amount.

(f) Whenever, at any time after the Administrative Agent has received from any Revolving Credit Lender such Lender's participating interest in the Swing Line Loans pursuant to clause (e) above, the Administrative Agent receives any payment on account thereof, the Administrative Agent will distribute to such Lender its participating interest in such amount (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Lender's participating interest was outstanding and funded) in like funds as received; provided, however, that in the event that such payment received by the Administrative Agent is required to be returned, such Lender will return to the Administrative Agent any portion thereof previously distributed by the Administrative Agent to it in like funds as such payment is required to be returned by the Administrative Agent.

(g) Each Revolving Credit Lender's obligation to purchase participating interests pursuant to clause (e) above shall be absolute and unconditional and shall not be affected by any circumstance, including, without limitation, (i) any set-off, counterclaim, recoupment, defense or other right which such Lender may have against the Administrative Agent, the Borrowers or any other Person for any reason whatsoever; (ii) the occurrence or continuance of a Default or Event of Default; (iii) any adverse change in the condition (financial or otherwise) of the Borrowers or any other Person; (iv) any breach of this Credit Agreement by the Borrowers or any other Lender or the Administrative Agent; or (v) any other circumstance, happening or event whatsoever, whether or not similar to any of the foregoing.

§2.9. Optional Prepayments or Repayments of Revolving Credit Loans. The Borrowers shall have the right, at their election, to repay or prepay the outstanding amount of the Revolving Credit Loans, as a whole or in part, at any time without penalty or premium; provided (a) each partial prepayment shall be in the principal amount of \$250,000 or an integral multiple thereof, and (b) that the full or partial prepayment of the outstanding amount of any Eurodollar Rate Loans pursuant to this §2.9, if made on a day other than the last day of the Interest Period relating thereto, shall be subject to the provisions of §5.12. The Borrowers shall give the Administrative Agent, no later than 11:00 a.m. (Boston time) (i) at least one (1) Business Days' prior written notice (or telephonic notice confirmed in writing) of such proposed prepayment or repayment pursuant to this §2.9 of Base Rate Loans and (ii) at least three (3) Eurodollar Business Days' prior written notice (or telephonic notice confirmed in writing) of any proposed prepayment or repayment pursuant to this §2.9 of Eurodollar Rate Loans, in each case, specifying the proposed date of prepayment or repayment or repayment or be paid. Each such partial prepayment shall be applied, in the absence of instruction by the Borrowers, first to the principal of Base Rate Loans and then to the principal of Eurodollar Rate Loans. Payments received from the Borrowers shall be applied *pro rata* to each Revolving Credit Lender in respect of its outstanding Commitment.

§3. LETTERS OF CREDIT.

§3.1. Letter of Credit Commitments.

(a) Subject to the terms and conditions hereof and the execution and receipt of a Loan and Letter of Credit Request reflecting the Maximum Drawing Amount of all Letters of Credit (including the requested Letter of Credit) and a Letter of Credit

Application, the Issuing Lender, on behalf of the Revolving Credit Lenders and in reliance upon the agreement of such Lenders set forth in §3.1(c) and upon the representations and warranties of the Borrowers contained herein, agrees to issue one or more standby letters of credit (individually, a "<u>Letter of Credit</u>"), in such form as may be requested from time to time by the Borrowers and agreed to by the Issuing Lender; <u>provided</u>, <u>however</u>, that, after giving effect to such request, (i) the aggregate Maximum Drawing Amount of all Letters of Credit issued at any time under this §3.1(a) shall not exceed \$80,000,000 and (ii) the aggregate Maximum Drawing Amount of all Letters of Credit and all Unpaid Reimbursement Obligations <u>plus</u> the aggregate outstanding amount of Revolving Credit Loans <u>plus</u> the aggregate outstanding amount of Swing Line Loans shall not exceed the Total Commitment; and <u>provided further</u> that no Letter of Credit shall have an expiration date later than the earlier of (A) one year after the date of issuance of such Letter of Credit (subject to periodic extensions for periods not to exceed one year), or (B) thirty (30) days prior to the Revolving Credit Maturity Date.

(b) Each Revolving Credit Lender severally agrees that it shall be absolutely liable, without regard to the occurrence of any Default or Event of Default or any other condition precedent whatsoever, to the extent of such Lender's Commitment Percentage thereof, to reimburse the Issuing Lender on demand for the amount of each draft paid by the Issuing Lender under each applicable Letter of Credit to the extent that such amount is not reimbursed by the Borrowers pursuant to §3.2 (such agreement for a Lender being called herein the "Letter of Credit Participation" of such Lender). The Issuing Lender shall not issue any Letter of Credit unless all of the conditions precedent under §11 hereof have been satisfied.

(c) Each such payment made by a Lender shall be treated as the purchase by such Lender of a participating interest in the Borrowers' Reimbursement Obligation under §3.2 in an amount equal to such payment. Each Lender shall share in accordance with its participating interest in any interest which accrues pursuant to §3.2.

(d) All "Letters of Credit" (as defined in the Existing Credit Agreement) outstanding under the Existing Credit Agreement on the Effective Date shall become Letters of Credit hereunder. The Revolving Credit Lenders' participations in such Letters of Credit will be reallocated on the Effective Date in accordance with each such Lender's applicable Commitment Percentage hereunder.

(e) Each Letter of Credit so issued, extended or renewed shall be subject to either the Uniform Customs and Practice for Documentary Credits (1993 Revision), International Chamber of Commerce Publication No. 500 or any successor version thereto adopted by the Issuing Lender in the ordinary course of its business as a letter of credit issuer and in effect at the time of issuance of such Letter of Credit or the International Standby Practices (ISP98), International Chamber of Commerce Publication No. 590, or any successor code of standby letter of credit practices among banks adopted by the Issuing Lender in the ordinary course of its business as a standby letter of credit issuer and in effect at the time of issuance of such Letter of Credit.

(f) If any Borrower so requests in an application for a Letter of Credit, the Issuing Lender may, in its sole and absolute discretion, agree to issue a Letter of Credit that has automatic renewal provisions (each, an "<u>Auto-Renewal Letter of Credit</u>"); provided that any such Auto-Renewal Letter of Credit must permit the Issuing Lender to

prevent any such renewal at least once in each twelve-month period (commencing with the date of issuance of such Letter of Credit) by giving prior notice to the beneficiary thereof not later than ten (10) days prior to the renewal date (the "<u>Nonrenewal Notice Date</u>") in each such twelve-month period to be agreed upon at the time such Letter of Credit is issued. Unless otherwise directed by the Issuing Lender, the Borrowers shall not be required to make a specific request to the Issuing Lender for any such renewal. Once an Auto-Renewal Letter of Credit has been issued, the Lenders shall be deemed to have authorized (but may not require) the Issuing Lender to permit the renewal of such Letter of Credit at any time to an expiry date not later than thirty (30) days prior to the Revolving Credit Maturity Date; <u>provided</u>, however, that the Issuing Lender shall not permit any such renewal if (A) the Issuing Lender has determined that it would have no obligation at such time to issue such Letter of Credit in its renewed form under the terms hereof, or (B) it has received notice (which may be by telephone or in writing) on or before the day that is two (2) Business Days before the Nonrenewal Notice Date (1) from the Administrative Agent that the Required Lenders have elected not to permit such renewal or (2) from the Administrative Agent, any Lender or the Borrowers that one or more of the applicable conditions specified in §11 is not then satisfied.

§3.2. <u>Reimbursement Obligations of the Borrowers.</u> In order to induce the Issuing Lender to issue, extend and renew Letters of Credit and the Lenders to participate therein, the Borrowers hereby jointly and severally agree to reimburse or pay to the Issuing Lender with respect to each Letter of Credit issued, extended or renewed by the Issuing Lender hereunder as follows:

(a) if any draft presented under any Letter of Credit is honored by the Issuing Lender or the Issuing Lender otherwise makes payment with respect thereto, the sum of (i) the amount paid by the Issuing Lender under or with respect to such Letter of Credit <u>plus</u> (ii) the amount of any taxes, fees, charges or other costs and expenses whatsoever incurred by the Issuing Lender in connection with any payment made by the Issuing Lender under, or with respect to, such Letter of Credit, <u>provided however</u>, if the Borrowers do not reimburse the Issuing Lender on the Drawdown Date, such amount shall, so long as no Event of Default under §§13.1(g) or 13.1(h) has occurred, become automatically a Revolving Credit Loan which is a Base Rate Loan advanced hereunder in an amount equal to such sum;

(b) upon the reduction (but not termination) of the Total Commitment to an amount less than the Maximum Drawing Amount, an amount equal to such difference, which amount shall be held by the Administrative Agent for the benefit of the Lenders and the Administrative Agent as cash collateral for all Reimbursement Obligations of the Borrowers;

(c) upon the Revolving Credit Maturity Date, the termination of the Total Commitment or the acceleration of the Reimbursement Obligations with respect to all Letters of Credit in accordance with §13, an amount equal to the then Maximum Drawing Amount of all Letters of Credit shall be paid by the Borrowers to the Administrative Agent to be held as cash collateral for the Reimbursement Obligations of the Borrowers; and

(d) the Borrowers promise to pay on the Revolving Credit Maturity Date all Unpaid Reimbursement Obligations on such date relating to Letters of Credit. All such

payments shall be made together with any and all accrued and unpaid interest thereon and any fees and other amounts owing hereunder.

Each such payment shall be made to the Administrative Agent at the Administrative Agent's Office in immediately available funds. Interest on any and all amounts remaining unpaid by the Borrowers under this §3.2 at any time from the date such amounts become due and payable (whether as stated in this §3.2, by acceleration or otherwise) until payment in full (whether before or after judgment) shall be payable to the Administrative Agent on demand at the rate of interest specified in §5.6 for overdue amounts.

§3.3. Letter of Credit Payments. If any draft shall be presented or other demand for payment shall be made under any Letter of Credit, the Issuing Lender shall notify the Borrowers of the date and amount of the draft presented or demand for payment and of the date and time when it expects to pay such draft or honor such demand for payment. On the date that such draft is paid or other payment is made by the Issuing Lender, the Issuing Lender shall promptly notify the Lenders of the amount of any Unpaid Reimbursement Obligation. All such Unpaid Reimbursement Obligations with respect to Letters of Credit shall, provided that no Event of Default under §13(g) or §13(h) has occurred, become automatically a Revolving Credit Loan which is a Base Rate Loan. No later than 3:00 p.m. (Boston time) on the Business Day next following the receipt of such notice, each Lender shall make available to the Issuing Lender, at the Administrative Agent's Office, in immediately available funds, such Lender's Commitment Percentage of such Unpaid Reimbursement Obligation, together with an amount equal to the product of (a) the average, computed for the period referred to in clause (c) below, of the weighted average interest rate paid by the Issuing Lender for federal funds acquired by the Issuing Lender furing each day included in such period, times (b) the anount equal to such Lender's Commitment Percentage of such Unpaid Reimbursement Obligation, times (c) a fraction, the numerator of which is the number of days that have elapsed from and including the date the Issuing Lender paid the draft presented for honor or otherwise made payment until the date on which such Lender's Commitment Percentage of such Unpaid Reimbursement Obligation shall become immediately available to the Issuing Lender, and the denominator of which is 365. The responsibility of the Issuing Lender to the Borrowers and the Lenders shall be only to determine that the documents (including each draft) delivered under each Letter of Credit in connection with such pres

§3.4. Obligations Absolute. The Borrowers' respective obligations under this §3 shall be absolute and unconditional under any and all circumstances and irrespective of the occurrence of any Default or Event of Default or any condition precedent whatsoever or any setoff, counterclaim or defense to payment which the Borrowers may have or have had against the Issuing Lender, any Lender or any beneficiary of a Letter of Credit. The Borrowers further agree with the Issuing Lender and the Revolving Credit Lenders that the Issuing Lender and the Revolving Credit Lenders shall not be responsible for, and the Borrowers' Reimbursement Obligations under §3.2 shall not be affected by, among other things, the validity or genuineness of documents or of any endorsements thereon, even if such documents should in fact prove to be in any or all respects invalid, fraudulent or forged, or any dispute between or among the Borrowers, the beneficiary of any Letter of Credit or any financing institution or other

party to which any Letter of Credit may be transferred or any claims or defenses whatsoever of the Borrowers against the beneficiary of any Letter of Credit or any such transferee. The Issuing Lender and the Revolving Credit Lenders shall not be liable for any error, omission, interruption or delay in transmission, dispatch or delivery of any message or advice, however transmitted, in connection with any Letter of Credit. The Borrowers agree that any action taken or omitted by the Issuing Lender or any Revolving Credit Lender under or in connection with each Letter of Credit and the related drafts and documents, if done in good faith, shall be binding upon the Borrowers and shall not result in any liability on the part of the Issuing Lender or any Revolving Credit Lender to the Borrowers.

§3.5. Reliance by Issuing Lender. To the extent not inconsistent with §3.3, the Issuing Lender shall be entitled to rely, and shall be fully protected in relying upon, any Letter of Credit, draft, writing, resolution, notice, consent, certificate, affidavit, letter, cablegram, telegram, telecopy, telex or teletype message, statement, order or other document believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons and upon advice and statements of legal counsel, independent accountants and other experts selected by the Issuing Lender.

§4. THE TERM LOAN.

<u>§4.1.</u> Commitment to Lend. Subject to the terms and conditions set forth in this Credit Agreement, on the Effective Date, each Term Loan Lender agrees to lend to the Borrowers its Term Loan Percentage of the principal amount of \$150,000,000.

§4.2. The Term Notes. The Term Loan shall be evidenced by separate promissory notes of the Borrowers in substantially the form of Exhibit A-3 hereto (each a "Term Note"), dated the Effective Date (or such other date on which a Term Loan Lender may become a party hereto in accordance with §19 hereof) and completed with appropriate insertions. One Term Note shall be payable to the order of each Term Loan Lender in a principal amount equal to such Lender's Term Loan Percentage of the Term Loan and representing the obligation of the Borrowers to pay to such Lender such principal amount or, if less, the outstanding amount of such Lender's Term Loan Percentage of the Term Loan Percentage of the Term Loan, plus interest accrued thereon, as set forth below. The Borrowers irrevocably authorize each Term Loan Lender to make or cause to be made a notation on such Lender's Term Note Record reflecting the original principal amount of such Lender's Term Loan Percentage of the Term Loan and, at or about the time of such Lender's receipt of any principal payment on such Lender's Term Note, an appropriate notation on such Lender's Term Note Record reflecting the original principal amount set forth on such Lender's Term Note Record shall be prima facie evidence of the principal amount thereof owing and unpaid to such Lender, but the failure to record, or any error in so recording, any such amount on such Lender's Term Note when due.

§4.3. Scheduled Installment Payments of Principal of Term Loan. The Borrowers jointly and severally promise to pay to the Administrative Agent for the account of the Term Loan Lenders, in accordance with their respective Term Loan Percentages, the principal amount of the Term Loan in (a) six (6) consecutive annual installment payments or (b) in the event that the Term Loan Maturity Date is changed in

accordance with the definition thereof, four (4) consecutive annual installment payments, in each case, each such payment equal to one percent (1%) of the notional amount of the Term Loan, and due and payable on each anniversary of the Effective Date, commencing on the first anniversary of the Effective Date, with a final additional payment on the Term Loan Maturity Date in an amount equal to the unpaid balance of the Term Loan.

§4.4. Mandatory Prepayments of Term Loan.

§4.4.1. Mandatory Prepayments.

(a) In the event that Net Cash Proceeds received by the Borrowers and the Excluded Subsidiaries from asset sales exceed \$5,000,000 per annum (other than asset sales in the ordinary course of business and sales permitted under §8.4.2(b)), the Borrowers will use any such excess Net Cash Proceeds to pay down the Term Loan in the manner set forth in §4.4.2.

(b) In the event that, after the Effective Date, any Borrower or any Excluded Subsidiary receives Net Cash Proceeds from a permitted debt offering in excess of \$100,000,000 in the aggregate, the Borrowers shall use one-hundred percent (100%) of such excess Net Cash Proceeds to pay down the Term Loan in the manner set forth in §4.4.2.

(c) In the event that, after the Effective Date, the Borrowers receive Net Equity Proceeds from Equity Offerings in excess of \$125,000,000 in the aggregate (other than from Capital Stock issued as payment for Permitted Acquisitions and Capital Stock consisting of options or other forms of equity-based compensation issued to employees, consultants and directors in accordance with a bona fide compensation plan approved by the Board of Directors of the Parent) the Borrowers shall use fifty-percent (50%) of such excess Net Equity Proceeds to pay down the Term Loan in the manner set forth in §4.4.2.

(d) The Borrowers shall use a percentage of Excess Operating Cash Flow in each fiscal year to pay down the Term Loan in the manner set forth in §4.4.2, which shall be payable within three (3) days of delivery of the year-end financial statements to the Administrative Agent. The applicable percentage of Excess Operating Cash Flow is set forth in the table below opposite the applicable ratio of Consolidated Total Funded Debt to Consolidated EBITDA for the period of four consecutive fiscal quarters ending on the last day of the Borrowers' fiscal year:

Ratio of Consolidated Total Funded	Percentage of Excess Operating Cash
Debt to Consolidated EBITDA	Flow
less than 3.00:1.00	0%
greater than or equal to 3.00:1.00 and	25%
less than 3.50:1.00	
greater than or equal to 3.50:1.00	50%

§4.4.2. <u>Application of Payments</u>. Each prepayment of the Term Loan required by §4.4.1 shall be allocated among the Term Loan Lenders in accordance with each such Lender's Term Loan Percentage. Any prepayment of principal of the Term

Loan shall include all interest accrued to the date of prepayment and shall be applied against the scheduled installments of principal due on the Term Loan in the inverse order of maturity. No amount prepaid or repaid with respect to the Term Loan may be reborrowed. Any Term Loan Lender may decline to accept any payments due to such Term Loan Lender pursuant to this §4.4 in which case such declined payments shall be used to repay the Revolving Credit Loans (but not reduce the Total Commitment) on a *pro rata* basis in accordance with each Lender's Commitment Percentage.

§4.5. Optional Prepayment of Term Loan. The Borrowers shall have the right at any time to prepay the Term Notes on or before the Term Loan Maturity Date, in whole, or in part, upon not less than three (3) Business Days' prior written notice to the Administrative Agent, without premium or penalty (other than the obligation to reimburse the Term Loan Lenders and the Administrative Agent pursuant to §5.12 hereof, or as otherwise stated herein), provided that (a) each partial prepayment shall be in the principal amount of \$1,000,000 or an integral multiple of \$500,000 thereof, and (b) each partial prepayment shall be allocated among the Term Loan Lenders in accordance with such Lender's Term Loan Percentage. Any prepayment of principal of the Term Loan shall include all interest accrued to the date of prepayment and shall be applied against the scheduled installments of principal due on the Term Loan in the inverse order of maturity. No amount prepaid or repaid with respect to the Term Loan may be reborrowed.

§4.6. Interest on Term Loan.

§4.6.1. Interest Rates. Except as otherwise provided in §5.6, the Term Loan shall bear interest during each Interest Period relating to all or any portion of the Term Loan at the following rates:

(a) To the extent that all or any portion of the Term Loan bears interest during such Interest Period at the Base Rate, the Term Loan or such portion thereof shall bear interest during such Interest Period at the Applicable Rate for Term Loans that are Base Rate Loans as in effect from time to time.

(b) To the extent that all or any portion of the Term Loan bears interest during such Interest Period at the Eurodollar Rate, the Term Loan or such portion thereof shall bear interest during such Interest Period at the rate per annum equal to the Applicable Rate for Term Loans that are Eurodollar Rate Loans as in effect from time to time.

The Borrowers jointly and severally promise to pay interest on the Term Loan or any portion thereof outstanding during each Interest Period in arrears on each Interest Payment Date applicable to such Interest Period and on the Term Loan Maturity Date. Any change in the interest rate resulting from a change in the Base Rate is to be effective at the beginning of the day of such change in the Base Rate.

§4.6.2. Notification by Borrowers. The Borrowers shall notify the Administrative Agent, such notice to be irrevocable, at least three (3) Eurodollar Business Days prior to the Drawdown Date of the Term Loan (or any portion thereof) if all or any portion of the Term Loan is to bear interest at the Eurodollar Rate. The provisions of §5.11 and §5.12 shall apply <u>mutatis mutandis</u> with respect to all or any portion of the

Term Loan so that the Borrowers may have the same interest rate options with respect to all or any portion of the Term Loan as they would be entitled to with respect to Revolving Credit Loans, <u>provided</u>, <u>however</u>, the Borrowers will have no more than ten (10) different maturities of Eurodollar Rate Loans (whether a portion of the Term Loan or Revolving Credit Loans) outstanding at any time. In the event that the Borrowers fail to give the Administrative Agent notice with respect to the continuation of any Eurodollar Rate Loan hereunder within three (3) days prior to the expiration of the Interest Period relating thereto, then such Eurodollar Rate Loan shall be converted to a Base Rate Loan.

§4.6.3. <u>Amounts, etc.</u> Any portion of the Term Loan bearing interest at the Eurodollar Rate relating to any Interest Period shall be in the amount of \$1,000,000 or an integral thereof. No Interest Period relating to the Term Loan or any portion thereof bearing interest at the Eurodollar Rate shall extend beyond the date on which any regularly scheduled installment payment of the principal of the Term Loan is to be made unless a portion of the Term Loan at least equal to such installment payment has an Interest Period ending on such date or is then bearing interest at the Base Rate.

§5. FEES; PAYMENTS; COMPUTATIONS; JOINT AND SEVERAL LIABILITY; CERTAIN GENERAL PROVISIONS.

§5.1. Fees.

(a) Commitment Fee. The Borrowers jointly and severally in accordance with §5.9 agree (to the fullest extent permitted by law) to pay to the Administrative Agent for the benefit of the Revolving Credit Lenders in accordance with their respective Commitment Percentages a commitment fee (the "<u>Commitment Fee</u>") calculated at the rate per annum equal to the Applicable Rate with respect to the Commitment Fee as in effect from time to time on the average daily amount during each calendar quarter or portion thereof from the Effective Date to the Revolving Credit Maturity Date by which the Total Commitment <u>minus</u> the sum of the Maximum Drawing Amount and all Unpaid Reimbursement Obligations exceeds the outstanding amount of Revolving Credit Loans (including Swing Line Loans) during such calendar quarter. The Commitment Fee shall be payable quarterly in arrears on the first day of each calendar quarter for the immediately preceding calendar quarter commencing on the first such date following the date hereof, with a final payment on the Revolving Credit Maturity Date or any earlier date on which the Commitments shall terminate.

(b) Letter of Credit Fees.

(i) The Borrowers jointly and severally in the case of Letters of Credit which are Financial L/Cs agree to pay, at the times specified in paragraph (iii) hereof, a fee (a "<u>Financial L/C Fee</u>") to the Administrative Agent for the benefit of the Revolving Credit Lenders, equal to the product of (A) the Letter of Credit Percentage <u>multiplied by</u> (B) the Maximum Drawing Amount of each Financial L/C on the date of calculation, to be shared *pro rata* by each of such Lenders in accordance with their respective Commitment Percentages.

(ii) The Borrowers jointly and severally in the case of Performance L/Cs agree to pay, at the times specified in paragraph (iii) hereof, a fee (a "<u>Performance L/C Fee</u>") to the Administrative Agent for the benefit of the Revolving Credit Lenders, equal to fifty

percent (50%) of the product of (A) the Letter of Credit Percentage <u>multiplied by</u> (B) the Maximum Drawing Amount of each such Letter of Credit on the date of calculation, to be shared *pro rata* by each of such Lenders in accordance with their respective Commitment Percentages.

(iii) The Financial L/C Fee and Performance L/C Fee and the Issuance Fee (as defined below) shall be payable quarterly in arrears on the first day of each calendar quarter for the immediately preceding calendar quarter and on the Revolving Credit Maturity Date with respect to the average daily Maximum Drawing Amount of Letters of Credit outstanding during such calendar quarter or portion thereof. In addition, the Borrowers jointly and severally agree to pay an issuing fee to the Issuing Lender for its account in an amount equal to one eighth of one percent (0.125%) per annum of the Maximum Drawing Amount of each Letter of Credit issued by the Issuing Lender, <u>plus</u> any customary issuance, amendment, negotiation or document examination and other administrative fees of such Issuing Lender in effect from time to time (the "<u>Issuance Fee</u>").

(c) The Borrowers shall also pay to the Administrative Agent for its own account and/or for the account of the Syndication Agent, the Co-Arrangers and the Lenders, such other fees as have been agreed to in writing from time to time by the Borrowers and the Agents, each as set forth in separate letter agreements between the Borrowers and the Agents.

§5.2. Payments.

(a) <u>Payments to Administrative Agent</u>. All payments of principal, interest, Reimbursement Obligations, fees and any other amounts due hereunder or under any of the other Loan Documents shall be made to the Administrative Agent, for the respective accounts of the applicable Lenders ratably in accordance with their respective Loan Percentages, the Issuing Lender and the Administrative Agent, to be received at such Administrative Agent's Office in immediately available funds by 1:00 p.m. (Boston time) on any due date. The Administrative Agent shall promptly distribute such amounts to the applicable Lenders.

(b) No Offset, etc. All payments by the Borrowers hereunder and under any of the other Loan Documents shall be made without setoff or counterclaim and free and clear of and without deduction for any taxes, levies, imposts, duties, charges, fees, deductions, withholdings, compulsory loans, restrictions or conditions of any nature now or hereafter imposed or levied by any jurisdiction or any political subdivision thereof or taxing or other authority therein unless the Borrowers are compelled by law to make such deduction or withholding. Except as otherwise provided in this §5.2, if any such obligation is imposed upon the Borrowers with respect to any amount payable by them hereunder or under any of the other Loan Documents, the Borrowers will pay to the Administrative Agent for the account of the applicable Lenders or (as the case may be) the Administrative Agent, on the date on which such amount is due and payable hereunder or under such other Loan Document, such additional amount in Dollars as shall be necessary to enable the applicable Lenders or the Administrative Agent to receive the same net amount which such Lenders or the Administrative Agent would have received on such due date had no such obligation been imposed upon the Borrowers. The Borrowers will deliver promptly to the Lender certificates or other valid vouchers for all taxes or other charges deducted from or paid with respect to payments made by the Borrowers hereunder or under such other Loan Document.

Non-U.S. Lenders. Each Lender that is not a U.S. Person as defined in Section 7701(a)(30) of the Code for federal income tax purposes (a "Non-U.S. Lender") agrees that, if and to the extent it is legally able to do so, it shall, prior to the first date on which any payment is due to it hereunder, deliver to the Borrowers and the Administrative Agent such certificates, documents or other evidence, as and when required by the Code or Treasury Regulations issued pursuant thereto, including, (a) in the case of a Non-U.S. Lender that is a "bank" for purposes of Section 881(c) (3)(A) of the Code, two (2) duly completed copies of Internal Revenue Service Form W-8BEN or W-8ECI or successor applicable form, as the case may be, and any other certificate or statement of exemption required by Treasury Regulations, establishing that, with respect to payments of principal, interest or fees hereunder, such Non-U.S. Lender is (i) not subject to United States federal withholding tax under the Code because such payment is effectively connected with the conduct by such Non-U.S. Lender of a trade or business in the United States or (ii) totally exempt or partially exempt from United States federal withholding tax under a provision of an applicable tax treaty and (b) in the case of a Non-U.S. Lender that is not a "bank" for purposes of Section 881(c)(3)(A) of the Code, a certificate in form and substance reasonably satisfactory to the Administrative Agent and the Borrowers and to the effect that (i) such Non-U.S. Lender is not a "bank" for purposes of Section 881(c)(3)(A) of the Code, is not subject to regulatory or other legal requirements as a bank in any jurisdiction, and has not been treated as a bank for purposes of any tax, securities law or other filing or submission made to any governmental authority, any application made to a rating agency or qualification for any exemption from any tax, securities law or other legal requirements, (ii) is not a ten (10) percent shareholder for purposes of Section 881(c)(3)(B) of the Code and (iii) is not a controlled foreign corporation receiving interest from a related person for purposes of Section 881(c)(3)(C) of the Code, together with a properly completed Internal Revenue Service Form W-8 or W-9, as applicable (or successor forms). Each Lender agrees that it shall, promptly upon a change of its lending office or the selection of any additional lending office, to the extent the forms previously delivered by it pursuant to this section are no longer effective, and promptly upon the Borrowers' or the Administrative Agent's reasonable request after the occurrence of any other event (including the passage of time) requiring the delivery of a Form W-8BEN, Form W-8ECI, Form W-8 or W-9 in addition to or in replacement of the forms previously delivered, deliver to the Borrowers and the Administrative Agent, as applicable, if and to the extent it is properly entitled to do so, a properly completed and executed Form W-8BEN, Form W-8ECI, Form W-9, as applicable (or any successor forms thereto)

(d) The Borrowers shall not be required to pay any additional amounts to any Non-U.S. Lender in respect of United States federal withholding tax pursuant to \$5.2(b) to the extent that (i) the obligation to withhold such amounts existed on the date such Non-U.S. Lender became a party to this Credit Agreement or, with respect to payments to a different lending office designated by the Non-U.S. Lender as its applicable lending office, the date such Non-U.S. Lender designated such new lending office with respect to a Loan; or (ii) the obligation to pay such additional amounts would not have arisen but for a failure by such Non-U.S. Lender to comply with the provisions of paragraph (c) above.

(e) In the event that the Borrowers are required to make such deduction or withholding as a result of the fact that a Lender is a Non-U.S. Lender, such Lender shall use its reasonable best efforts to transfer its Loans to an affiliate that is a U.S. Lender if such transfer would have no adverse effect on such Lender or the Loans.

§5.3. <u>Computations.</u> Except as otherwise expressly provided herein, all computations of interest, Commitment Fees, Financial L/C Fees, Performance L/C Fees or other fees shall be based on a 360-day year and paid for the actual number of days elapsed. Computations of the interest on Base Rate Loans shall be based on a 365/366- day year and paid for the actual number of days elapsed. Whenever a payment hereunder or under any of the other Loan Documents becomes due on a day that is not a Business Day, the due date for such payment shall be extended to the next succeeding Business Day, and interest shall accrue during such extension.

§5.4. Capital Adequacy. If any Lender or the Administrative Agent shall have determined that, after the date hereof, (a) the adoption of, or change in, any law, rule, regulation, policy, guideline or directive (whether or not having the force of law) regarding capital requirements for banks or bank holding companies or any change in the interpretation or, application or administration thereof by any governmental authority, central bank or comparable agency with appropriate jurisdiction, or (b) compliance by such Lender or the Administrative Agent or any corporation controlling such Lender or the Administrative Agent with any law, governmental rule, regulation, policy, guideline or directive (whether or not having the force of law) of any such entity regarding capital adequacy, in either case, has or would have the effect of reducing the rate of return on such Lender's or the Administrative Agent's commitment with respect to any Loans to a level below that which such Lender or the Administrative Agent's compliance (taking into consideration such Lender's or the Administrative Agent to be material, then the Borrowers shall, within thirty (30) days after being presented with a certificate in accordance with §5.5, pay such Lender or (as the case may be) the Administrative Agent's reasonable determination, fairly compensate such Lender or the Administrative Agent for such reduction in the return on capital. Each Lender shall allocate such cost increases among its customers in good faith and on an equitable basis.

§5.5. Certificate. A certificate setting forth any additional amounts payable pursuant to §5.4 or §5.8 and a reasonable explanation of such amounts which are due, submitted by any Lender or the Administrative Agent to the Borrowers, shall be conclusive, absent manifest error, that such amounts are due and owing.

§5.6. Interest After Default. Overdue principal and (to the extent permitted by applicable law) interest on the Loans and all other overdue amounts payable hereunder or under any of the other Loan Documents shall bear interest compounded monthly and payable on demand at a rate per annum equal to two percent (2.00%) above the rate of interest otherwise applicable to such Loans until such overdue amounts shall be paid in full (after as well as before judgment).

§5.7. Interest Limitation. Notwithstanding any other term of this Credit Agreement or the Notes, any other Loan Document or any other document referred to herein or therein, the maximum amount of interest which may be charged to or collected from any Person liable hereunder or under the Notes by any Lender shall be absolutely limited to, and shall in no event exceed, the maximum amount of interest which could lawfully be charged or collected by such Lender under applicable laws (including, to the extent applicable, the provisions of §5197 of the Revised Statutes of the United States of America, as amended, 12 U.S.C. §85).

§5.8. Additional Costs, Etc. If any present or future applicable law, which expression, as used herein, includes statutes, rules and regulations thereunder and interpretations thereof by any competent court or by any governmental or other regulatory body or official charged with the administration or the interpretation thereof and requests, directives, instructions and notices at any time or from time to time hereafter made upon or otherwise issued to any Lender, the Issuing Lender or the Administrative Agent by any central bank or other fiscal, monetary or other authority (whether or not having the force of law), shall:

(a) impose on any Lender, the Issuing Lender or the Administrative Agent any tax, levy, impost, duty, charge, fees, deduction or withholdings of any nature or requirements with respect to this Credit Agreement, the other Loan Documents, the Loans, such Lender's Commitment, the Letters of Credit or any class of loans or commitments or letters of credit of which any of the Loans, the Commitment or the Letters of Credit forms a part (other than taxes based upon or measured by the income or profits of such Lender, the Issuing Lender or the Administrative Agent), or

(b) materially change the basis of taxation (except for changes in taxes on income or profits) of payments to any Lender of the principal of or the interest on any Loans or any other amounts payable to any Lender, the Issuing Lender or the Administrative Agent under this Credit Agreement or any of the other Loan Documents, or

(c) impose or increase or render applicable (other than to the extent specifically provided for elsewhere in this Credit Agreement) any special deposit, reserve, assessment, liquidity, capital adequacy or other similar requirements (whether or not having the force of law) against assets held by, or deposits in or for the account of, or loans by, or letters of credit issued by, or commitments of an office of any Lender or the Issuing Lender, or

(d) impose on any Lender, the Issuing Lender or the Administrative Agent any other conditions or requirements with respect to this Credit Agreement, the other Loan Documents, any Letters of Credit, the Loans, such Lender's Commitment, or any class of loans, letters of credit or commitments of which any of the Loans or such Lender's Commitment forms a part, and the result of any of the foregoing is:

(e) to increase the cost to any Lender or the Issuing Lender of making, funding, issuing, renewing, extending or maintaining the Loans, such Lender's Commitment or any Letters of Credit; or

(f) to reduce the amount of principal, interest, Reimbursement Obligation, fees or other amount payable to such Lender, the Issuing Lender or the Administrative Agent hereunder on account of such Lender's Commitment, the Loans, or drawings under the Letters of Credit, or

(g) to require such Lender, the Issuing Lender or the Administrative Agent to make any payment or to forego any interest or Reimbursement Obligation other sum payable hereunder, the amount of which payment or foregone interest or Reimbursement Obligation or other sum is calculated by reference to the gross amount of any sum receivable or deemed received by such Lender, the Issuing Lender or the Administrative Agent from the Borrowers hereunder,

then, and in each such case, the Borrowers will, upon demand made by such Lender, the Issuing Lender or (as the case may be) the Administrative Agent at any time and from time to time and as often as the occasion therefor may arise, pay to such Lender, the Issuing Lender or the Administrative Agent such additional amounts as will be sufficient to compensate such Lender, the Issuing Lender or the Administrative Agent for such additional cost, reduction, payment or foregone interest or Reimbursement Obligation or other sum (after such Lender, the Issuing Lender or (as the case may be) the Administrative Agent shall have allocated the same fairly and equitably among all customers of any class generally affected thereby).

§5.9. Concerning Joint and Several Liability of the Borrowers.

(a) Each of the Borrowers is accepting joint and several liability for all of the Obligations hereunder and under the other Loan Documents in consideration of the financial accommodations to be provided by the Agents, the Issuing Lender and the Lenders under this Credit Agreement, for the mutual benefit, directly and indirectly, of each of the Borrowers and in consideration of the undertakings of each other Borrower to accept joint and several liability for the Obligations of the Borrowers.

(b) Each of the Borrowers, jointly and severally, hereby irrevocably and unconditionally accepts, not merely as a surety but also as a co-debtor, joint and several liability with the other Borrowers with respect to the payment and performance of all of the Obligations of the Borrowers (including, without limitation, any Obligations arising under this §5.9), it being the intention of the parties hereto that all of the Obligations shall be the joint and several obligations of each of the Borrowers without preferences or distinction among them.

(c) If and to the extent that any of the Borrowers shall fail to make any payment with respect to any of the Obligations as and when due or to perform any of the Obligations in accordance with the terms thereof, then in each such event the other Borrowers will make such payment with respect to, or perform, such Obligation.

(d) The Obligations of each of the Borrowers under the provisions of this §5.9 constitute full recourse obligations of each such Borrower enforceable against each such Borrower to the full extent of its properties and assets, to the fullest extent permitted by applicable law, irrespective of the validity, regularity or enforceability of this Credit Agreement against any other Borrower or any other circumstance whatsoever.

Except as otherwise expressly provided in this Credit Agreement, each of the Borrowers, to the fullest extent permitted by (e) applicable law, hereby waives notice of acceptance of its joint and several liability, notice of any Loans made under this Credit Agreement, notice of any action at any time taken or omitted by the Agents, the Issuing Lender or the Lenders under or in respect of any of the Obligations, and, generally, to the extent permitted by applicable law, all demands, notices and other formalities of every kind in connection with this Credit Agreement. Each Borrower, to the fullest extent permitted by applicable law, hereby waives all defenses which may be available by virtue of any valuation, stay, moratorium law or other similar law now or hereafter in effect, any right to require the marshaling of assets of the Borrowers and any other entity or Person primarily or secondarily liable with respect to any of the Obligations, and all suretyship defenses generally. Each of the Borrowers, to the fullest extent permitted by applicable law, hereby assents to, and waives notice of, any extension or postponement of the time for the payment of any of the Obligations, the acceptance of any payment of any of the Obligations, the acceptance of any partial payment thereon, any waiver, consent or other action or acquiescence by the Lenders, the Agents or the Issuing Lender at any time or times in respect of any default by any of the Borrowers in the performance or satisfaction of any term, covenant, condition or provision of this Credit Agreement, any and all other indulgences whatsoever by the Lenders, the Agents or the Issuing Lender in respect of any of the Obligations, and the taking, addition, substitution or release, in whole or in part, at any time or times, of any security for any of the Obligations or the addition, substitution or release, in whole or in part, of any of the Borrowers. Without limiting the generality of the foregoing, to the fullest extent permitted by law, each of the Borrowers assents to any other action or delay in acting or failure to act on the part of the Lenders, the Agents or the Issuing Lender with respect to the failure by any of the Borrowers to comply with any of its respective Obligations including, without limitation, any failure strictly or diligently to assert any right or to pursue any remedy or to comply fully with applicable laws or regulations thereunder, which might, but for the provisions of this §5.9, afford grounds for terminating, discharging or relieving any of the Borrowers, in whole or in part, from any of its Obligations under this §5.9, it being the intention of each of the Borrowers that, so long as any of the Obligations hereunder remain unsatisfied, the Obligations of such Borrowers under this §5.9 shall not be discharged except by performance and then only to the extent of such performance. The Obligations of each of the Borrowers under this \$5.9 shall not be diminished or rendered unenforceable by any winding up, reorganization, arrangement, liquidation, re-construction or similar proceeding with respect to any of the Borrowers, the Agents, the Issuing Lender or the Lenders. The joint and several liability of the Borrowers hereunder shall continue in full force and effect notwithstanding any absorption, merger, amalgamation or any other change whatsoever in the name, membership, constitution or place of formation of any of the Borrowers, the Agents, the Issuing Lender or the Lenders.

(f) To the extent any Borrower makes a payment hereunder in excess of the aggregate amount of the benefit received by such Borrower in respect of the extensions of credit under the Credit Agreement (the "<u>Benefit Amount</u>"), then such Borrower, after the payment in full, in cash, of all of the Obligations, shall be entitled to recover from each other Borrower such excess payment, <u>pro rata</u>, in accordance with the ratio of the Benefit Amount received by each such other Borrower to the total Benefit Amount received by all Borrowers, and the right to such recovery shall be deemed to be an asset and property of such Borrower so funding; <u>provided</u>, that each of the Borrowers hereby agrees that it will not enforce any of its rights of contribution or subrogation against the other

Borrowers with respect to any liability incurred by it hereunder or under any of the other Loan Documents, any payments made by it to any of the Lenders or the Administrative Agent with respect to any of the Obligations or any collateral security therefor until such time as all of the Obligations have been irrevocably paid in full in cash. Any claim which any Borrower may have against any other Borrower with respect to any payments to the Lenders or the Administrative Agent hereunder or under any other Loan Document are hereby expressly made subordinate and junior in right of payment, without limitation as to any increases in the Obligations arising hereunder or thereunder, to the prior payment in full of the Obligations and, in the event of any insolvency, bankruptcy, receivership, liquidation, reorganization or other similar proceeding under the laws of any jurisdiction relating to any Borrower, its debts or its assets, whether voluntary or involuntary, all such Obligations shall be paid in full before any payment or distribution of any character, whether in cash, securities or other property, shall be made to any other Borrower therefor.

(g) Each of the Borrowers hereby agrees that it will not enforce any of its rights of contribution or subrogation against the other Borrowers with respect to any liability incurred by it hereunder or under any of the other Loan Documents, any payments made by it to any of the Lenders, the Issuing Lender or either Agent with respect to any of the Obligations or any collateral security therefor until such time as all of the Obligations have been irrevocably paid in full in cash. Any claim which any Borrower may have against any other Borrower with respect to any payments to the Lenders, the Issuing Lender or either Agent hereunder or under any other Loan Document are hereby expressly made subordinate and junior in right of payment, without limitation as to any increases in the Obligations arising hereunder or thereunder, to the prior payment in full of the Obligations and, in the event of any insolvency, bankruptcy, receivership, liquidation, reorganization or other similar proceeding under the laws of any jurisdiction relating to any Borrower, its debts or its assets, whether voluntary or involuntary, all such Obligations shall be paid in full before any payment or distribution of any character, whether in cash, securities or other property, shall be made to any other Borrower therefor.

(h) Each of Borrowers hereby agrees that the payment of any amounts due with respect to the Indebtedness owing by any Borrower to any other Borrower is hereby subordinated to the prior payment in full in cash of the Obligations. Each Borrower hereby agrees that after the occurrences and during the continuance of any Default or Event of Default, such Borrower will not demand, sue for or otherwise attempt to collect any Indebtedness of any other Borrower owing to such Borrower until the Obligations shall have been paid in full in cash. If, notwithstanding the foregoing sentence, such Borrower shall collect, enforce or receive any amounts in respect of such Indebtedness before payment in full in cash of the Obligations, such amounts shall be collected, enforced, received by such Borrower as trustee for the Administrative Agent and be paid over to the Administrative Agent for the *pro rata* accounts of the relevant Lenders (in accordance with each such Lender's Loan Percentage) to be applied to repay (or be held as security for the repayment of) the Obligations.

(i) The provisions of this §5.9 are made for the benefit of the Agents, the Issuing Lender and the Lenders and their successors and assigns, and may be enforced in good faith by them from time to time against any or all of the Borrowers as often as the occasion therefor may arise and without requirement on the part of the Agents, the

Issuing Lender or the Lenders first to marshal any of their claims or to exercise any of their rights against any other Borrower or to exhaust any remedies available to them against any other Borrower or to resort to any other source or means of obtaining payment of any of the Obligations hereunder or to elect any other remedy. The provisions of this §5.9 shall remain in effect until all of the Obligations shall have been paid in full or otherwise fully satisfied. If at any time, any payment, or any part thereof, made in respect of any of the Obligations, is rescinded or must otherwise be restored or returned by the Agents, the Issuing Lender or the Lenders upon the insolvency, bankruptcy or reorganization of any of the Borrowers or is repaid in good faith settlement of a pending or threatened avoidance claim, or otherwise, the provisions of this §5.9 will forthwith be reinstated in effect, as though such payment had not been made.

(j) Each of the Borrowers hereby appoints the Parent, and the Parent hereby agrees, to act as its representative and authorized signor with respect to any notices, demands, communications or requests under this Credit Agreement or the other Loan Documents, including, without limitation, with respect to Loan and Letter of Credit Requests and Compliance Certificates and pursuant to §21 of this Credit Agreement.

(k) It is the intention and agreement of the Borrowers and the Lenders that the obligations of the Borrowers under this Credit Agreement shall be valid and enforceable against the Borrower to the maximum extent permitted by applicable law. Accordingly, if any provision of this Credit Agreement creating any obligation of the Borrowers in favor of the Lenders shall be declared to be invalid or unenforceable in any respect or to any extent, it is the stated intention and agreement of the Borrowers and the Lenders that any balance of the obligation created by such provision and all other obligations of the Borrowers to the Lenders created by other provisions of this Credit Agreement shall remain valid and enforceable. Likewise, if by final order a court of competent jurisdiction shall declare any sums which the Lenders may be otherwise entitled to collect from the Borrowers under this Credit Agreement to be in excess of those permitted under any law (including any federal or state fraudulent conveyance or like statute or rule of law) applicable to the Borrower's obligations under this Credit Agreement, it is the stated intention and agreement of the Borrowers and the Lenders that all sums not in excess of those permitted under such applicable law shall remain fully collectible by the Lenders from the Borrowers.

\$5.10. Currency of Account. All of the Loans and Letters of Credit hereunder shall be denominated and payable in

Dollars. If, for the purpose of obtaining judgment in any court it is necessary to convert a sum due hereunder in one currency (the "<u>first currency</u>") into any other currency (the "<u>second currency</u>") the conversion shall be made at the Spot Rate of exchange of the Administrative Agent (as conclusively determined by such Administrative Agent absent manifest error) on the Business Day preceding the day on which the final judgment is given. If, however, on the Business Day following receipt by the Administrative Agent in the second currency of any sum adjudged to be due hereunder (or any proportion thereof) the Administrative Agent purchases the first currency with the amount of the second currency so received and the first currency so purchased falls short of the sum originally due hereunder in the first currency (or the same proportion thereof) the Borrowers, shall, as a separate obligation and notwithstanding any judgment, pay to the Administrative Agent in the first currency an amount equal to such shortfall.

§5.11. Election of Eurodollar Rate; Notice of Election; Interest Periods; Minimum Amounts.

(a) At the Borrowers' option, so long as no Default or Event of Default has occurred and is then continuing, the Borrowers may (i) elect to convert any Base Rate Loan or a portion thereof to a Eurodollar Rate Loan, (ii) at the time of any Loan and Letter of Credit Request, specify that such requested Loan shall be a Eurodollar Rate Loan, or (iii) upon expiration of the applicable Interest Period, elect to maintain an existing Eurodollar Rate Loan as such, provided that the Borrowers give notice to the Administrative Agent pursuant to §5.11(b) hereof. Upon determining any Eurodollar Rate, the Administrative Agent shall forthwith provide notice thereof to the Borrowers and each Lender, and each such notice to the Borrowers shall be considered prima facie correct and binding, absent manifest error.

(b) Three (3) Eurodollar Business Days prior to the making of any Eurodollar Rate Loan or the conversion of any Base Rate Loan to a Eurodollar Rate Loan, or, in the case of an outstanding Eurodollar Rate Loan, the expiration date of the applicable Interest Period, the Borrowers shall give written, telex or telecopy notice received by the Administrative Agent not later than 11:00 a.m. (Boston time) of their election pursuant to §5.11(a). Each such notice delivered to the Administrative Agent shall specify the aggregate principal amount of the Loans to be borrowed or maintained as or converted to Eurodollar Rate Loans and the requested duration of the Interest Period that will be applicable to such Eurodollar Rate Loan, and such notice shall be irrevocable and binding upon the Borrowers. If the Borrowers shall fail to give the Administrative Agent notice of their election hereunder together with all of the other information required by this §5.11(b) with respect to any Loan, whether at the end of an Interest Period or otherwise, such Loan shall be deemed a Base Rate Loan, and, if such Loan is an existing Eurodollar Rate Loan may be continued as such when any Default or Event of Default has occurred and is continuing, but shall be automatically converted to a Base Rate Loan on the last day of the Interest Period relating thereto. The Administrative Agent shall promptly notify the Lenders in writing (or by telephone confirmed in writing or by telecopy) of such election by the Borrowers hereunder.

(c) Notwithstanding anything herein to the contrary, the Borrowers may not specify an Interest Period that would extend beyond the Revolving Credit Maturity Date, in the case of Revolving Credit Loans, or the Term Loan Maturity Date, in the case of Term Loans.

(d) In no event shall the Borrowers have more than ten (10) different maturities of borrowings of Eurodollar Rate Loans outstanding at any time.

§5.12. Eurodollar Indemnity. The Borrowers agree to indemnify the Lenders and the Administrative Agent and to hold them harmless from and against any reasonable loss, cost or expense (including loss of anticipated profits) that the Lenders and the Administrative Agent may sustain or incur as a consequence of (a) default by the Borrowers in payment of the principal amount of or any interest on any Eurodollar Rate Loans as and when due and payable, including any such loss or expense arising from interest or fees payable by any Lender or the Administrative Agent to lenders of funds

obtained by it in order to maintain its Eurodollar Rate Loans, (b) default by the Borrowers in making a borrowing or conversion after the Borrowers have given (or are deemed to have given) notice pursuant to §2.6 or §5.11 and (c) the making of any payment of a Eurodollar Rate Loan or the making of any conversion of any such Eurodollar Rate Loan to a Base Rate Loan on a day that is not the last day of the applicable Interest Period with respect thereto, including interest or fees payable by such Lender to lenders of funds obtained by it in order to maintain any such Loans or upon a transfer of interest in Eurodollar Rate Loans to an Acceding Lender pursuant to §19(g).

<u>§5.13.</u> Illegality; Inability to Determine Eurodollar Rate. Notwithstanding any other provision of this Credit Agreement if (a) any present or future law, regulation, treaty or directive or the interpretation or application thereof shall make it unlawful for any Lender to make or maintain Eurodollar Rate Loans, or (b) if any Lender or the Administrative Agent, as applicable shall reasonably determine with respect to Eurodollar Rate Loans that (i) by reason of circumstances affecting any Eurodollar interbank market, adequate and reasonable methods do not exist for ascertaining the Eurodollar Rate which would otherwise be applicable during any Interest Period, or (ii) deposits of Dollars in the relevant amount for the relevant Interest Period are not available to such Lender or the Administrative Agent in any Eurodollar interbank market, or (iii) the Eurodollar Rate does not or will not accurately reflect the cost to such Lender or the Administrative Agent of obtaining or maintaining the applicable Eurodollar Rate Loans during any Interest Period, then such Lender or the Administrative Agent shall promptly give telephonic, telex or cable notice of such determination to the Borrowers (which notice shall be conclusive and binding upon the Borrowers). Upon such notification by such Lender or the Administrative Agent, the obligation of the Lenders and the Administrative Agent to make Eurodollar Rate Loans shall be suspended until the Lenders or the Administrative Agent, as the case may be, determine that such circumstances no longer exist, and to the extent permitted by law the outstanding Eurodollar Rate Loans shall continue to bear interest at the applicable rate based on the Eurodollar Rate until the end of the applicable Interest Period, and thereafter shall be deemed converted to Base Rate Loans in equal principal amounts of such former Eurodollar Rate Loans. The Borrowers hereby agree promptly to pay to the Administrative Agent, for the account of the applicable Lenders or (as the case may be) the Administrative Agent, upon demand by such Lenders or the Administrative Agent, any additional amounts necessary to compensate such Lenders for any costs incurred in making any conversion in accordance with this \$5.13, including any interest or fees payable by such Lenders or the Administrative Agent to lenders of funds obtained in order to make or maintain its Eurodollar Rate Loans hereunder.

§6. REPRESENTATIONS AND WARRANTIES. The Borrowers jointly and severally represent and warrant to the Lenders, the Issuing Lender and the Agents that, on and as of the date of this Credit Agreement (any disclosure on a schedule pursuant to this §6 shall be deemed to apply to all relevant representations and warranties, regardless of whether such schedule is referenced in each relevant representation):

§6.1. Corporate Authority.

(a) **Incorporation:** Good Standing. Each of the Borrowers (i) is a corporation (or similar business entity) duly organized, validly existing and in good standing or in current status under the laws of its respective jurisdiction of organization,

(ii) has all requisite corporate (or the equivalent company or partnership) power to own its property and conduct its business as now conducted and as presently contemplated, and (iii) is in good standing as a foreign corporation (or similar business entity) and is duly authorized to do business in each jurisdiction in which its property or business as presently conducted or contemplated makes such qualification necessary except where a failure to be so qualified would not have a material adverse effect on the business, assets or financial condition of such Borrower.

(b) <u>Authorization</u>. The execution, delivery and performance of the Loan Documents and the transactions contemplated hereby and thereby (i) are within the corporate (or the equivalent company or partnership) authority of each of the Borrowers, (ii) have been duly authorized by all necessary corporate (or other) proceedings, (iii) do not conflict with or result in any material breach or contravention of any provision of law, statute, rule or regulation to which any of the Borrowers is subject or any judgment, order, writ, injunction, license or permit applicable to any of the Borrowers so as to materially adversely affect the assets, business or any activity of the Borrowers, and (iv) do not conflict with any provision of the corporate charter, articles or bylaws (or equivalent other company or partnership documents) of the Borrowers or any agreement or other instrument binding upon the Borrowers, including, without limitation, the Indenture.

(c) **Enforceability**. The execution, delivery and performance of the Loan Documents will result in valid and legally binding obligations of the Borrowers enforceable against each in accordance with the respective terms and provisions hereof and thereof, except as enforceability is limited by bankruptcy, insolvency, reorganization, moratorium or other laws relating to or affecting generally the enforcement of creditors' rights and except to the extent that availability of the remedy of specific performance or injunctive relief or other equitable remedy is subject to the discretion of the court before which any proceeding therefor may be brought.

§6.2. Governmental Approvals. The execution, delivery and performance by the Borrowers of the Loan Documents and the transactions contemplated hereby and thereby do not require any approval or consent of, or filing with, any governmental agency or authority other than those already obtained.

§6.3. <u>Title to Properties</u>; <u>Leases</u>. The Borrowers own all of the assets reflected in the consolidated balance sheets as at the Balance Sheet Date or acquired since that date (except property and assets sold or otherwise disposed of in the ordinary course of business since that date), subject to no mortgages, Capitalized Leases, conditional sales agreements, title retention agreements, liens or other encumbrances except Permitted Liens.

§6.4. Financial Statements; Solvency.

(a) **Financial Statements**. There has been furnished to the Lenders (i) consolidated balance sheets of the Parent and its Subsidiaries dated the Balance Sheet Date and consolidated statements of operations for the fiscal year then ended, certified by Pricewaterhouse Coopers LLP or an independent accounting firm of national standing acceptable to the Lenders (the "<u>Accountants</u>"). Said balance sheets and statements of operations have been prepared in accordance with GAAP, fairly present in all material

respects the financial condition of the Parent and its Subsidiaries, on a consolidated basis as at the close of business on the date thereof and the results of operations for the period then ended. There are no contingent liabilities of the Borrowers as of such dates involving material amounts, known to the officers of the Borrowers which have not been disclosed in said balance sheets and the related notes thereto, as the case may be.

(b) <u>Solvency</u>. The Borrowers (both before and after giving effect to the transactions contemplated by this Credit Agreement, including the issuance of the Senior Subordinated Debt) are and will be solvent (i.e., they have assets having a fair value in excess of the amount required to pay their probable liabilities on their existing debts as they become absolute and matured) and have, and expect to have, the ability to pay their debts from time to time incurred in connection therewith as such debts mature.

§6.5. No Material Changes, Etc. Since the Balance Sheet Date there have occurred no material adverse changes in the financial condition or business of the Borrowers as shown on or reflected in the consolidated balance sheet of the Borrowers as at the Balance Sheet Date or the consolidated statements of income for the periods then ended other than changes in the ordinary course of business which have not had a material adverse effect either individually or in the aggregate on the business or financial condition of the Borrowers. Since the Balance Sheet Date there has not been any Restricted Payment.

§6.6. Permits, Franchises, Patents, Copyrights, Etc. Each of the Borrowers possesses all material franchises, patents, copyrights, trademarks, trade names, licenses and permits, and rights in respect of the foregoing, adequate for the conduct of its business substantially as now conducted without known conflict with any rights of others.

§6.7. Litigation. Except as shown on <u>Schedules 6.7</u> and <u>6.16</u> hereto, there are no actions, suits, proceedings or investigations of any kind pending or, to the knowledge of the Borrowers, threatened against any Borrower before any court, tribunal or administrative agency or board which, if adversely determined, might, either in any case or in the aggregate materially adversely affect the properties, assets, financial condition or business of the Borrowers, considered as a whole, or materially impair the right of the Borrowers, considered as a whole, to carry on business substantially as now conducted, or result in any substantial liability not adequately covered by insurance, or for which adequate reserves are not maintained on the consolidated balance sheet or which question the validity of any of the Loan Documents, or any action taken or to be taken pursuant hereto or thereto and none of the scheduled matters individually or in the aggregate could reasonably be expected to have a material adverse effect on the properties, assets, financial condition or business of the Borrowers as a whole.

<u>§6.8. No Materially Adverse Contracts, Etc.</u> None of the Borrowers is subject to any charter, corporate or other legal restriction, or any judgment, decree, order, rule or regulation which in the judgment of the Borrowers' officers has or is expected in the future to have a materially adverse effect on the business, assets or financial condition of the Borrowers as a whole. None of the Borrowers is a party to any contract or agreement which in the judgment of the Borrowers' officers has or is expected to have any materially adverse effect on the business of the Borrowers as a whole, except as otherwise reflected in adequate reserves.

§6.9. Compliance With Other Instruments, Laws, Etc. None of the Borrowers is violating any provision of its charter documents or by-laws (or equivalent company documents) or any agreement or instrument by which any of them may be subject or by which any of them or any of their properties may be bound or any decree, order, judgment, or any statute, license, rule or regulation, in a manner which could result in the imposition of substantial penalties or materially and adversely affect the financial condition, properties or business of any of the Borrowers.

§6.10. Tax Status. The Borrowers have made or filed all United States federal and state income and all Canadian federal and provincial or territorial income, as applicable, and all other tax returns, reports and declarations required by any jurisdiction to which any of them are subject (unless and only to the extent that any Borrower has set aside on its books provisions reasonably adequate for the payment of all unpaid and unreported taxes); and have paid all taxes and other governmental assessments and charges that are material in amount, shown or determined to be due on such returns, reports and declarations, except those being contested in good faith; and have set aside on their books provisions reasonably adequate for the payment of all taxes for periods subsequent to the periods to which such returns, reports or declarations apply to the extent required in accordance with GAAP. All tax returns, report and declarations required by any jurisdiction accurately disclose (except for discrepancies which are not material) the amount of tax payable by the Borrowers in the relevant jurisdiction except for the amounts being contested in good faith by the Borrowers. There are no unpaid taxes in any material amount claimed to be due by the taxing authority of any jurisdiction, and the officers of the Borrowers know of no basis for any such claim.

<u>§6.11.</u> No Event of Default. No Default or Event of Default has occurred and is continuing.

§6.12. Holding Company and Investment Company Acts. None of the Borrowers is a "holding company", or a "subsidiary company" of a "holding company", or an "affiliate" of a "holding company", as such terms are defined in the Public Utility Holding Company Act of 1935; nor is any of them an "investment company", or an "affiliated company" or a "principal underwriter" of an "investment company", as such terms are defined in the Investment Company Act of 1940, as amended.

§6.13. Absence of Financing Statements, Etc. Except with respect to Permitted Liens and as set forth on <u>Schedule 8.2(f)</u> hereto, there is no effective financing statement, security agreement, chattel mortgage, real estate mortgage or other document filed or recorded with any filing records, registry, or other public office, which covers, affects or gives notice of any present or possible future lien on, or security interest in, any assets or property of any of the Borrowers or rights thereunder.

§6.14. Employee Benefit Plans.

(a) In General. Each Employee Benefit Plan and each Guaranteed Pension Plan has been maintained and operated in compliance in all material respects with the provisions of ERISA and, to the extent applicable, the Code, including but not limited to the provisions thereunder respecting prohibited transactions and the bonding of fiduciaries and other Persons handling plan funds as required by §412 of ERISA. Each

Borrower has heretofore delivered to the Administrative Agent the most recently completed annual report, Form 5500, with all required attachments, and actuarial statement required to be submitted under §103(d) of ERISA, with respect to each Guaranteed Pension Plan.

(b) <u>**Terminability of Welfare Plans.**</u> No Employee Benefit Plan, which is an employee welfare benefit plan within the meaning of \$3(1) or \$3(2)(B) of ERISA, provides benefit coverage subsequent to termination of employment, except as required by Title I, Part 6 of ERISA or the applicable state insurance laws. A Borrower may terminate each such Plan at any time (or at any time subsequent to the expiration of any applicable bargaining agreement) in the discretion of such Borrower without liability to any Person other than for claims arising prior to termination.

(c) Guaranteed Pension Plans. Each contribution required to be made to a Guaranteed Pension Plan, whether required to be made to avoid the incurrence of an accumulated funding deficiency, the notice or lien provisions of §302(f) of ERISA, or otherwise, has been timely made. No waiver of an accumulated funding deficiency or extension of amortization periods has been received with respect to any Guaranteed Pension Plan, and no Borrower nor any ERISA Affiliate is obligated to or has posted security in connection with an amendment to a Guaranteed Pension Plan pursuant to §307 of ERISA or §401(a)(29) of the Code. No liability to the PBGC (other than required insurance premiums, all of which have been paid) has been incurred by any Borrower or any ERISA Affiliate with respect to any Guaranteed Pension Plan and there has not been any ERISA Reportable Event (other than an ERISA Reportable Event as to which the requirement of 30 days notice has been waived), or any other event or condition which presents a material risk of termination of any Guaranteed Pension Plan by the PBGC. Based on the latest valuation of each Guaranteed Pension Plan (which in each case occurred within twelve months of the date of this representation), and on the actuarial methods and assumptions employed for that valuation, the aggregate benefit liabilities of all such Guaranteed Pension Plans within the meaning of §4001 of ERISA did not exceed the aggregate value of the assets of all such Guaranteed Pension Plans, disregarding for this purpose the benefit liabilities and assets of any Guaranteed Pension Plan with assets in excess of benefit liabilities.

(d) <u>Multiemployer Plans.</u> No Borrower nor any ERISA Affiliate has incurred any material liability (including secondary liability) to any Multiemployer Plan as a result of a complete or partial withdrawal from such Multiemployer Plan under §4201 of ERISA or as a result of a sale of assets described in §4204 of ERISA. No Borrower nor any ERISA Affiliate has been notified that any Multiemployer Plan is in reorganization or insolvent under and within the meaning of §4241 or §4245 of ERISA or is at risk of entering reorganization or becoming insolvent, or that any Multiemployer Plan intends to terminate or has been terminated under §4041A of ERISA.

§6.15. Use of Proceeds. The proceeds of the Loans shall be used (a) to refinance the existing Indebtedness of the Borrowers under the Existing Credit Agreement, (b) to fund Permitted Acquisitions, (c) for Capital Expenditures and (d) for working capital and other general corporate purposes. No proceeds of the Loans are to be used, and no portion of any Letter of Credit is to be obtained, in any way that will violate Regulations U or X of the Board of Governors of the Federal Reserve System. The Borrowers will obtain Letters of Credit solely for general corporate purposes.

§6.16. Environmental Compliance. The Borrowers have taken all necessary steps to investigate the past and present condition and usage of the Real Properties and the operations conducted thereon and, based upon such diligent investigation, have determined that, except as shown on Schedule 6.16:

(a) none of the Borrowers or Excluded Subsidiaries, nor any operator of their properties, is in violation, or alleged violation, of any judgment, decree, order, law, permit, license, rule or regulation pertaining to environmental matters, including without limitation, those arising under RCRA, CERCLA, the Superfund Amendments and Reauthorization Act of 1986 ("<u>SARA</u>"), the Federal Clean Water Act, the Federal Clean Air Act, the Toxic Substances Control Act, or any state or local or Canadian federal or provincial statute, regulation, ordinance, order or decree relating to health, safety or the environment (the "<u>Environmental Laws</u>"), which violation would have a material adverse effect on the business, assets or financial condition of the Parent and its Subsidiaries on a consolidated basis; and

(b) (i) except where it would not have a material adverse effect on the business, assets or financial condition of the Borrowers on a consolidated basis, no portion of the Real Property has been used for the handling, processing, storage or disposal of Hazardous Substances and no underground tank or other underground storage receptacle for Hazardous Substances is located on such properties; (ii) in the course of any activities conducted by the Borrowers, or, to the Borrowers' knowledge by any other operators of the Real Property, no Hazardous Substances have been generated or are being used on such properties; and (iii) to the Borrowers' knowledge, there have been no unpermitted Releases or threatened Releases of Hazardous Substances on, upon, into or from the Real Property.

§6.17. Perfection of Security Interests. All filings, assignments, pledges and deposits of documents or instruments have been made or will be made and all other actions have been taken or will be taken that are necessary under applicable law, or reasonably requested by the Administrative Agent or any of the Lenders, to establish and perfect the Administrative Agent's security interests (as collateral agent for the Lenders and the Agents) in the Collateral to the extent required pursuant to §12. The Collateral and the Administrative Agent's rights (as collateral agent for the Lenders and the Agents) with respect to the Collateral are not subject to any setoff, claims, withholdings or other defenses, except for Permitted Liens. The Borrowers are the owners of the Collateral free from any lien, security interest, encumbrance and any other claim or demand, except for Permitted Liens.

§6.18. Certain Transactions. Except as set forth on Schedule 6.18 or as permitted in §8.3, and except for arm's length transactions pursuant to which the Borrowers make payments in the ordinary course of business upon terms no less favorable than the Borrowers could obtain from third parties, none of the officers, directors, or employees of the Borrowers are presently a party to any transaction with the Borrowers (other than for services as employees, officers and directors), including any contract, agreement or other arrangement providing for the furnishing of services to or by, providing for rental of real or personal property to or from, or other wise requiring payments to or from any officer, director, or any such employee has a substantial interest or is an officer, director, trustee

or partner, the value of such transaction, when aggregated with all other such transactions occurring during the term of this Credit Agreement, exceeds \$5,000,000.

§6.19. Subsidiaries. Schedule 1 (as updated from time to time pursuant to §7.19) sets forth a complete and accurate list of the Parent's Subsidiaries, including the name of each Subsidiary and its jurisdiction of incorporation, together with the number of authorized and outstanding shares of Capital Stock of each Subsidiary. Each Subsidiary is directly or indirectly wholly-owned by the Parent. The Parent or a Subsidiary of the Parent has good and marketable title to all of the shares it purports to own of the Capital Stock of each Subsidiary, free and clear in each case of any lien. All such shares of Capital Stock have been duly issued and are fully paid and non-assessable. Each Subsidiary of the Parent, other than the Excluded Subsidiaries, is a Borrower hereunder.

§6.20. Capitalization.

(a) <u>Capital Stock</u>. As of January 21, 2003, the authorized Capital Stock of the Parent consists of (i) 100,000,000 shares of Class A common stock (par value \$.01 per share) authorized of which 22,739,148 shares are outstanding, (ii) 1,000,000 shares of Class B common stock (par value \$.01 per share) authorized of which 28,200 shares are outstanding, and (iii) 1,000,000 shares of preferred stock (par value \$.01 per share) authorized of which 55,750 shares of Series A Preferred Stock are outstanding and held by the Series A Holders. All such outstanding shares of Capital Stock have been duly issued and are fully paid and non-assessable.

(b) **Options, Etc.** As of the Effective Date, except as set forth on <u>Schedule 6.20(b)</u>, no Person has outstanding any rights (either preemptive or other) or options (except for the options for common stock or other forms of equity-based compensation issued to employees, consultants or directors in accordance with a bona fide compensation plan approved by the Board of Directors of the Parent) to subscribe for or purchase from the Parent, or any warrants or other agreements providing for or requiring the issuance by the Parent of, any capital stock or any securities convertible into or exchangeable for its capital stock.

§6.21. True Copies of Charter and Other Documents. The Borrowers have furnished the Administrative Agent copies, in each case true and complete as of the Effective Date, of (a) all charter and other incorporation or constituent documents (together with any amendments thereto) and (b) by-laws (or equivalent company documents) (together with any amendments thereto).

§6.22. Disclosure. No representation or warranty made by the Borrowers in this Credit Agreement or in any agreement, instrument, document, certificate, statement or letter furnished to the Lenders or the Administrative Agent by or on behalf of or at the request of the Borrowers in connection with any of the transactions contemplated by the Loan Documents contains any untrue statement of a material fact or omits to state a material fact necessary in order to make the statements contained therein not misleading in light of the circumstances in which they are made.

§6.23. Guarantees of Excluded Subsidiaries. Except as permitted under §8.1 or §8.3, no Borrower has executed a guarantee with respect to debt incurred by an Excluded Subsidiary.

§6.24. Obligations Constitute "Senior Debt". The Obligations of the Borrowers hereunder are and will continue to be "Senior Debt" and "Designated Senior Debt" under and as defined in the Indenture.

§7. AFFIRMATIVE COVENANTS OF THE BORROWERS. The Borrowers covenant and agree that, so long as any Obligation or any Letter of Credit is outstanding or the Lenders have any obligation to make Loans or the Issuing Lender has any obligation to issue, extend or renew any Letters of Credit hereunder, or the Lenders have any obligations to reimburse the Issuing Lender for drawings honored under any Letter of Credit hereunder:

§7.1. Punctual Payment. Each Borrower will duly and punctually pay or cause to be paid the principal and interest on the Loans, all Reimbursement Obligations, fees and other amounts provided for in this Credit Agreement and the other Loan Documents for which it is liable, all in accordance with the terms of this Credit Agreement and such other Loan Documents.

§7.2. <u>Maintenance of Office</u>. The Borrowers will maintain their chief executive offices at the locations set forth on the Perfection Certificates delivered pursuant to §10.12, or at such other place in the United States of America as each Borrower shall designate upon thirty (30) days' prior written notice to the Administrative Agent.

§7.3. Records and Accounts. Each of the Borrowers will (a) keep true and accurate records and books of account in which full, true and correct entries will be made in accordance with GAAP and with the requirements of all regulatory authorities, (b) maintain adequate accounts and reserves for all taxes (including income taxes), depreciation, depletion, obsolescence and amortization of its properties, all other contingencies, and all other proper reserves and (c) at all times engage Pricewaterhouse Coopers LLP or other independent certified public accountants satisfactory to the Administrative Agent as the independent certified public accountants of the Parent and its Subsidiaries and will not permit more than thirty (30) days to elapse between the cessation of such firm's (or any successor firm's) engagement as the independent certified public accountants of the Parent and its Subsidiaries and the appointment in such capacity of a successor firm as shall be satisfactory to the Administrative Agent.

§7.4. Financial Statements, Certificates and Information. The Borrowers will deliver to the Administrative Agent and each of the Lenders the following:

(a) as soon as practicable, but, in any event not later than ninety (90) days after the end of each fiscal year of the Borrowers, the consolidated balance sheets of the Borrowers and their Subsidiaries as at the end of such year, statements of cash flows, and the related consolidated statements of operations, setting forth in comparative form the figures for the previous fiscal year, all such consolidated financial statements to be in reasonable detail, prepared, in accordance with GAAP and Certified by the Accountants. In addition, simultaneously therewith, the Borrowers will use their best efforts to provide the Lenders with a written statement from such Accountants to the effect that the Borrowers are in compliance with the covenants set forth in §9 hereof, and that, in making the examination necessary to said certification, nothing has come to the attention of such Accountants that would indicate that any Default or Event of Default exists, or, if

such accountants shall have obtained knowledge of any then existing Default or Event of Default they shall disclose in such statement any such Default or Event of Default; <u>provided</u>, that such Accountants shall not be liable to the Lenders for failure to obtain knowledge of any Default or Event of Default;

(b) as soon as practicable, but in any event not later than forty-five (45) days after the end of each fiscal quarter of the Borrowers, copies of the consolidated balance sheets and statement of operations of the Borrowers and their Subsidiaries as at the end of such quarter, subject to year end adjustments, and the related statement of cash flows, all in reasonable detail and prepared in accordance with GAAP with a certification by the principal financial or accounting officer of the Borrowers (the "<u>CFO</u>") that such consolidated financial statements were prepared in accordance with GAAP and fairly present the consolidated financial condition of the Borrowers and their Subsidiaries as at the close of business on the date thereof and the results of operations for the period then ended;

(c) simultaneously with the delivery of the financial statements referred to in (a) and (b) above, (i) a statement in the form of Exhibit <u>C</u> hereto (the "<u>Compliance Certificate</u>") certified by the CFO that the Borrowers are in compliance with the covenants contained in \$7, \$8 and \$9 hereof as of the end of the applicable period setting forth in reasonable detail computations evidencing such compliance, <u>provided</u> that, if the Borrowers shall at the time of issuance of such certificate or at any other time obtain knowledge of any Default or Event of Default, the Borrowers will include in such Compliance Certificate or otherwise deliver forthwith to the Lenders a certificate specifying the nature and period of existence thereof and what action the Borrowers propose to take with respect thereto and attaching, in the event such Default or Event of Default relates to Environmental Matters, a certificate in the form attached hereto as <u>Exhibit D</u> (the "<u>Environmental Compliance Certificate</u>");

(d) contemporaneously with, or promptly following, the filing or mailing thereof, copies of all material of a financial nature filed with the Securities and Exchange Commission or sent to the stockholders of the Parent or any of the Borrowers to the extent the same are not available on EDGAR;

(e) simultaneously with the delivery of the financial statements referred to in (a) and (b) above, copies of the Borrowers' revenue, EBITDA and pre-tax reports, all in reasonable detail and prepared in accordance with GAAP;

(f) as soon as practicable, but in any event not later than fifteen (15) days prior to the commencement of each fiscal year of the Borrowers and the Excluded Subsidiaries, a copy of the annual budget, projections and business plan for the Borrowers and the Excluded Subsidiaries for such fiscal year; and

(g) from time to time such other financial data and other information (including accountants' management letters) as the Lenders may reasonably request.

The Borrowers hereby authorize the Lenders to disclose any information obtained pursuant to this Credit Agreement to all appropriate governmental regulatory authorities where required by law; <u>provided</u>, <u>however</u>, that the Lenders shall, to the extent practicable and allowable under law, notify the Borrowers within a reasonable period

prior to the time any such disclosure is made; and <u>provided further</u>, this authorization shall not be deemed to be a waiver of any rights to object to the disclosure by the Lenders of any such information which any Borrower has or may have under the federal Right to Financial Privacy Act of 1978, as in effect from time to time.

§7.5. Legal Existence and Conduct of Business. Except where the failure of a Borrower to remain so qualified would not materially adversely impair the financial condition of the Borrowers on a consolidated basis, each Borrower will do or cause to be done all things necessary to preserve and keep in full force and effect its legal existence, legal rights and franchises; effect and maintain its foreign qualifications, licensing, domestication or authorization except as terminated by its Board of Directors in the exercise of its reasonable judgment; use its best efforts to comply with all applicable laws; and shall not become obligated under any contract or binding arrangement which, at the time it was entered into would materially adversely impair the financial condition of the Borrowers, on a consolidated basis. Each Borrower will continue to engage primarily in the businesses now conducted by it and in related businesses.

§7.6. Maintenance of Properties. The Borrowers will cause all material properties used or useful in the conduct of their businesses to be maintained and kept in good condition, repair and working order and supplied with all necessary equipment and will cause to be made all necessary repairs, renewals, replacements, betterments and improvements thereof, all as in the judgment of the Borrowers may be necessary so that the businesses carried on in connection therewith may be properly and advantageously conducted at all times; provided, however, that nothing in this section shall prevent any Borrower from discontinuing the operation and maintenance of any of its properties if such discontinuance is, in the judgment of such Borrower, desirable in the conduct of its or their business and which does not in the aggregate materially adversely affect the business of the Borrowers on a consolidated basis.

§7.7. Insurance. The Borrowers will maintain with financially sound and reputable insurance companies, funds or underwriters' insurance, including self-insurance, of the kinds, covering the risks and in the relative proportionate amounts usually carried by reasonable and prudent companies conducting businesses similar to that of the Borrowers. In addition, the Borrowers will furnish from time to time, upon the Administrative Agent's request, a summary of the insurance coverage of each of the Borrowers, which summary shall be in form and substance satisfactory to the Administrative Agent and, if requested by the Administrative Agent, will furnish to the Administrative Agent copies of the applicable policies naming the Administrative Agent as a loss payee thereunder.

§7.8. Taxes. The Borrowers will each duly pay and discharge, or cause to be paid and discharged, before the same shall become overdue, all taxes, assessments and other governmental charges (other than taxes, assessments and other governmental charges imposed by jurisdictions other than the United States or Canada or a political division thereof which in the aggregate are not material to the business or assets of any Borrower on an individual basis or of the Borrowers on a consolidated basis) imposed upon it and its real properties, sales and activities, or any part thereof, or upon the income or profits therefrom, as well as all claims for labor, materials, or supplies, which if unpaid might by law become a lien or charge upon any of its property; provided, however, that

any such tax, assessment, charge, levy or claim need not be paid if the validity or amount thereof shall currently be contested in good faith by appropriate proceedings and if such Borrower shall have set aside on its books adequate reserves with respect thereto; and <u>provided</u>, <u>further</u>, that such Borrower will pay all such taxes, assessments, charges, levies or claims forthwith upon the commencement of proceedings to foreclose any lien which may have attached as security therefor.

§7.9. Inspection of Properties, Books, and Contracts. The Borrowers shall permit the Lenders, the Agents or any of their designated representatives, upon reasonable notice, to visit and inspect any of the properties of the Borrowers, to examine the books of account of the Borrowers (including the making of periodic accounts receivable reviews), or contracts (and to make copies thereof and extracts therefrom), and to discuss the affairs, finances and accounts of the Borrowers with, and to be advised as to the same by, their officers, all at such times and intervals as the Lenders or the Agents may reasonably request.

§7.10. Compliance with Laws, Contracts, Licenses and Permits; Maintenance of Material Licenses and Permits. Each Borrower will, and will cause the Excluded Subsidiaries to, (a) comply with the provisions of its charter documents, articles of incorporation, other constituent documents and by-laws and all agreements and instruments by which it or any of its properties may be bound; (b) comply with all applicable laws and regulations (including Environmental Laws), decrees, orders, judgments, licenses and permits, including, without limitation, all environmental permits hereto ("Applicable Laws"), except where noncompliance with such Applicable Laws would not have a material adverse effect in the aggregate on the consolidated financial condition, properties or businesses of the Borrowers and the Excluded Subsidiaries; (c) comply in all material respects with all agreements and instruments by which it or any of its properties may be bound; (d) maintain all material operating permits for all landfills now owned or hereafter acquired; and (e) dispose of hazardous waste only at licensed disposal facilities operating, to the best of such Borrower's knowledge after reasonable inquiry, in compliance with Environmental Laws. If at any time while the Notes, any Loan or Letter of Credit is outstanding or any Lender, the Issuing Lender or any Agent has any obligation to make Loans or issue Letters of Credit hereunder, any authorization, consent, approval, permit or license from any officer, agency or instrumentality of any government shall become necessary or required in order that any Borrower may fulfill any of its obligations hereunder, such Borrower will immediately take or cause to be taken all reasonable steps within the power of such Borrower to obtain such authorization, consent, approval, permit or license and furnish the Lenders with evidence thereof.

§7.11. Environmental Indemnification. The Borrowers covenant and agree that they will jointly and severally, in accordance with §5.9, indemnify and hold the Agents, the Issuing Lender and the Lenders, and their respective affiliates, agents, directors, officers and shareholders, harmless from and against any and all claims, expense, damage, loss or liability incurred by such indemnified parties (including all costs of legal representation incurred by such indemnified parties) relating to (a) any Release or threatened Release of Hazardous Substances on the Real Property; (b) any violation of any Environmental Laws with respect to conditions at the Real Property or the operations conducted thereon; or (c) the investigation or remediation of offsite locations at which the Borrowers, or their predecessors are alleged to have directly or

indirectly Disposed of Hazardous Substances. It is expressly acknowledged by the Borrowers that this covenant of indemnification shall survive any foreclosure or any modification, release or discharge of any or all of the Security Documents or the payment of the Loans and shall inure to the benefit of the Agents and the Lenders and their respective successors and assigns.

§7.12. Further Assurances. The Borrowers will cooperate with the Lenders and the Agents and execute such further instruments and documents as the Lenders or the Agents shall reasonably request to carry out to their satisfaction the transactions contemplated by this Credit Agreement.

§7.13. Notice of Potential Claims or Litigation. The Borrowers shall deliver to the Lenders and the Agents, within thirty (30) days of receipt thereof, written notice of the initiation of any action, claim, complaint, or any other notice of dispute or potential litigation (including without limitation any alleged violation of any Environmental Law), wherein the potential liability is in excess of \$2,500,000, or could otherwise reasonably be expected to have a material adverse effect on the business of the Borrowers as a whole, together with a copy of each such notice received by any Borrower or the Excluded Subsidiaries.

§7.14. Notice of Certain Events Concerning Insurance and Environmental Claims.

(a) The Borrowers will provide the Lenders and the Agents with written notice as to any material cancellation or material adverse change in any insurance of any of the Borrowers within ten (10) Business Days after such Borrower's receipt of any notice (whether formal or informal) of such material cancellation or material change by any of its insurers.

- (b) The Borrowers will promptly notify the Lenders and the Agents in writing of any of the following events:
- (i) upon any Borrower's obtaining knowledge of any violation of any Environmental Law which violation could have a material adverse effect on the business, financial condition, or assets of the Borrowers on a consolidated basis;
- upon any Borrower's obtaining knowledge of any potential or known Release, or threat of Release, of any Hazardous Substance at, from, or into the Real Property which could materially affect the business, financial condition, or assets of the Borrowers on a consolidated basis;
- (iii) upon any Borrower's receipt of any notice of any material violation of any Environmental Law or of any Release or threatened Release of Hazardous Substances, including a notice or claim of liability or potential responsibility from any third party (including any federal, state, provincial, territorial or local governmental officials) and including notice of any formal inquiry, proceeding, demand, investigation or other action with regard to (A) any Borrower's or any Person's operation of the Real Property, (B) the presence or Release of Hazardous Substances on,

from, or into the Real Property, or (C) investigation or remediation of offsite locations at which any Borrower or its predecessors are alleged to have directly or indirectly Released Hazardous Substances, and with respect to which the liability associated therewith could be reasonably expected to exceed \$2,500,000; or

(iv) upon any Borrower's obtaining knowledge that any expense or loss which individually or in the aggregate exceeds \$1,000,000 has been incurred by such governmental authority in connection with the assessment, containment, removal or remediation of any Hazardous Substances with respect to which any Borrower may be liable or for which a lien may be imposed on the Real Property.

§7.15. Notice of Default. The Borrowers will promptly notify the Lenders and the Agents in writing of the occurrence of any Default or Event of Default. If any Person shall give any notice or take any other action in respect of a claimed default (whether or not constituting an Event of Default) under this Credit Agreement or any other note, evidence of Indebtedness, indenture or other obligation evidencing Indebtedness in excess of \$1,000,000 (including, without limitation, the Indenture) as to which any Borrower is a party or obligor, whether as principal or surety, the Borrowers shall forthwith give written notice thereof to the Lenders and the Agents, describing the notice of action and the nature of the claimed default.

<u>§7.16.</u> Closure and Post Closure Liabilities. The Borrowers shall at all times adequately accrue, in accordance with GAAP, and fund, as required by applicable Environmental Laws, all closure and post closure liabilities with respect to the operations of the Borrowers.

§7.17. Subsidiaries. The Parent shall at all times directly or indirectly through a Subsidiary own all of the shares of the Capital Stock of each Subsidiary of the Parent.

§7.18. Interest Rate Protection. The Borrowers will, within ninety (90) days of the Effective Date, have a minimum aggregate amount of not less than 30% of the notional amount of Consolidated Total Funded Debt as of the Effective Date on a fixed rate long term basis (whether through Swap Contracts or as a result of having a fixed rate of interest by its terms) on terms and conditions reasonably acceptable to the Administrative Agent.

§7.19. Additional Borrowers. To the extent that such creation or acquisition is permitted under this Credit Agreement, all newlycreated or newly-acquired Subsidiaries (other than Excluded Subsidiaries) shall become Borrowers hereunder by signing allonges to the Notes, entering into a joinder and affirmation to this Credit Agreement in substantially the form of Exhibit F attached hereto (a "Joinder Agreement") providing that such Subsidiary shall become a Borrower hereunder, and providing such other documentation as the Lenders or the Administrative Agent may reasonably request including, without limitation, documentation with respect to conditions noted in §10 hereof. In such event, the Administrative Agent is hereby authorized by the parties to amend <u>Schedule 1</u> hereto to include such Subsidiary as a Borrower hereunder.

§8. CERTAIN NEGATIVE COVENANTS OF THE BORROWERS. The Borrowers covenant and agree that, so long as any Obligation or any Letter of Credit is outstanding or the Lenders have any obligation to make Loans or the Issuing Lender has any obligation to issue, extend or renew any Letters of Credit hereunder, or the Lenders have any obligations to reimburse the Issuing Lender for drawings honored under any Letter of Credit hereunder:

§8.1. <u>Restrictions on Indebtedness</u>. None of the Borrowers or Excluded Subsidiaries shall become or be a guarantor or surety of, or otherwise create, incur, assume, or be or remain liable, contingently or otherwise, with respect to any Indebtedness, or become or be responsible in any manner (whether by agreement to purchase any obligations, stock, assets, goods or services, or to supply or advance any funds, assets, goods or services or otherwise) with respect to any undertaking or Indebtedness of any other Person, or incur any Indebtedness other than:

(a) Indebtedness of the Borrowers to the Lenders, the Issuing Lender and the Agents arising under this Credit Agreement and the Loan Documents;

(b) Subject to §8.9, Seller Subordinated Debt not to exceed \$15,000,000 in aggregate outstanding principal amount at any time;

(c) Existing Indebtedness of the Borrowers with respect to loans and Capitalized Leases listed on <u>Schedule 8.1(c)</u>, on the terms and conditions in effect as of the date hereof, together with any renewals, extensions or refinancings thereof on terms which are not materially different than those in effect as of the Effective Date;

(d) Endorsements for collection, deposit or negotiation and warranties of products or services (including unsecured performance and payment bonds ("<u>Performance Bonds</u>")), in each case incurred in the ordinary course of business;

(e) Indebtedness of the Borrowers incurred in connection with the acquisition or lease of any equipment by the Borrowers under any Synthetic Lease, Capitalized Lease or other lease arrangement or purchase money financing; provided that the aggregate outstanding principal amount of such Indebtedness of the Borrowers (including Indebtedness of such type listed on <u>Schedule 8.1(c)</u>) shall not exceed \$30,000,000 at any time;

(f) Indebtedness of the Borrowers in respect of Swap Contracts satisfactory to the Administrative Agent;

(g) Indebtedness of the Borrowers under fuel price swaps, fuel price caps, and fuel price collar or floor agreements, and similar agreements or arrangements designed to protect against or manage fluctuations in fuel prices with respect to fuel purchased in the ordinary course of business of the Borrowers ("<u>Fuel Derivatives Obligations</u>"), <u>provided</u> that the aggregate notional amount of such agreements do not exceed \$10,000,000 outstanding at any time, the maturity of such agreements do not exceed thirty-six (36) months, and the terms are consistent with past practices of the Borrowers;

(h) Other unsecured Indebtedness incurred in connection with the acquisition by the Borrowers of real or personal property, including any Indebtedness incurred with respect to non-compete payments in connection with such acquisition(s), provided that the aggregate outstanding principal amount of such Indebtedness of the Borrowers shall not exceed \$15,000,000 at any time;

(i) Intercompany Indebtedness among the Borrowers;

(j) Indebtedness with respect to mandatory redemption obligations as set forth in the Series A Certificate and accrued dividends on the Borrower's preferred stock; provided that no Restricted Payments shall be made with respect to such Indebtedness during the term of this Credit Agreement except as set forth in §8.6 hereof, or as otherwise permitted by the prior written consent of the Required Lenders;

(k) Senior Subordinated Debt not to exceed \$250,000,000 in aggregate principal amount;

(l) Surety and similar bonds and completion bonds and bid guarantees provided by or issued on behalf of the Borrowers with respect to the closure, final-closure and post-closure liabilities related to landfills owned or operated by the Borrowers; <u>provided</u> that the aggregate amount of such Indebtedness shall not exceed \$70,000,000 at any time outstanding;

(m) indemnification, adjustment of purchase price or similar obligations, in each case, incurred or assumed in connection with the Permitted Acquisitions or permitted dispositions of Capital Stock or assets of the Borrowers; <u>provided</u> that the maximum aggregate liability in respect of all such obligations shall at no time exceed the gross proceeds, including non-cash proceeds, (the fair market value of such non-cash proceeds being measured at the time received or paid and without giving effect to any subsequent changes in value) actually received or paid by the Borrowers in connection with such Permitted Acquisition or disposition;

(n) Indebtedness arising in connection with (i) the acquisition by the Parent of the Capital Stock of Hardwick Landfill, Inc., a Massachusetts corporation ("<u>Hardwick</u>"), pursuant to the terms of that certain Stock Purchase Agreement, dated as of January 3, 2003 (the "<u>Hardwick</u> <u>Purchase Agreement</u>"), by and among the Parent, Hardwick and the shareholders of Hardwick; <u>provided</u> that the original principal amount of such Indebtedness does not exceed \$2,000,000, and (ii) the exercise by any Borrower of the option to purchase the Capital Stock or the assets of Roach Enterprises, LLC pursuant to the terms and conditions set forth in the Purchase Option Agreement attached as <u>Exhibit D</u> to the Hardwick Purchase Agreement (the "<u>Hardwick Option Agreement</u>"); provided that the original principal amount of such Indebtedness does not exceed 25% of the Base Purchase Price (as defined in the Hardwick Option Agreement), and is calculated as set forth in the applicable provisions of the Hardwick Option Agreement; and

(o) Guarantees of Indebtedness permitted pursuant to this §8.1 made by any of the Borrowers or their Subsidiaries, the amount of such guarantees not to exceed the amount of the underlying Indebtedness.

§8.2. Restrictions on Liens. None of the Borrowers or Excluded Subsidiaries will create or incur or suffer to be created or incurred or to exist any lien, encumbrance, mortgage, pledge, negative pledge, charge, restriction or other security interest of any kind upon any property or assets of any character, whether now owned or hereafter acquired, or upon the income or profits therefrom; or transfer any of such property or assets or the income or profits therefrom for the purpose of subjecting the same to the payment of Indebtedness or performance of any other obligation in priority to payment of its general creditors; or acquire, or agree or have an option to acquire, any property or assets upon conditional sale or other title retention or purchase money security agreement, device or arrangement; or suffer to exist for a period of more than thirty (30) days after the same shall have been incurred any Indebtedness or claim or demand against it which if unpaid might by law or upon bankruptcy or insolvency, or otherwise, be given any priority whatsoever over its general creditors; or sell, assign, pledge or otherwise transfer any accounts, contract rights, general intangibles or chattel paper, with or without recourse, except as follows (the "Permitted Liens"):

(a) Liens on property to secure Indebtedness permitted under §8.1(e) hereof, provided that such Liens (i) shall encumber only the specific equipment being financed or leased, (ii) shall not exceed the fair market value thereof and (iii) shall not encumber property with an aggregate value in excess of \$30,000,000;

(b) Liens to secure taxes, assessments and other government charges or claims for labor, material or supplies in respect of obligations not overdue and government liens in existence less than 90 days from the date of creation thereof to secure taxes, assessments, charges, levies or claims being contested in good faith by appropriate proceedings if the Borrower shall have set aside on its books adequate reserves with respect thereto;

(c) Deposits or pledges made in connection with, or to secure payment of, workmen's compensation, unemployment insurance, old age pensions or other social security obligations;

(d) Liens of carriers, warehousemen, mechanics and materialmen, and other like liens, in existence less than 120 days from the date of creation thereof in respect of obligations not overdue;

(e) Encumbrances consisting of easements, rights of way, zoning restrictions, restrictions on the use of Real Property and defects and irregularities in the title thereto, landlord's or lessor's liens under leases to which any Borrower is a party, and other minor liens or encumbrances none of which in the opinion of the respective Borrower interferes materially with the use of the property affected in the ordinary conduct of the business of such Borrower, which defects do not individually or in the aggregate have a material adverse effect on the business of such Borrower individually or of the Borrowers on a consolidated basis;

(f) Liens existing as of the date hereof securing Indebtedness permitted under §8.1(c) hereof and listed on <u>Schedule 8.2(f)</u> hereto;

(g) Liens granted pursuant to the Security Documents to secure the Obligations (provided that secured Obligations hereunder with respect to Fuel Derivatives Obligations with Lenders shall not exceed \$10,000,000 in the aggregate); and

(h) Liens granted (i) on the Capital Stock of Hardwick to secure the Indebtedness permitted under §8.1(n)(i) and the contingent royalty payment obligations of the Parent under the Hardwick Purchase Agreement; <u>provided</u> that such Liens shall be subordinated and second in priority to the Administrative Agent's first priority Liens on such Capital Stock pursuant to the provisions of the Collateral Pledge Agreement attached as <u>Exhibit E</u> to the Hardwick Purchase Agreement and a Subordination Agreement in the form of <u>Exhibit E</u> to the Existing Credit Agreement, with such changes as the Administrative Agent has previously approved, (ii) on the Capital Stock of Roach Enterprises, LLC or the Borrower exercising the option under the Hardwick Option Agreement and securing the Indebtedness permitted under §8.1(n)(ii) and the contingent royalty payment obligations of such Borrower under the Hardwick Option Agreement, and (iii) on landfills acquired by a Borrower securing the landfill royalty payment obligations of such Borrower so long as the Administrative Agent shall have been granted a first mortgage on such landfills; <u>provided</u>, that, in the case of each of clause (ii) and (iii) above, such Liens shall be subordinated and second in priority to the Administrative Agent's first priority Liens on the Capital Stock or landfills, as applicable, on terms and conditions satisfactory in all respects to the Administrative Agent.

§8.3. <u>Restrictions on Investments.</u> None of the Borrowers shall make or permit to exist or to remain outstanding any other Investment other than:

(a) Investments in obligations of the United States of America or Canada and agencies thereof and obligations guaranteed by the United States of America or Canada that are due and payable within one (1) year from the date of acquisition;

(b) certificates of deposit, time deposits, bankers' acceptances or repurchase agreements which are fully insured or are issued by commercial banks organized under the laws of the United States of America or any state thereof or Canada and having total assets in excess of \$1,000,000,000;

(c) commercial paper maturing not more than nine (9) months from the date of issue, <u>provided</u> that, at the time of purchase, such commercial paper is not rated lower than "P-1" by Moody's or "A-1" by S&P;

(d) Investments associated with insurance policies required or allowed by state or provincial law to be posted as financial assurance for landfill closure and post-closure liabilities;

(e) Investments by any Borrower in any wholly-owned Subsidiary which is also a Borrower;

(f) Investments existing on the Effective Date and listed on <u>Schedule 8.3(f)</u> hereto;

(g) any money market account, short-term asset management account or similar investment account maintained with one of the Lenders;

- (h) loans made to employees in an aggregate amount not to exceed \$2,000,000 at any time outstanding;
- (i) up to \$10,000,000 in Investments in the Insurance Subsidiary at any time outstanding;
- (j) Existing Investments in the Excluded Subsidiaries (other than the Insurance Subsidiary) listed on <u>Schedule 8.3(f)</u> hereto;

(k) Investments made after the Effective Date in the Cellulose Joint Venture and the New Heights Investment to the extent that the aggregate amount of such Investments does not exceed the aggregate amount of actual cash dividends and cash partnership or limited liability company distributions received by the Borrowers therefrom since the Effective Date;

(1) Investments in the form of Permitted Acquisitions permitted pursuant to §8.4.1; and

(m) other Investments not to exceed \$15,000,000 in the aggregate at any time outstanding (including, without limitation, Investments made after the Effective Date in the Cellulose Joint Venture and the New Heights Investment in excess of actual cash dividends and partnership or limited liability company distributions received by the Borrowers from the Cellulose Joint Venture and the New Heights Investment after the Effective Date);

provided; that none of the Borrowers shall make or permit to exist or to remain outstanding any Investment in any Subsidiary unless both before and after giving effect thereto there does not exist a Default or Event of Default and no Default or Event of Default would be created by the making of such Investment.

§8.4. Mergers, Consolidations, Sales.

§8.4.1. <u>Mergers and Acquisitions</u>. The Borrowers will not become a party to any merger, amalgamation, or consolidation, or agree to or effect any asset acquisition or stock acquisition (other than the acquisition of assets in the ordinary course of business consistent with past practices) except the merger or consolidation of, or asset or stock acquisitions between, existing Borrowers and except as otherwise provided in this §8.4.1. The Borrowers may purchase or otherwise acquire all or substantially all of the assets or stock or other equity interests of any other Person (a "Permitted Acquisition") provided that:

(a) the Borrowers are in current compliance with and, giving effect to the proposed acquisition (including any borrowings made or to be made in connection therewith), will continue to be in compliance with all of its covenants and agreements contained in this Credit Agreement, including the financial covenants in §9 hereof on a pro forma historical combined basis as if the transaction occurred on the first day of the period of measurement;

(b) at the time of such acquisition, no Default or Event of Default has occurred and is continuing, and such acquisition will not otherwise create a Default or an Event of Default hereunder;

(c) the business to be acquired is predominantly in the same lines of business as the Borrowers, or businesses reasonably related or incidental thereto (e.g., non-hazardous solid waste collection, transfer, hauling, recycling, or disposal);

(d) the business to be acquired operates predominantly in the United States or Canada;

(e) (i) in the case of an asset acquisition, all of the assets acquired shall be acquired by an existing Borrower or a newly-created Subsidiary of the Parent, which Subsidiary shall become a Borrower hereunder in accordance with §7.19, and 100% of its Capital Stock and its assets shall be pledged simultaneously with such acquisition to the Administrative Agent for the benefit of the Lenders and the Agents or, (ii) in the case of an acquisition of Capital Stock, the acquired company, simultaneously with such acquisition, shall become a Borrower in accordance with §7.19 and 100% of its Capital Stock and its assets shall be pledged simultaneously with such acquisition to the Administrative Agent for the benefit of the Lenders and the Agents or the acquired company shall be merged or amalgamated with and into a wholly-owned Subsidiary that is a Borrower and such newly-acquired or newly-created Subsidiary shall otherwise comply with the provisions of §7.19 hereof;

(f) if the total consideration in connection with any such acquisition, including the aggregate amount of all liabilities assumed, but excluding the payment of all fees and expenses relating to such purchase, exceeds \$5,000,000 (a "<u>Material Acquisition</u>"), then not later than seven (7) days prior to the proposed acquisition date, the Borrowers shall furnish the Administrative Agent with (i) a copy of the purchase agreement, (ii) its audited (if available, or otherwise unaudited) financial statements for the preceding two (2) fiscal years or such shorter period of time as such entity or division has been in existence, (iii) a summary of the Borrowers' results of their standard due diligence review, (iv) in the case of a landfill acquisition or if the target company owns a landfill, a review by a Consulting Engineer and a copy of the Consulting Engineer's report, (v) a Compliance Certificate demonstrating compliance with §9 on a pro forma historical combined basis as if the transaction occurred on the first day of the period of measurement, (vi) written evidence that the board of directors and (if required by applicable law) the shareholders, or the equivalent thereof, of the business to be acquired have approved such acquisition, and (vii) such other information as the Administrative Agent may reasonably request, which in each case shall be in form and substance acceptable to the Administrative Agent;

(g) the board of directors and (if required by applicable law) the shareholders, or the equivalent thereof, of the business to be acquired shall have approved such acquisition;

(h) if such acquisition is made by a merger or amalgamation, a Borrower, or a wholly-owned Subsidiary of the Parent (which may be the acquired company) which shall become a Borrower in connection with such merger, shall be the surviving entity; and

(i) cash consideration to be paid by any Borrower in connection with any acquisition or series of related acquisitions (including cash deferred payments, contingent or otherwise, and the aggregate amount of all liabilities assumed or, in the case of a stock acquisition, including all liabilities of the target company) shall not exceed \$15,000,000 without the consent of the Administrative Agent and the Required Lenders.

§8.4.2. Dispositions of Assets. Except as otherwise provided in this §8.4.2, none of the Borrowers will become a party to or agree to or effect any disposition of assets; provided that, subject to §4.4.1, so long as no Default or Event of Default has occurred and is continuing, during the term of this Credit Agreement, the Borrowers may (a) sell or transfer assets (including in connection with an asset swap) having an aggregate fair market value not in excess of 5% of Consolidated Total Assets (the "Basket"), for fair and reasonable value, as determined by the board of directors of the Parent in good faith and evidenced by a resolution of such directors which shall be delivered by the Parent to the Administrative Agent prior to the consummation of such sale or transfer, and, in the case of an asset swap, so long as such asset swap in the reasonable business judgement of the Parent does not have a material adverse effect on the business or financial condition of the Borrowers and (b) sell the Capital Stock or assets of the Specified Entities. Notwithstanding the foregoing, the sale of inventory, the licensing of intellectual property and the disposition of obsolete assets or assets that are no longer useful, in each case in the ordinary course of business consistent with past practices, are permitted hereunder without being charged against the Basket.

§8.5. Sale and Leaseback. None of the Borrowers shall enter into any arrangement, directly or indirectly, whereby any Borrower shall sell or transfer any property owned by it in order then or thereafter to lease such property or lease other property which such Borrower intends to use for substantially the same purpose as the property being sold or transferred, without the prior written consent of the Required Lenders.

§8.6. Restricted Payments. None of the Borrowers will make any Restricted Payments except that, (a) any Subsidiary may declare or pay cash Distributions to the Parent, (b) the Parent may cause quarterly Distributions on its preferred stock (including the Series A Preferred Stock) to accrue and be added to the liquidation value of such preferred stock, and (c) the Parent may convert all or a portion of the Series A Preferred Stock into shares of its common stock; provided, however, that such conversion shall not be made unless permitted under the terms of the Senior Subordinated Debt Documents. In addition, except as otherwise expressly permitted in this §8.6, the Borrowers shall not prepay, redeem, convert, retire, repurchase or otherwise acquire shares of any class of Capital Stock (including the Series A Preferred Stock) of the Borrowers or Excluded Subsidiaries without the prior written consent of the Administrative Agent and the Required Lenders.

§8.7. Employee Benefit Plans. None of the Borrowers nor any ERISA Affiliate will:

(a) engage in any "prohibited transaction" within the meaning of §406 of ERISA or §4975 of the Code which could result in a material liability for any Borrower; or

(b) permit any Guaranteed Pension Plan to incur an "accumulated funding deficiency", as such term is defined in §302 of ERISA, whether or not such deficiency is or may be waived; or

(c) fail to contribute to any Guaranteed Pension Plan to an extent which, or terminate any Guaranteed Pension Plan in a manner which, could result in the imposition of a lien or encumbrance on the assets of any Borrower pursuant to \$302(f) or \$4068 of ERISA; or

(d) amend any Guaranteed Pension Plan in circumstances requiring the posting of security pursuant to §307 of ERISA or §401(a)(29) of the Code;

(e) permit or take any action which would result in the aggregate benefit liabilities (with the meaning of §4001 of ERISA) of all Guaranteed Pension Plans exceeding the value of the aggregate assets of such Plans, disregarding for this purpose the benefit liabilities and assets of any such Plan with assets in excess of benefit liabilities.

The Borrowers will (i) promptly upon filing the same with the Department of Labor or Internal Revenue Service, furnish to the Lenders a copy of the most recent actuarial statement required to be submitted under §103(d) of ERISA and Annual Report, Form 5500, with all required attachments, in respect of each Guaranteed Pension Plan and (ii) promptly upon receipt or dispatch, furnish to the Lenders any notice, report or demand sent or received in respect of a Guaranteed Pension Plan under §§302, 4041, 4042, 4043, 4063, 4065, 4066 and 4068 of ERISA, or in respect of a Multiemployer Plan, under §§4041A, 4202, 4219, or 4245 of ERISA.

§8.8. Prepayments of Certain Obligations: Modifications of Subordinated Debt. None of the Borrowers will, nor will they permit their Subsidiaries to, (a) amend, supplement or otherwise modify the terms of any Subordinated Debt; provided, that the Borrowers may amend, supplement or otherwise modify the terms of any Subordinated Debt with the consent of the Administrative Agent if, in the judgment of the Administrative Agent, such amendments, supplements or modifications do not adversely effect the rights of the Lenders, or (b) prepay, redeem or repurchase or issue any notice or offer of redemption with respect to, elect to make, or effect, a defeasance with respect to, or take any other action which would require the Borrowers or any of their Subsidiaries to, prepay, redeem or repurchase any of the Subordinated Debt, (c) make any payments with respect to any Seller Subordinated Debt other than scheduled payments of principal and interest as and to the extent permitted under the applicable Subordination Agreements, provided that no Default or Event of Default shall have occurred or be continuing on the date of such payment, nor would be created by the making of such payment, or (d) make any payments with respect to any Senior Subordinated Debt other than scheduled payments of interest as and to the extent

permitted under the Indenture, provided that no Default or Event of Default shall have occurred or be continuing on the date of such payment, nor would be created by the making of such payment.

§8.9. Negative Pledges and Upstream Limitations. The Borrowers will not, nor will they permit their Subsidiaries to (a) enter into or permit to exist any agreement or arrangement (excluding this Credit Agreement, the other Loan Documents and the Indenture) which directly or indirectly prohibits the creation or assumption or incurrence by the Borrowers or their Subsidiaries of any lien or security interest upon their properties (other than prohibitions on liens for particular assets set forth in a security instrument in connection with secured Indebtedness permitted by §8.1(e) to the extent such prohibition relates only to such assets and the granting or effect of such liens does not otherwise constitute a Default or Event of Default), revenues or assets, whether now owned or hereafter acquired, or (b) enter into any agreement, contract or arrangement (excluding this Credit Agreement, the other Loan Documents and the Indenture) restricting the ability of (i) the Borrowers to amend or modify this Credit Agreement or any other Loan Document, or (ii) any Borrower or its Subsidiaries to pay or make dividends or distributions in cash or kind to any Borrower or to make loans, advances or other payments of whatsoever nature to any Borrower or to make transfers or distributions of all or any part of such Borrower's or such Subsidiary's assets to a Borrower; in each case other than (x) restrictions on specific assets which assets are the subject of purchase money security interests to the extent permitted under §8.1(e), and (y) customary anti-assignment provisions contained in leases and licensing agreements entered into by such Borrower or such Subsidiary in the ordinary course of its business.

§8.10. Transactions with Affiliates. The Borrowers will not, and will not permit any of their Subsidiaries to, engage in any transaction with any Affiliate (other than for services as employees, officers and directors), including any contract, agreement or other arrangement providing for the furnishing of services to or by, providing for rental of real or personal property to or from, or otherwise requiring payments to or from any such Affiliate or, to the knowledge of the Borrowers, any corporation, partnership, trust or other entity in which any such Affiliate has a substantial interest or is an officer, director, trustee or partner, on terms more favorable to such Person than would have been obtainable on an arm's-length basis in the ordinary course of business.

§8.11. Business Activities. The Borrowers will not, and will not permit any of their Subsidiaries to, engage directly or indirectly (whether through Subsidiaries or otherwise) in any type of business other than the businesses conducted by them on the Effective Date and in related businesses or in connection with the acquisition of Greenfiber.

§8.12. No Other Senior Debt. The Borrowers (a) have not designated, and will not designate, any Indebtedness of the Borrowers or any of their Subsidiaries as "Designated Senior Debt" for purposes of (and as defined in) the Indenture, other than the Obligations, and (b) have no "Senior Debt" as such term is defined in the Indenture other than the Obligations and the Indebtedness permitted under §8.1 which ranks pari passu with the Obligations.

§8.13. Actions Otherwise Prohibited by Subordinated Debt. Notwithstanding anything contained in this §8 that permits the

Borrowers or any

of their Subsidiaries to enter into transactions or take certain actions, the Borrowers shall not enter into such transactions or take such actions if otherwise prohibited from so doing by the terms of the Senior Subordinated Debt outstanding from time to time.

§9. FINANCIAL COVENANTS.

The Borrowers covenant and agree that, so long as any Obligation or any Letter of Credit is outstanding or the Lenders have any obligation to make Loans, or the Issuing Lender has any obligation to issue, extend or renew any Letters of Credit hereunder, or the Lenders have any obligations to reimburse the Issuing Lender for drawings honored under any Letter of Credit hereunder:

§9.1. Interest Coverage Ratio. As at the end of any fiscal quarter ended on or during any period set forth in the table below, the ratio of (a) Consolidated EBITDA for the period of four (4) consecutive fiscal quarters ended on the date of calculation to (b) Consolidated Total Interest Expense for such period shall not be less than the stated ratio set forth opposite such period in such table:

Period	Interest Coverage Ratio
Effective Date through April 30, 2005	2.60:1.00
May 1, 2005 and thereafter	2.70:1.00

§9.2. Profitable Operations. The Borrowers will not permit, as at the end of any fiscal quarter, the cumulative Consolidated Adjusted Net Income for the period of two consecutive fiscal quarters then ended to be less than \$0.

§9.3. Consolidated Total Funded Debt to Consolidated EBITDA. As at the end of any fiscal quarter, the ratio of (a) Consolidated Total Funded Debt as of such date to (b) Consolidated EBITDA for the period of four (4) consecutive fiscal quarters ending on the date of calculation shall not exceed 4.50:1.00.

§9.4. Consolidated Senior Funded Debt to Consolidated EBITDA. As at the end of any fiscal quarter, the ratio of (a) Consolidated Senior Funded Debt as of such date to (b) Consolidated EBITDA for the period of four (4) consecutive fiscal quarters ending on the date of calculation shall not exceed 3.00:1.00.

§9.5. Consolidated Net Worth. The Borrowers will not permit Consolidated Net Worth at any time to be less than the sum of (a) \$237,526,000 minus (b) \$62,825,000 plus (c) on a cumulative basis, fifty percent (50%) of positive Consolidated Adjusted Net Income (after the payment of dividends with respect to any preferred stock of the Parent) for each fiscal quarter beginning with the fiscal quarter ended July 31, 2002, plus (d) one hundred percent (100%) of the proceeds of any sale by the Borrowers after the Effective Date of (i) equity securities issued by the Borrowers, or (ii) warrants or subscription rights for equity securities issued by the Borrowers.

§9.6. Capital Expenditures. The Borrowers will not permit, as at the end of any fiscal quarter, the amount of Capital Expenditures (excluding any Permitted Acquisitions) made by the Borrowers for the period of four (4) consecutive fiscal quarters then ended to exceed 1.5 <u>multiplied by</u> the sum of depreciation and landfill amortization expense for such four (4) fiscal quarter period (calculated in accordance with GAAP).



§10. CLOSING CONDITIONS. The "Effective Date" shall be such date on which all of the conditions precedent set forth in this §10 have been met. The obligations of the Existing Lenders to convert their claims against the Borrowers with respect to the Existing Credit Agreement into claims under this Credit Agreement, the obligations of the New Lenders to become parties to this Credit Agreement, and the obligations of all Lenders to make the initial Loans provided for in this Credit Agreement and otherwise be bound by the terms hereof, and of the Issuing Lender to issue any Letters of Credit hereunder, shall be subject to the satisfaction of each of the following conditions precedent:

<u>§10.1.</u> Corporate Action. All corporate action necessary for the valid execution, delivery and performance by each Borrower of the Loan Documents and the Senior Subordinated Debt Documents shall have been duly and effectively taken, and evidence thereof satisfactory to the Administrative Agent shall have been provided to the Administrative Agent.

<u>§10.2.</u> Loan Documents; Senior Subordinated Debt Documents. (a) Each of the Loan Documents shall have been duly and properly authorized, executed and delivered by the respective parties thereto and shall be in full force and effect in a form satisfactory to the Lenders and (b) each of the Senior Subordinated Debt Documents shall have been duly and properly authorized, executed and delivered by the respective parties thereto, and shall be in full force and effect in form and substance satisfactory to the Agents.

§10.3. Officer's Certificate: Certified Copies of Charter Documents. For each Borrower, the Administrative Agent shall have received a copy, certified by a duly authorized officer of such Person to be true and complete on the Effective Date, of each of (a) its charter or other incorporation or constituent documents (including certificates of merger or amalgamation and name changes) as in effect on such date of certification, and (b) its by-laws (or equivalent company documents) as in effect on such date.

<u>§10.4.</u> Incumbency Certificate. The Administrative Agent shall have received an incumbency certificate from each Borrower, dated as of the Effective Date, signed by duly authorized officers giving the name and bearing a specimen signature of each individual who shall be authorized: (a) to sign the Loan Documents and the Senior Subordinated Debt Documents on behalf of the Borrowers; (b) to make Loan and Letter of Credit Requests and Conversion Requests; and (c) to give notices and to take other action on the Borrowers' behalf under the Loan Documents.

<u>§10.5.</u> Validity of Liens. The Security Documents shall be effective to create in favor of the Administrative Agent (as collateral agent for the Lenders) a legal, valid and enforceable first priority, perfected (except in the case of Real Property and motor vehicles as set forth in §12) security interest in and lien upon the Collateral, subject only to Permitted Liens. All filings, recordings, deliveries of instruments and other actions necessary or desirable in the opinion of the Administrative Agent to protect and preserve such security interests, shall have been duly effected. The Administrative Agent shall have received evidence thereof in form and substance satisfactory to the Administrative Agent.

<u>§10.6.</u> Certificates of Insurance. The Administrative Agent shall have received (a) a certificate of insurance from an independent insurance broker, dated as of the Effective Date, or within fifteen (15) days prior thereto, identifying insurers, types of insurance, insurance limits, and policy terms, and otherwise describing the insurance obtained in accordance with the provisions of the Security Documents and (b) copies of all policies evidencing such insurance.

<u>§10.7.</u> Opinion of Counsel. The Lenders shall have received favorable legal opinions from counsel to the Borrowers addressed to the Administrative Agent, for the benefit of the Lenders, dated the Effective Date, in form and substance satisfactory to the Administrative Agent.

<u>§10.8.</u> Payment of Fees. The Borrowers shall have paid to the Administrative Agent for the accounts of the Lenders or its own account, as applicable, all fees and expenses that are due and payable as of the Effective Date and shall have paid the fees and disbursements of counsel to the Administrative Agent.

§10.9. Payoff. The Administrative Agent shall have received satisfactory evidence of the cancellation and payment in full of the Existing Credit Agreement.

§10.10. Financial Statements.

(a) The Lenders shall have received the financial projections of the Borrowers and its Subsidiaries, in form and substance satisfactory to the Agents and the Co-Arrangers, for the period from the Effective Date through April 30, 2008.

(b) The Agents shall have received a satisfactory day-one balance sheet and sources and uses of funds, showing the effects of the Senior Subordinated Debt and compliance with all terms and conditions of this Credit Agreement, including the covenants in §9 hereof.

§10.11. Financial Condition. The Administrative Agent shall have received a certificate of a duly authorized officer of the Parent, dated as of the Effective Date, and in form and detail satisfactory to the Agents and the Lenders, demonstrating that the ratio of (a) Consolidated Total Funded Debt on the Effective Date to (b) Consolidated EBITDA for the period of four (4) consecutive fiscal quarters most recently ended prior to the Effective Date, after giving effect, on a pro forma basis, to the transactions contemplated by this Credit Agreement, including the issuance of the Senior Subordinated Debt, does not exceed 3.75:1.00.

<u>§10.12.</u> Perfection Certificates and UCC Search Results. The Administrative Agent shall have received from each of the Borrowers a completed and fully-executed Perfection Certificate and the results of UCC searches (and the equivalent thereof in all applicable foreign jurisdictions) with respect to the Collateral, indicating no Liens other than Permitted Liens and otherwise in form and substance satisfactory to the Administrative Agent.

§10.13. Issuance of Senior Subordinated Debt. The Borrowers shall have issued Senior Subordinated Debt in principal amount at least equal to one hundred

fifty million Dollars (\$150,000,000) and shall have used the proceeds thereof to repay a portion of the obligations outstanding under the Existing Credit Agreement and the related transaction costs.

§11. CONDITIONS OF ALL LOANS. The obligations of the Lenders to make any Loan and of the Issuing Lender to issue, extend or renew any Letter of Credit, in each case on and subsequent to the Effective Date is subject to the following conditions precedent:

<u>§11.1. Representations True; No Event of Default.</u> Each of the representations and warranties of the Borrowers contained in this Credit Agreement or in any document or instrument delivered pursuant to or in connection with this Credit Agreement shall be true as of the date as of which they were made and shall also be true at and as of the time of the making of the Loans and the issuance, extension or renewal of Letters of Credit with the same effect as if made at and as of that time (except to the extent of changes resulting from transactions contemplated or permitted by this Credit Agreement and changes occurring in the ordinary course of business which singly or in the aggregate are not materially adverse, and to the extent that such representations and warranties relate expressly to an earlier date) and no Default or Event of Default shall have occurred and be continuing.

§11.2. Performance: No Event of Default. The Borrowers shall have performed and complied with all terms and conditions herein required to be performed or complied with by them prior to or at the time of the making of any Loan or issuance, extension or renewal of any Letter of Credit and at the time of the making of any Loan or issuance, extension or renewal of any Letter of Credit, there shall exist no Event of Default or condition which would result in an Event of Default upon consummation of such Loan or Letter of Credit issuance. Each request by the Borrowers for a Loan or Letter of Credit subsequent to the initial Loans and Letters of Credit shall constitute certification by the Borrowers that the conditions specified in §§11.1 and 11.2 will be duly satisfied on the date of such Loan or Letter of Credit issuance.

<u>§11.3.</u> No Legal Impediment. No change shall have occurred in any law or regulations thereunder or interpretations thereof which in the reasonable opinion of the Lenders would make it illegal for the Lenders to make Loans or the Issuing Lender to issue, extend or renew Letters of Credit hereunder.

§11.4. Proceedings and Documents. All proceedings in connection with the transactions contemplated by this Credit Agreement shall be satisfactory to the Administrative Agent and all documents incident thereto shall have been delivered to the Administrative Agent as of the Effective Date in substance and in form satisfactory to the Lenders, including without limitation, a Letter of Credit and Loan Request, in the form attached hereto as Exhibit B and the Lenders shall have received all information and such counterpart originals or certified or other copies of such documents as the Lenders may reasonably request.

§12. COLLATERAL SECURITY. The Obligations shall be secured by (a) a perfected (except in the case of Real Property and motor vehicles, subject to the following proviso) first-priority security interest (subject to Permitted Liens entitled to priority under applicable law) in all assets of each Borrower, whether now owned or hereafter acquired, pursuant to the terms of the Security Agreement to which each Borrower is a party; (b) a pledge of 100% of the capital

stock or other equity interests of such Borrowers (other than the Parent) to the Administrative Agent on behalf of the Lenders and the Agents pursuant to the Pledge Agreement; and (c) a pledge of 65% of the capital stock or other equity interests of each Foreign Subsidiary; <u>provided</u> that the Borrowers hereby agree, upon the request of the Administrative Agent and the Required Lenders, to deliver, as promptly as practicable, but in any event within sixty (60) days, titles to motor vehicles and mortgages with respect to Real Property and take such other steps as may be reasonably requested (including, without limitation, the delivery of legal opinions, Consulting Engineer's reports and title insurance) so as to provide the Administrative Agent, for the benefit of the Lenders and the Agents, a perfected first-priority security interest in such assets.

§13. EVENTS OF DEFAULT; ACCELERATION; TERMINATION OF COMMITMENT.

§13.1. Events of Default and Acceleration. If any of the following events ("Events of Default" or, if the giving of notice or the lapse of time or both is required, then, prior to such notice and/or lapse of time, "Defaults") shall occur:

(a) if the Borrowers shall fail to pay any principal of the Loans or any Reimbursement Obligation when the same shall become due and payable, whether at the Revolving Credit Maturity Date, the Term Loan Maturity Date or any accelerated date of maturity or at any other date fixed for payment;

(b) if the Borrowers shall fail to pay any interest or fees or other amounts owing hereunder within five (5) Business Days after the same shall become due and payable whether at the Revolving Credit Maturity Date, the Term Loan Maturity Date or any accelerated date of maturity or at any other date fixed for payment;

(c) if the Borrowers shall fail to comply with any of the covenants contained in \$7 (other than \$7.2, 7.3, 7.6, 7.7, 7.8, 7.9, 7.12, 7.14, 7.16, 7.17, and 7.18), <math>\$8 or \$9 hereof;

(d) if the Borrowers shall fail to perform any term, covenant or agreement contained herein or in any of the other Loan Documents (other than those specified in subsections (a), (b), and (c) above) within thirty (30) days after written notice of such failure has been given to the Borrowers by the Lenders;

(e) if any representation or warranty contained in this Credit Agreement or in any document or instrument delivered pursuant to or in connection with this Credit Agreement shall prove to have been false in any material respect upon the date when made or repeated;

(f) if any Borrower or Excluded Subsidiary shall fail to pay at maturity, or within any applicable period of grace, any and all obligations for borrowed money or any guaranty with respect thereto or credit received or in respect of any Capitalized Leases, in each case, in an aggregate amount greater than \$1,000,000 (including, without limitation, the Indebtedness evidenced by the Indenture), or fail to observe or perform any material term, covenant or agreement contained in any agreement by which it is bound, evidencing or securing borrowed money or credit received or in respect of any Capitalized Leases in an aggregate amount greater than \$1,000,000 (including, without limitation, the Indenture) for such period of time as would permit, assuming the giving of

appropriate notice if required, the holder or holders thereof or of any obligations issued thereunder to accelerate the maturity thereof;

(g) if any Borrower or Excluded Subsidiary makes an assignment for the benefit of creditors, or admits in writing its inability to pay or generally fails to pay its debts as they mature or become due, or petitions or applies for the appointment of a trustee or other custodian, liquidator, receiver or receiver/manager of any Borrower or Excluded Subsidiary or of any substantial part of the assets of any Borrower or Excluded Subsidiary or commences any case or other proceeding relating to any Borrower or Excluded Subsidiary under any bankruptcy, reorganization, arrangement, insolvency, readjustment of debt, dissolution or liquidation or similar law of any jurisdiction, now or hereafter in effect, or takes any action to authorize or in furtherance of any of the foregoing, or if any such petition or application is filed or any such case or other proceeding is commenced against any Borrower or Excluded Subsidiary and any Borrower or Excluded Subsidiary indicates its approval thereof, consent thereto or acquiescence therein;

(h) a decree or order is entered appointing any such trustee, custodian, liquidator, receiver or receiver/manager or adjudicating any Borrower or Excluded Subsidiary bankrupt or insolvent, or approving a petition in any such case or other proceeding, or a decree or order for relief is entered in respect of any Borrower or Excluded Subsidiary in an involuntary case under federal bankruptcy laws as now or hereafter constituted, and such decree or order remains in effect for more than sixty (60) days, whether or not consecutive;

(i) if there shall remain in force, undischarged, unsatisfied and unstayed, for more than forty-five (45) days, whether or not consecutive, any final judgment against any Borrower or Excluded Subsidiary which, with other outstanding final judgments, against the Borrowers and Excluded Subsidiaries exceeds in the aggregate \$2,500,000 after taking into account any undisputed insurance coverage;

(j) any Borrower or Excluded Subsidiary or any ERISA Affiliate incurs any liability to the PBGC or similar Canadian authorities or a Guaranteed Pension Plan (or any corresponding plan described in any Applicable Canadian Pension Legislation) pursuant to Title IV of ERISA in an aggregate amount exceeding \$1,000,000, or any Borrower or Excluded Subsidiary or any ERISA Affiliate is assessed withdrawal liability pursuant to Title IV of ERISA by a Multiemployer Plan requiring aggregate annual payments exceeding \$1,000,000, or any of the following occurs with respect to a Guaranteed Pension Plan (or any corresponding plan described in any Applicable Canadian Pension Legislation): (i) an ERISA Reportable Event or similar event under Applicable Canadian Pension Legislation, or a failure to make a required installment or other payment (within the meaning of §302(f)(1) of ERISA), provided that the Administrative Agent determines in its reasonable discretion that such event (A) could be expected to result in liability of any Borrower or Excluded Subsidiary to the PBGC, similar Canadian authorities or such Plan in an aggregate amount exceeding \$1,000,000 and (B) could constitute grounds for the termination of such Plan by the PBGC or similar Canadian authorities, for the appointment by the appropriate United States District Court or Canadian Court of a trustee to administer such Plan or for the imposition of a lien in favor of such Plan; or (ii) the appointment by a United States District Court or Canadian

Court of a trustee to administer such Plan; or (iii) the institution by the PBGC or similar Canadian authorities of proceedings to terminate such Plan;

(k) if any of the Loan Documents shall be cancelled, terminated, revoked or rescinded otherwise than in accordance with the terms thereof or with the express prior written agreement, consent or approval of the Lenders, or any action at law, suit or in equity or other legal proceeding to cancel, revoke or rescind any of the Loan Documents shall be commenced by or on behalf of the Borrowers or any of their respective stockholders, or any court or any other governmental or regulatory authority or agency of competent jurisdiction shall make a determination that, or issue a judgment, order, decree or ruling to the effect that, any one or more of the Loan Documents is illegal, invalid or unenforceable in accordance with the terms thereof;

(1) any Person or group of Persons (within the meaning of Section 13 or 14 of the Securities Exchange Act of 1934, as amended) shall have acquired beneficial ownership (within the meaning of Rule 13d-3 promulgated by the Securities and Exchange Commission under said Act) of 25% or more of the outstanding shares of common stock of the Parent (other than Berkshire Partners LLC); or, during any period of twelve consecutive calendar months, individuals who were directors of the Parent on the first day of such period shall cease to constitute a majority of the board of directors of the Parent;

- (m) a "<u>Change of Control</u>" as defined in the Series A Certificate shall occur; or
- (n) a "<u>Change of Control</u>" as defined in the Indenture shall occur;

then, and in any such event, so long as the same may be continuing, the Administrative Agent shall upon the request of the Required Lenders, by notice in writing to the Borrowers, declare all amounts owing with respect to this Credit Agreement, the Notes and the other Loan Documents and all Reimbursement Obligations to be, and they shall thereupon forthwith become, immediately due and payable without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Borrowers; <u>provided</u> that in the event of any Event of Default specified in §13.1(g) or 13.1(h), all such amounts shall become immediately due and payable automatically and without any requirement of notice from the Administrative Agent or any Lender. Upon demand by the Required Lenders after the occurrence of any Event of Default, the Borrowers shall immediately provide to the Administrative Agent cash in an amount equal to the aggregate Maximum Drawing Amount of all Letters of Credit outstanding, to be held by the Administrative Agent as collateral security for the Obligations.

§13.2. Termination of Commitments. If any one or more of the Events of Default specified in §13.1(g) or (h) shall occur, any unused portion of the Total Commitment hereunder shall forthwith terminate and the Lenders shall be relieved of all obligations to make Loans to, or issue Letters of Credit for the account of, any of the Borrowers. If any other Event of Default shall have occurred and be continuing, or if on any Drawdown Date the conditions precedent to the making of the Loans to be made on such Drawdown Date or the issuance of any Letters of Credit to be issued on such date are not satisfied (except as a consequence of a default on the part of the Lenders), the Administrative Agent may, and upon request of the Required Lenders, shall, by notice to

the Borrowers, terminate the unused portion of the Total Commitment hereunder, and upon such notice being given, such unused portion of the Total Commitment hereunder shall terminate immediately and the Lenders shall be relieved of all further obligations to make Loans to, or issue Letters of Credit for, the account of the Borrowers hereunder. No termination of any portion of the Total Commitment hereunder shall relieve the Borrowers of any of their existing Obligations to the Lenders hereunder or elsewhere.

§13.3 Remedies. Subject to §15.8, in case any one or more of the Events of Default shall have occurred and be continuing, and whether or not the Lenders shall have accelerated the maturity of the Loans pursuant to §13.1, each Lender with the consent of the Required Lenders, if owed any amount with respect to the Loans or the Reimbursement Obligations, may proceed to protect and enforce its rights by suit in equity, action at law or other appropriate proceeding, whether for the specific performance of any covenant or agreement contained in this Credit Agreement and the other Loan Documents or any instrument pursuant to which the Obligations to such Lender are evidenced, including, without limitation, as permitted by applicable law, the obtaining of the <u>ex parte</u> appointment of a receiver, and, if such amount shall have become due, by declaration or otherwise, proceed to enforce the payment thereof or any legal or equitable right of such Lender. No remedy herein conferred upon any Lender or the Agents or the holder of any Note or purchaser of any Letter of Credit Participation is intended to be exclusive of any other remedy and each and every remedy shall be cumulative and shall be in addition to every other remedy given hereunder or now or hereafter existing at law or in equity or by statute or any other provision of law.

<u>§13.4.</u> Distribution of Collateral Proceeds. In the event that, following the occurrence or during the continuance of any Default or Event of Default, the Administrative Agent or any Lender, as the case may be, receives any monies in connection with the enforcement of any the Security Documents, or otherwise with respect to the realization upon any of the Collateral, such monies shall be distributed for application as follows:

(a) <u>First</u>, to the payment of, or (as the case may be) the reimbursement of the Administrative Agent for or in respect of all reasonable costs, expenses, disbursements and losses which shall have been incurred or sustained by the Administrative Agent in connection with the collection of such monies by the Administrative Agent, for the exercise, protection or enforcement by the Administrative Agent of all or any of the rights, remedies, powers and privileges of the Administrative Agent under this Agreement or any of the other Loan Documents or in respect of the Collateral or in support of any provision of adequate indemnity to the Administrative Agent against any taxes or liens which by law shall have, or may have, priority over the rights of the Administrative Agent to such monies;

(b) <u>Second</u>, to all other Obligations pari passu among the Administrative Agent and the Lenders; <u>provided</u>, <u>however</u>, that (i) distributions shall be made with respect to each type of Obligation owing to the Lenders, such as interest, principal, fees and expenses, among the Lenders on a <u>pro rata</u> basis, and (ii) the Administrative Agent may in its discretion make proper allowance to take into account any Obligations not then due and payable;

(c) Third, upon payment and satisfaction in full or other provisions for payment in full satisfactory to the Lenders and the Administrative Agent of all of the Obligations, to the payment of any obligations required to be paid pursuant to \$9-608(a)(1)(C) or 9-615(a)(3) of the Uniform Commercial Code of the Commonwealth of Massachusetts; and

(d) Fourth, the excess, if any, shall be returned to the Borrowers or to such other Persons as are entitled thereto.

§14. SETOFF. The Borrowers hereby grant to the Agents, the Issuing Lender and each of the Lenders a continuing lien, security interest and right of setoff as security for all liabilities and obligations to the Agents, the Issuing Lender and each Lender, whether now existing or hereafter arising, upon and against all deposits, credits, collateral and property, now or hereafter in the possession, custody, safekeeping or control of such Agent, the Issuing Lender or such Lender or any of their Affiliates and their respective successors and assigns or in transit to any of them. Regardless of the adequacy of any collateral, if any of the Obligations are due and payable and have not been paid or during the continuance of an Event of Default, any deposits or other sums credited by or due from any Lender or the Administrative Agent to the Borrowers and any securities or other property of the Borrowers in the possession of such Lender or the Administrative Agent may be applied to or set off against the payment of the Obligations and any and all other liabilities, direct, or indirect, absolute or contingent, due or to become due, now existing or hereafter arising, of the Borrowers to the Lenders. ANY AND ALL RIGHTS TO REQUIRE ANY LENDER TO EXERCISE ITS RIGHTS OR REMEDIES WITH RESPECT TO ANY OTHER COLLATERAL WHICH SECURES THE OBLIGATIONS, PRIOR TO EXERCISING ITS RIGHT OF SETOFF WITH RESPECT TO SUCH DEPOSITS, CREDITS OR OTHER PROPERTY OF THE BORROWERS ARE HEREBY KNOWINGLY, VOLUNTARILY AND IRREVOCABLY WAIVED. Each of the Lenders agrees with each other Lender that if such Lender shall receive from the Borrowers any amount in respect of the Obligations, whether by voluntary payment, exercise of the right of setoff, counterclaim, cross action, enforcement of the claim evidenced by the Loan Documents, by proceedings against the Borrowers at law or in equity or by proof thereof in bankruptcy, reorganization, liquidation, receivership or similar proceedings, or otherwise, and shall retain and apply to the payment of the Obligations owed to such Lender any amount in excess of its ratable portion of the payments received by all of the Lenders with respect to the Obligations owed to all of the Lenders, such Lender will make such disposition and arrangements with the other Lenders with respect to such excess, either by way of distribution, pro tanto assignment of claims, subrogation or otherwise as shall result in each Lender receiving in respect of the Obligations such Lender's proportionate payment as contemplated by this Credit Agreement; provided that if all or any part of such excess payment is thereafter recovered from such Lender, such disposition and arrangements shall be rescinded and the amount restored to the extent of such recovery, but without interest.

§15. THE AGENTS.

§15.1. Appointment, Powers and Immunities.

(a) Each Lender hereby irrevocably appoints and authorizes Fleet to act as the Administrative Agent and BOA to act as Syndication Agent hereunder and under the other Loan Documents. Each Lender irrevocably authorizes the Administrative Agent to execute the Security Documents and all other instruments relating thereto and to take

from time to time any action with respect to any Collateral or the Security Documents which may be necessary to perfect, maintain perfected or insure the priority of the security interest in and liens upon the Collateral granted pursuant to the Security Documents, and authorizes the Agents to take such other action on behalf of each of the Lenders and to exercise all such powers as are delegated to the Agents hereunder and under any of the other Loan Documents and all related documents, together with such other powers as are reasonably incidental thereto, <u>provided</u> that no duties or responsibilities not expressly assumed herein or therein shall be implied to have been assumed by the Agents.

(b) The relationship between the Agents and each of the Lenders is that of an independent contractor. The use of the terms "Administrative Agent" and "Syndication Agent" is for convenience only and is used to describe, as a form of convention, the independent contractual relationship between the Agents and each of the Lenders. Nothing contained in this Credit Agreement nor the other Loan Documents shall be construed to create an agency, trust or other fiduciary relationship between the Agents and any of the Lenders. As an independent contractor empowered by the Lenders to exercise certain rights and perform certain duties and responsibilities hereunder and under the other Loan Documents, the Agents are nevertheless "representatives" of the Lenders, as that term is defined in Article 1 of the Uniform Commercial Code, for purposes of actions for the benefit of the Lenders and the Agents with respect to all collateral security and guaranties contemplated by the Loan Documents. Such actions include the designation of the Administrative Agent as "secured party", "mortgagee" or the like on all financing statements and other documents and instruments, whether recorded or otherwise, relating to the attachment, perfection, priority or enforcement of any security interests, mortgages or deeds of trust in collateral security intended to secure the payment or performance of any of the Obligations, all for the benefit of the Lenders.

(c) The Administrative Agent and the Syndication Agent may exercise their powers and execute their duties by or through employees or agents and attorneys-in-fact and shall not be responsible for the negligence or misconduct of any such employees, agents or attorneys-in-fact selected by it with reasonable care. The Agents shall be entitled to take, and to rely on, advice of counsel concerning all matters pertaining to their rights and duties under this Credit Agreement and the other Loan Documents. Each Agent may utilize the services of such Persons as it in its sole discretion may reasonably determine, and all reasonable fees and expenses of any such Persons shall be paid by the Borrowers.

(d) Neither of the Agents nor any of their respective shareholders, directors, officers or employees nor any other Person assisting them in their duties nor any agent or employee thereof, shall be liable for any waiver, consent or approval given or any action taken, or omitted to be taken, in good faith by it or them hereunder or under any of the other Loan Documents, or in connection herewith or therewith, or be responsible for the consequences of any oversight or error of judgment whatsoever, except that the Agents or such other Person, as the case may be, may be liable for losses due to their willful misconduct or gross negligence.

(e) Each Agent in its separate capacity as a Lender shall have the same rights and powers hereunder as any other Lender. It is agreed that the duties, rights,

privileges and immunities of the Issuing Lender, in its capacity as issuer of Letters of Credit hereunder, shall be identical to its duties, rights, privileges and immunities as a Lender as provided in this §15.

§15.2. Actions By Agents. The Agents shall be fully justified in failing or refusing to take any action under this Credit Agreement as they reasonably deem appropriate unless they shall first have received such advice or concurrence of the Lenders and shall be indemnified to its reasonable satisfaction by the Lenders against any and all liability and expense which may be incurred by them by reason of taking or continuing to take any such action. The Agents shall in all cases be fully protected in acting, or in refraining from acting, under this Credit Agreement or any of the Loan Documents in accordance with a request of the Lenders, and such request and any action taken or failure to act pursuant thereto shall be binding upon the Lenders and all future holders of the Notes or any Letter of Credit Participation.

§15.3. Indemnification of Agents. Without limiting the obligations of the Borrowers hereunder or under any other Loan Document, the Lenders agree to indemnify the Agents and their respective affiliates, agents, directors, officers and shareholders, and ratably in accordance with their respective Loan Percentages for any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements or any kind or nature whatsoever which may at any time be imposed on, incurred by or asserted against the Agents in any way relating to or arising out of this Credit Agreement or any other Loan Document or any documents contemplated by or referred to herein or therein or the transactions contemplated hereby or thereby or the enforcement of any of the terms hereof or thereof or of any such other documents; provided, that no Lender shall be liable for any of the foregoing to the extent they arise from the gross negligence or willful misconduct of such Agent (or any agent thereof).

§15.4. Reimbursement for Advances Made by Administrative Agent. Without limiting the provisions of §15.3, the Lenders and the Administrative Agent hereby agree that the Administrative Agent shall not be obliged to make available to any Person any sum which the Administrative Agent is expecting to receive for the account of that Person until the Administrative Agent has determined that it has received that sum. The Administrative Agent may, however, disburse funds prior to determining that the sums which the Administrative Agent expects to receive have been finally and unconditionally paid to the Administrative Agent, if the Administrative Agent wishes to do so. If and to the extent that the Administrative Agent does disburse funds and it later becomes apparent that the Administrative Agent did not then receive a payment in an amount equal to the sum paid out, then any Person to whom the Administrative Agent made the funds available shall, on demand from the Administrative Agent, received by it in such capacity hereunder or under the Loan Documents might involve it in liability, it may refrain from making distribution until its right to make distribution shall have been adjudicated by a court of competent jurisdiction. If a court of competent jurisdiction shall adjudge that any amount received and distributed by the Administrative Agent is to be repaid, each Person to whom any such distribution shall have been made shall either repay to the Administrative Agent its proportionate share of the amount so adjudged to be repaid or

shall pay over the same in such manner and to such Persons as shall be determined by such court.

§15.5. Closing Documentation, etc. For purposes of determining compliance with the conditions set forth in §10, each Lender that has executed this Credit Agreement shall be deemed to have consented to, approved or accepted, or to be satisfied with, each document and matter either sent, or made available, by the Agents or the Co-Arrangers to such Lender for consent, approval, acceptance or satisfaction, or required thereunder to be to be consent to or approved by or acceptable or satisfactory to such Lender, unless an officer of such Agent or such Co-Arranger active upon the Borrowers' account shall have received notice from such Lender prior to the Effective Date specifying such Lender's objection thereto and such objection shall not have been withdrawn by notice to such Agent or such Co-Arranger to such effect on or prior to the Effective Date. The Administrative Agent will forward to each Lender, promptly after the Administrative Agent's receipt thereof, a copy of each notice or other document furnished to the Administrative Agent for such Lender hereunder; provided, however, that, notwithstanding the foregoing, the Administrative Agent may furnish to the Lenders a monthly summary with respect to Letters of Credit issued hereunder in lieu of copies of the related Letter of Credit Applications.

§15.6. Non-Reliance on Agents and Other Lenders. Each Lender represents that it has, independently and without reliance on either Agent or any other Lender, and based on such documents and information as it has deemed appropriate, made its own appraisal of the financial condition and affairs of the Borrowers and decision to enter into this Credit Agreement and the other Loan Documents and agrees that it will, independently and without reliance upon either Agent or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own appraisals and decisions in taking or not taking action under this Credit Agreement or any other Loan Document. The Agents shall not be responsible for the due execution or validity or enforceability of this Credit Agreement, the Notes, the Letters of Credit, any of the other Loan Documents or any instrument at any time constituting, or intended to constitute, collateral security for the Notes, or for the value of any such collateral security or for the validity, enforceability or collectability of any such amounts owing with respect to the Notes, or for any recitals or statements, warranties or representations made herein or in any of the other Loan Documents or in any certificate or instrument hereafter furnished to it by or on behalf of the Borrowers, or be bound to ascertain or inquire as to the performance or observance by the Borrowers of any of the terms, conditions, covenants or agreements in this Credit Agreement, the other Loan Documents or any other document referred to or provided for herein or therein or to make inquiry of, or to inspect the properties or books of, any Person. The Agents shall not be bound to ascertain whether any notice, consent, waiver or request delivered to it by the Borrowers or any holder of any of the Notes shall have been duly authorized or is true, accurate and complete. Except for notices, reports and other documents and information expressly required to be furnished to the Lenders by the Agents hereunder, the Agents shall not have any duty or responsibility to provide any Lender with any credit or other information concerning any Person which may come into the possession of the Agents or any of their affiliates. The Agents have not made nor do they now make any representations or warranties, express or implied, nor do they assume any liability to the Lenders, with respect to the credit worthiness or financial conditions of the Borrowers. Each Lender shall have access to all documents relating to each Agent's performance of

its duties hereunder at such Lender's request. Unless any Lender shall promptly object to any action taken by an Agent hereunder (other than actions to which the provisions of §15.8 are applicable and other than actions which constitute gross negligence or willful misconduct by such Agent), such Lender shall conclusively be presumed to have approved the same.

§15.7. Resignation. The Administrative Agent may resign at any time by giving sixty (60) days' prior written notice thereof to the Lenders and the Borrowers. Upon any such resignation, the Required Lenders shall have the right to appoint a successor Administrative Agent. If no successor Administrative Agent shall have been so appointed by the Required Lenders and shall have accepted such appointment within thirty (30) days after the retiring Administrative Agent's giving of notice of resignation, then the retiring Administrative Agent may, on behalf of the Lenders, appoint a successor Lender Agent, which shall be a financial institution which shall be a financial institution having a rating of not less than "A" or its equivalent by S&P. Upon the acceptance of any appointment as Administrative Agent hereunder by a successor Administrative Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Administrative Agent, and the retiring Administrative Agent shall be discharged from its duties and obligations hereunder. After any retiring Administrative Agent's resignation, the provisions of this Credit Agreement shall continue in effect for its benefit in respect of any actions taken or omitted to be taken by it while it was acting as Administrative Agent. Any new Administrative Agent appointed pursuant to this §15.7 shall immediately issue new Letters of Credit in place of Credit previously issued by the prior Administrative Agent.

§15.8. Action by the Lenders, Consents, Amendments, Waivers, Etc. Any consent or approval required or permitted by this Credit Agreement to be given by the Lenders may be given, and any term of this Credit Agreement, the other Loan Documents or any other instrument related hereto or mentioned herein may be amended, and the performance or observance by the Borrower or any of its Subsidiaries of any terms of this Credit Agreement, the other Loan Documents or such other instrument or the continuance of any Default or Event of Default may be waived (either generally or in a particular instance and either retroactively or prospectively) with, but only with, the written consent of the Borrowers and the written consent of the Required Lenders. Notwithstanding the foregoing, no amendment, waiver or consent shall:

- (a) without the written consent of the Borrowers and each Lender directly affected thereby:
- (i) reduce or forgive the principal amount of any Loans or Reimbursement Obligations, or reduce the rate of interest on the Notes or the amount of the Commitment Fee, Financial L/C Fee or Performance L/C Fee;
- (ii) increase the amount of such Lender's Commitment or extend the expiration date of such Lender's Commitment;
- (iii) postpone or extend the Revolving Credit Maturity Date or the Term Loan Maturity Date or any other regularly scheduled dates for payments of principal of, or interest on, the Loans or Reimbursement Obligations or

any fees or other amounts payable to such Lender (it being understood that any vote to rescind any acceleration made pursuant to \$13.1 of amounts owing with respect to the Loans and other Obligations shall require only the approval of the Required Lenders);

- (iv) other than pursuant to a transaction permitted by the terms of this Credit Agreement, release (A) all or substantially all of the Collateral (excluding, if any Borrower becomes a debtor under the federal Bankruptcy Code, the release of "cash collateral", as defined in Section 363(a) of the federal Bankruptcy Code pursuant to a cash collateral stipulation with the debtor approved by the Required Lenders) or (B) any Borrower from its Obligations; and
- (v) amend or modify the provisions of §4.4 (Mandatory Prepayments of Term Loan), §13.4 (Distribution of Collateral Proceeds) or §27(a) (Pari Passu Treatment);
- (b) without the written consent of all of the Lenders, amend or waive this §15.8 or the definition of Required Lenders;

(c) without the written consent of the Administrative Agent, amend or waive §2.8 or §15, the amount or time of payment of the Administrative Agent's fees payable for the Administrative Agent's account or any other provision applicable to the Administrative Agent; or

(d) without the written consent of the Issuing Lender, amend or waive, the amount or time of payment of any Financial L/C Fees or Performance L/C Fees or other fees payable for the Issuing Lender's account or any other provision applicable to the Issuing Lender.

No waiver shall extend to or affect any obligation not expressly waived or impair any right consequent thereon. No course of dealing or delay or omission on the part of the Administrative Agent or any Lender in exercising any right shall operate as a waiver thereof or otherwise be prejudicial thereto. No notice to or demand upon the Borrowers shall entitle the Borrowers to other or further notice or demand in similar or other circumstances.

§16. EXPENSES. Whether or not the transactions contemplated herein shall be consummated, the Borrowers hereby promise to pay (a) the reasonable costs of producing and reproducing this Credit Agreement, the other Loan Documents and the other agreements and instruments mentioned herein, (b) subject to §5.2(d), any taxes (including any interest and penalties in respect thereto) payable by either Agent, the Issuing Lender or any of the Lenders (other than taxes based upon such Agent's, the Issuing Lender's or any Lender's net income) on or with respect to the transactions contemplated by this Credit Agreement (the Borrowers hereby agreeing to indemnify the Agents, the Issuing Lender and the Lenders with respect thereto), (c) all reasonable fees, expenses and disbursements of counsel for the Agents or any local counsel to the Administrative Agent incurred or expended in connection with the preparation, syndication, negotiation, administration or interpretation of this Credit Agreement, the other Loan Documents, each closing hereunder, any amendment, modification, approval, consent or waiver hereto or hereunder, or the cancellation of any Loan Document upon payment in full in cash of all of the

Obligations or pursuant to any terms of such Loan Document providing for such cancellation, (d) all reasonable out-of-pocket costs, fees and expenses of the Agents or any of their affiliates incurred in connection with the preparation, syndication, administration or interpretation of the Loan Documents and other instruments mentioned herein (including, without limitation, collateral evaluation costs and Consulting Engineer's fees), (e) all reasonable out-of-pocket expenses (including, without limitation, reasonable attorneys' fees and costs, which attorneys may be employees of any Lender or the Administrative Agent, and reasonable consulting, accounting, appraisal, investment banker and similar professional fees and charges) incurred by any Lender or the Administrative Agent in connection with (i) the enforcement of, or preservation of rights under, any of the Loan Documents against the Borrowers or the administration thereof after the occurrence of a Default or Event of Default and (ii) any litigation, proceeding or dispute, whether arising hereunder or otherwise, in any way related to the credit hereunder and (f) all reasonable fees, expenses and disbursements of any Lender or the Administrative Agent incurred in connection with UCC searches, UCC filings, intellectual property searches or intellectual property filings. The covenants contained in this §16 shall survive payment or satisfaction in full of all other obligations.

§17. INDEMNIFICATION. The Borrowers jointly and severally agree to indemnify and hold harmless the Agents, the Issuing Lender and the Lenders, as well as their respective shareholders, directors, agents, officers, subsidiaries, affiliates, trustees and advisors, from and against all damages, losses, settlement payments, obligations, liabilities, claims, actions, suits (whether groundless or otherwise), penalties, assessments, citations, directives, demands, judgments, actions or causes of action, whether statutory created or under the common law, and reasonable costs and expenses incurred, suffered, sustained or required to be paid by an indemnified party by reason of or resulting from this Credit Agreement, the other Loan Documents or the transactions contemplated hereby or thereby including, without limitation, (a) any actual or proposed use by the Borrower or any of its Subsidiaries of the proceeds of any of the Loans or Letters of Credit, (b) the Borrowers entering into or performing this Credit Agreement or any of the other Loan Documents or (c) with respect to the Borrowers and their respective properties and assets, the violation of any Environmental Law, the presence, disposal, escape, seepage, leakage, spillage, discharge, emission, Release or threatened Release of any Hazardous Substances or any action, suit, proceeding or investigation brought or threatened with respect to any Hazardous Substances (including, but not limited to, claims with respect to wrongful death, personal injury or damage to property), in each case including, without limitation, the reasonable fees and disbursements of counsel and allocated costs of internal counsel incurred in connection with any such investigation, litigation or other proceeding, except in each case, as any of the foregoing result from the gross negligence or willful misconduct of the indemnified party. In any investigation, proceeding or litigation, or the preparation therefor, each Lender shall be entitled to select its own counsel and, in addition to the foregoing indemnity, the Borrowers agree to pay promptly the reasonable fees and expenses of such counsel. In the event of the commencement of any such proceeding or litigation, the Borrowers shall be entitled to participate in such proceeding or litigation with counsel of their choice at their expense, provided that such counsel shall be reasonably satisfactory to the Lenders. The covenants of this §17 shall survive payment or satisfaction in full of the Obligations.

§18. SURVIVAL OF COVENANTS, ETC. Unless otherwise stated herein, all covenants, agreements, representations and warranties made herein, in the other Loan Documents or in any documents or other papers delivered by or on behalf of the Borrowers pursuant hereto shall be deemed to have been relied upon by the Lenders, the Issuing Lender and the Agents, notwithstanding any investigation heretofore or hereafter made by any of them, and shall survive

the making by the Lenders of the Loans and the issuance, extension or renewal of any Letters of Credit, as herein contemplated, and shall continue in full force and effect so long as any amount due under this Credit Agreement, any Letter of Credit or the Notes remains outstanding and unpaid or any Lender has any obligation to make any Loans or issue any Letters of Credit hereunder. All statements contained in any certificate or other paper delivered by or on behalf of the Borrowers pursuant hereto or in connection with the transactions contemplated hereby shall constitute representations and warranties by the Borrowers hereunder.

§19. ASSIGNMENTS AND PARTICIPATION.

(a) Assignments. It is understood and agreed that each Lender shall have the right to assign at any time all or any portion of its Commitment and interests in the risk relating to any Revolving Credit Loans and outstanding Letters of Credit and/or its Term Loan Percentage of the Term Loan to any Person, provided that: (i) each such assignment shall be in a minimum amount of \$1,000,000 (or, if less, in a minimum amount equal to all of such Lender's Commitment and interests in the risk relating to any Revolving Credit Loans and outstanding Letters of Credit and/or its Term Loan Percentage of the Term Loan); (ii) the Administrative Agent and, so long as no Event of Default has occurred and is continuing, the Parent, shall have consented to such assignment, which such consent of the Parent shall not be unreasonably withheld; provided that the consent of the Administrative Agent and the Parent shall not apply, if the assignment is to a Lender or a Lender or a Lender Affiliate so long as such assignment would not result in increased costs to the Borrowers hereunder; and (iii) the proposed assignee and the assigning Lender execute and deliver to the Administrative Agent and the Borrowers hereunder; and Acceptance in the form attached hereto as Exhibit G (in each case, an "Assignment and Acceptance"). Upon the execution and delivery of such Assignment and Acceptance, (A) the Borrowers shall issue to the assignee applicable Notes in the amount of such assignee's Commitment and/or portion of the Term Loan, dated the effective date of such Assignment and Acceptance and otherwise completed in substantially the form of the Notes executed and delivered to the Lenders on the Effective Date and, if applicable, the assignor shall return to the Borrowers its existing Notes marked "cancelled"; and (B) the assignee shall pay a processing and recordation fee of \$3,500 to the Administrative Agent. Only one such assignment fee shall be payable for concurrent assignments to Lender Affiliates of an assigning Lender.

(b) <u>Participations</u>. Each Lender shall also have the right to grant participations to one or more banks, other financial institutions or other entities whose business is to purchase and sell loan assets in the normal course (each a "<u>Participant</u>") in or to all or any part of any Loans owing to such Lender and the Note held by such Lender; <u>provided</u> that (i) any such sale or participation shall not affect the rights and duties of the selling Lender hereunder to the Borrowers and (ii) the only rights granted to the Participant pursuant to such participation arrangements with respect to waivers, amendments or modifications of the Loan Documents shall be the rights to approve waivers, amendments or modifications that would require consent by all of the Lenders under §15.8, and (iii) any Participant shall be entitled to the benefits of §5.4, §5.5, §5.8, §5.12 and §17 as if it were a Lender hereunder, <u>provided</u>, however, that no Borrower shall be required to pay any amount which is greater than such amount that otherwise would have been payable to the Lender which sold such participation.

(c) <u>Miscellaneous</u>. Notwithstanding the foregoing, no assignment, participation or accession shall operate to (i) except in accordance with §19(g), increase the Total Commitment or amount of the Term Loan hereunder unless consented to by the Required Lenders or (ii) reduce

the Commitment or portion of the Term Loan of any Lender to an amount less than \$1,000,000 (or, if less, in a minimum amount equal to all of such Lender's Commitment and interests in the risk relating to any Revolving Credit Loans and outstanding Letters of Credit or its Term Loan Percentage of the Term Loan), or (iii) otherwise alter the substantive terms of this Credit Agreement. Anything contained in this \$19 to the contrary notwithstanding, any Lender may at any time grant a security interest in all or any portion of its rights under this Credit Agreement and the other Loan Documents to secure obligations of such Lender, including without limitation (a) any pledge or assignment to secure obligations to any of the twelve Federal Reserve Banks organized under \$4 of the Federal Reserve Act, 12 U.S.C. \$341 and (b) with respect to any Lender that is a Fund, to any lender or any trustee for, or any other representative of, holders of obligations owed or securities issued by such Fund as security for such obligations hereunder, provide any voting rights hereunder to the secured party thereof, substitute any such secured party for such Lender as a party hereto or affect any rights or obligations of the Borrowers or Agents hereunder.

(d) <u>Register</u>. On the date specified in any Assignment and Acceptance or Instrument of Accession and upon the satisfaction of the other conditions set forth in this §19, such bank or financial institution shall become a party to this Credit Agreement and the other Loan Documents for all purposes of this Credit Agreement and the other Loan Documents, and its Commitment and/or portion of the Term Loan shall be as set forth in the register of Lenders (the "<u>Register</u>") maintained by the Administrative Agent for the recordation of the names and addresses of the Lenders and the Commitment Percentage of, Term Loan Percentage of, and principal amount of the Loans owing to and Letter of Credit Participations purchased by, the Lenders from time to time. The entries in the Register shall be conclusive, in the absence of manifest error, and the Borrowers, the Administrative Agent and the Lenders may treat each person whose name is recorded in the Register as a Lender hereunder for all purposes of this Credit Agreement. The Register shall be available for inspection by the Borrowers and the Lenders at any reasonable time and from time to time upon reasonable prior notice.

(e) Assignee or Participant Affiliated with a Borrower. If any assignee Lender is an Affiliate of any Borrower, then any such assignee Lender shall have no right to vote as a Lender hereunder or under any of the other Loan Documents for purposes of granting consents or waivers or for purposes of agreeing to amendments or other modifications to any of the Loan Documents or for purposes of making requests to the Administrative Agent pursuant to §13.1 or §13.2, and the determination of the Required Lenders shall for all purposes of this Credit Agreement and the other Loan Documents be made without regard to such assignee Lender's interest in any of the Loans or Reimbursement Obligations. If any Lender sells a participating interest in any of the Loans or Reimbursement Obligations to a Participant, and such Participant is a Borrower or an Affiliate of a Borrower, then such transferor Lender shall promptly notify the Administrative Agent of the sale of such participant is or waivers or for purposes of agreeing to amendments or modifications to any of the Loan Documents or for purposes of agreeing to amendments or modifications to any of the Loans or Reimbursement Obligations to a Participant, and such Participant is a Borrower or an Affiliate of a Borrower, then such transferor Lender shall promptly notify the Administrative Agent of the sale of such participantion. A transferor Lender shall have no right to vote as a Lender hereunder or under any of the Loan Documents or for purposes of granting consents or waivers or for purposes of agreeing to amendments or modifications to any of the Loan Documents or for purposes of agreeing to amendments or modifications to any of the Loan Documents or for purposes of making requests to the Administrative Agent pursuant to §13.1 or §13.2 to the extent that such participation is beneficially owned by a Borrower or any Affiliate of a Borrower, and the determination of the Required Lenders shall for all purposes of this Credit Agreement and the other Loan Documents be made withou

disclosed to the other Lenders that it is an Affiliate of a Borrower and which, following such disclosure, has been excepted from the provisions of this §19(e) in a writing signed by the Required Lenders determined without regard to the interest of such assignee Lender or transferor Lender, to the extent of such participation, in Loans or Reimbursement Obligations.

Special Purpose Funding Vehicle. Notwithstanding anything to the contrary contained in this §19, any Lender (a "Granting Lender") may grant to a special purpose funding vehicle (an "SPV") of such Granting Lender, identified as such in writing from time to time delivered by the Granting Lender to the Administrative Agent and the Borrowers, the option to provide to the Borrowers all or any part of any Loan that such Granting Lender would otherwise be obligated to make to the Borrowers pursuant to this Credit Agreement, provided that (a) nothing herein shall constitute a commitment to make any Loan by any SPV, (b) the Granting Lender's obligations under this Credit Agreement shall remain unchanged, (c) the Granting Lender shall retain the sole right to enforce this Credit Agreement and to approve any amendment, modification or waiver of any provision of this Credit Agreement and (d) if an SPV elects not to exercise such option or otherwise fails to provide all or any part of such Loan, the Granting Lender shall be obligated to make such Loan pursuant to the terms hereof. The making of a Loan by an SPV hereunder shall utilize the Commitment of the Granting Lender to the same extent, and as if, such Loan were made by the Granting Lender. Each party hereto hereby agrees that no SPV shall be liable for any expense reimbursement, indemnity or similar payment obligation under this Credit Agreement (all liability for which shall remain with the Granting Lender). In furtherance of the foregoing, each party hereto hereby agrees (which agreement shall survive the termination of this Credit Agreement) that, prior to the date that is one year and one day after the later of (i) the payment in full of all outstanding senior indebtedness of any SPV and (ii) the Revolving Credit Maturity Date, or, as applicable, the Term Loan Maturity Date, it will not institute against, or join any other person in instituting against, such SPV any bankruptcy, reorganization, arrangement, insolvency or liquidation proceedings or similar proceedings under the laws of the United States of America or any State thereof. In addition, notwithstanding anything to the contrary contained in this §19, any SPV may (A) with notice to, but (except as specified below) without the prior written consent of, the Borrowers or the Administrative Agent and without paying any processing fee therefor, assign all or a portion of its interests in any Loans to its Granting Lender or to any financial institutions (consented to by the Administrative Agent and, so long as no Default or Event of Default has occurred and is continuing, the Borrowers, which consents shall not be unreasonably withheld or delayed) providing liquidity and/or credit facilities to or for the account of such SPV to fund the Loans made by such SPV or to support the securities (if any) issued by such SPV to fund such Loans and (B) disclose on a confidential basis any non-public information relating to its Loans (other than financial statements referred to in §6.4 or §7.4) to any rating agency, commercial paper dealer or provider of a surety, guarantee or credit or liquidity enhancement to such SPV. In no event shall the Borrowers be obligated to pay to an SPV that has made a Loan any greater amount than the Borrowers would have been obligated to pay under this Agreement if the Granting Lender had made such Loan. An amendment to this \$19(f) without the written consent of an SPV shall be ineffective insofar as it alters the rights and obligations of such SPV.

(g) <u>Acceding Lenders</u>. One or more commercial banks, other financial institutions or other Persons (in each case, an "<u>Acceding Lender</u>") may, at the request of the Borrowers, become party to this Credit Agreement as a Lender by entering into an Instrument of Accession in substantially the form of <u>Exhibit H</u> hereto (an "<u>Instrument of Accession</u>") with the Borrowers and the Administrative Agent and assuming thereunder the rights and obligations of a Lender hereunder, including, without limitation, Commitments to make Revolving Credit Loans and

participate in the risk relating to Letters of Credit and (as the case may be) the obligation to fund a portion of the Term Loan in amounts to be agreed upon by the Borrowers and the Acceding Lender subject to the terms hereof, and the Total Commitment and (as the case may be) the Term Loan shall thereupon be increased (each such increase referred to as a "Post-Closing Increase") by the amount of such Acceding Lender's interest; provided that:

(i) no Default or Event of Default has occurred or is continuing at the time of such accession;

(ii) the Lenders party to this Credit Agreement shall have the first option, and may elect, to fund their *pro rata* share of any Post-Closing Increase, but no Lender shall have any obligation to do so;

(iii) in the event that an Acceding Lender was not a Lender party to this Credit Agreement prior to giving effect to the Instrument of Accession, such Acceding Lender shall be acceptable to the Administrative Agent;

(iv) in no event shall the sum of (a) the Term Loan <u>plus</u> (b) the Total Commitment (after giving effect to all Instruments of Accession) exceed in the aggregate \$375,000,000 <u>minus</u> any previously effected reductions of the Total Commitment and the Term Loan pursuant to \$2.2, 4.4 or 4.5, respectively; and

(v) the Borrowers shall indemnify the Lenders and the Administrative Agent for any cost or expense incurred as a consequence of the reallocation of any Eurodollar Rate Loans to an Acceding Lender pursuant to the provisions of §5.12.

On the effective date specified in any Instrument of Accession, <u>Schedule 2</u> hereto shall be deemed to be amended to reflect (x) the name, address, Commitment, Commitment Percentage and Term Loan Percentage of the Acceding Lender, (y) the Total Commitment and the Term Loan after giving effect to the Post-Closing Increase, and (z) the changes to the respective Commitments, Commitment Percentages and Term Loan Percentages of the other Lenders, as applicable, resulting from such Post-Closing Increase.

§20. PARTIES IN INTEREST. All the terms of this Credit Agreement and the other Loan Documents shall be binding upon and inure to the benefit of and be enforceable by the parties hereto and their respective successors and permitted assigns of the parties hereto and thereto; provided, that no Borrower shall assign or transfer its rights or obligations hereunder without the prior written consent of each Lender. Nothing in this Credit Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby) any legal or equitable right, remedy or claim under or by reason of this Credit Agreement or any of the other Loan Documents.

§21. NOTICES, ETC.

Except as otherwise expressly provided in this Credit Agreement, all notices and other communications made or required to be given pursuant to this Credit Agreement or the other Loan Documents shall be in writing and shall be delivered in hand, mailed by United States registered or certified first-class mail, postage prepaid, sent by overnight courier, or sent by telegraph, telex or telecopier and confirmed by delivery via courier or postal service, addressed as follows:

(a) if to the Borrowers, c/o Casella Waste Systems, Inc. at 25 Greens Hill Lane, P.O. Box 866, Rutland, Vermont 05701, Attention: Chairman and Chief Financial Officer, telecopy number 802-775-6198, or at such other address for notice as the Borrowers shall last have furnished in writing to the Person giving the notice;

(b) if to the Administrative Agent or Fleet, at 100 Federal Street, Boston, Massachusetts 02110, USA, Attention: Timothy M. Laurion, Managing Director, telecopy number 617-434-2160, or at such other address for notice as the Administrative Agent or Fleet shall last have furnished in writing to the Person giving the notice; or

(c) if to any Lender, at such Lender's address set forth on <u>Schedule 2</u> hereto, or such other address for notice as such Lender shall have last furnished in writing to the Person giving the notice.

Any such notice or demand shall be deemed to have been duly given or made and to have become effective (a) if delivered by hand, overnight courier or facsimile to a responsible officer of the party to which it is directed, at the time of the receipt thereof by such officer, (b) if sent by registered or certified first-class mail, postage prepaid, five (5) Business Days after the posting thereof, and (c) if sent by telex or cable, at the time of the dispatch thereof, if in normal business hours in the country of receipt, or otherwise at the opening of business on the following Business Day.

§22. MISCELLANEOUS. The rights and remedies herein expressed are cumulative and not exclusive of any other rights which the Lenders, the Issuing Lender or the Agents would otherwise have. The captions in this Credit Agreement are for convenience of reference only and shall not define or limit the provisions hereof. This Credit Agreement and any amendment hereof may be executed in several counterparts and by each party on a separate counterpart, each of which when so executed and delivered shall be an original, but all of which together shall constitute one instrument. In proving this Credit Agreement it shall not be necessary to produce or account for more than one such counterpart signed by the party against whom enforcement is sought. Delivery by facsimile by any of the parties hereto of an executed counterpart hereof or of any amendment or waiver hereto shall be as effective as an original executed counterpart hereof or of such amendment or waiver and shall be considered a representation that an original executed counterpart hereof or such amendment or waiver.

§23. ENTIRE AGREEMENT, ETC. The Loan Documents and any other documents executed in connection herewith or therewith express the entire understanding of the parties with respect to the transactions contemplated hereby. Neither this Credit Agreement nor any term hereof may be changed, waived, discharged or terminated, except as provided in §15.8. No waiver shall extend to or affect any obligation not expressly waived or impair any right consequent thereon. No course of dealing or omission on the part of the Administrative Agent or any Lender in exercising any right shall operate as a waiver thereof or otherwise be prejudicial thereto. No notice to or demand upon the Borrowers shall entitle the Borrowers to other or further notice or demand in similar or other circumstances.

§24. WAIVER OF JURY TRIAL. EACH OF THE PARTIES HERETO HEREBY WAIVES ITS RIGHT TO A JURY TRIAL WITH RESPECT TO ANY ACTION OR CLAIM ARISING OUT OF ANY DISPUTE IN CONNECTION WITH THIS CREDIT AGREEMENT, THE NOTES OR ANY OF THE OTHER LOAN DOCUMENTS, ANY RIGHTS OR OBLIGATIONS HEREUNDER OR THEREUNDER OR THE PERFORMANCE OF SUCH

RIGHTS AND OBLIGATIONS OR ANY COURSE OF CONDUCT, COURSE OF DEALINGS, STATEMENTS (WHETHER VERBAL OR WRITTEN) OR ACTIONS OF ANY PARTY, INCLUDING ANY COURSE OF CONDUCT, COURSE OF DEALINGS, STATEMENTS OR ACTIONS OF THE ADMINISTRATIVE AGENT OR ANY LENDER RELATING TO THE ADMINISTRATION OF THE LOANS OR ENFORCEMENT OF THE LOAN DOCUMENTS AND AGREES THAT IT WILL NOT SEEK TO CONSOLIDATE ANY SUCH ACTION WITH ANY OTHER ACTION IN WHICH A JURY TRIAL CANNOT BE OR HAS NOT BEEN WAIVED. Except as prohibited by law, each of the parties hereto hereby waives any right it may have to claim or recover in any litigation referred to in the preceding sentence any special, exemplary, punitive or consequential damages or any damages other than, or in addition to, actual damages. Each of the Borrowers (a) certifies that no representative, agent or attorney of any Lender, the Issuing Lender or the Agents would not, in the event of litigation, seek to enforce the foregoing waivers and (b) acknowledges that the Agents, the Issuing Lender and the Lenders have been induced to enter into this Credit Agreement, the other Loan Documents to which it is a party by, among other things, the waivers and certifications contained herein.

§25. GOVERNING LAW. THIS CREDIT AGREEMENT AND, EXCEPT AS OTHERWISE SPECIFICALLY PROVIDED THEREIN, EACH OF THE OTHER LOAN DOCUMENTS ARE CONTRACTS UNDER THE LAWS OF THE COMMONWEALTH OF MASSACHUSETTS AND SHALL BE CONSTRUED IN ACCORDANCE WITH AND GOVERNED BY THE LAWS OF THE COMMONWEALTH OF MASSACHUSETTS (WITHOUT REFERENCE TO CONFLICTS OR CHOICE OF LAW). THE BORROWERS CONSENT AND AGREE THAT ANY SUIT FOR THE ENFORCEMENT OF THIS CREDIT AGREEMENT OR ANY OF THE OTHER LOAN DOCUMENTS MAY BE BROUGHT IN THE COURTS OF THE COMMONWEALTH OF MASSACHUSETTS (WITHOUT REFERENCE TO CONFLICTS OR CHOICE OF LAW). THE BORROWERS MAY BE BROUGHT IN THE COURTS OF THE COMMONWEALTH OF MASSACHUSETTS OR ANY FEDERAL COURT SITTING THEREIN AND CONSENTS TO THE NONEXCLUSIVE JURISDICTION OF SUCH COURT AND SERVICE OF PROCESS IN ANY SUCH SUIT BEING MADE UPON THE BORROWERS IN ACCORDANCE WITH LAW AT THE ADDRESS SPECIFIED IN §21. THE BORROWERS HEREBY WAIVE ANY OBJECTION THAT THEY MAY NOW OR HEREAFTER HAVE TO THE VENUE OF ANY SUCH SUIT OR ANY SUCH COURT OR THAT SUCH SUIT IS BROUGHT IN AN INCONVENIENT COURT.

§26. **SEVERABILITY.** The provisions of this Credit Agreement are severable and if any one clause or provision hereof shall be held invalid or unenforceable in whole or in part in any jurisdiction, then such invalidity or unenforceability shall affect only such clause or provision, or part thereof, in such jurisdiction, and shall not in any manner affect such clause or provision in any other jurisdiction, or any other clause or provision of this Credit Agreement in any jurisdiction.

§27. PARI PASSU TREATMENT.

(a) Notwithstanding anything to the contrary set forth herein (other than as set forth in §13.4), each payment or prepayment of principal and interest received after the occurrence of an Event of Default hereunder shall be distributed pari passu among the Lenders, in accordance with the aggregate outstanding amount of the Obligations owing to each Lender divided by the aggregate outstanding amount of all Obligations.

(b) Following the occurrence and during the continuance of any Event of Default, each Lender agrees that if it shall, through the exercise of a right of banker's lien, setoff or counterclaim against any Borrower (pursuant to §13.3 or otherwise), including a secured claim under §506 of the Bankruptcy Code or other security or interest arising from or in lieu of, such secured claim, received by such Lender under any applicable bankruptcy, insolvency or other similar law or otherwise, obtain payment (voluntary or involuntary) in respect of the Notes, Loans, and other Obligations held by it as a result of which the unpaid principal portion of the Notes, Loans and the Obligations held by it shall be proportionately less than the unpaid principal portion of the Notes, Loans and the Obligations held by it shall be deemed to have simultaneously purchased from such other Lender a participation in the Notes, Loans and Obligations held by each Lender shall be in the same proportion to the aggregate unpaid principal amount of the Notes, Loans and Obligations then outstanding as the principal amount of the Notes, Loans and other Obligations held by it prior to such exercise of banker's lien, setoff or counterclaim was to the principal amount of all Notes, Loans and other Obligations outstanding prior to such exercise of banker's lien, setoff or counterclaim was to the principal amount of all Notes, Loans and obligations outstanding prior to such exercise of banker's lien, setoff or counterclaim; provided, however, that if any such purchase or purchases or adjustments shall be made pursuant to this §27 and the payment giving rise thereto shall thereafter be recovered, such purchase or purchases or adjustments shall be rescinded to the extent of such recovery and the purchase price or prices or adjustments shall be rescinded to the extent of such recovery and the purchase price or prices or adjustments restored without interest.

(c) Each Borrower expressly consents to the foregoing arrangements and agrees that any Person holding such a participation in the Notes, Loans and the Obligations deemed to have been so purchased may exercise any and all rights of banker's lien, setoff or counterclaim with respect to any and all moneys owing by such Borrower to such Person as fully as if such Person had made a Loan directly to such Borrower in the amount of such participation.

(d) Nothing contained in this §27 shall impair, as between the Borrowers and any Lender, the obligation of the applicable Borrowers to pay such Lender all amounts payable in respect of such Lender's Notes, Loans, and other Obligations as and when the same shall become due and payable in accordance with the terms thereof.

§28. EXISTING CREDIT AGREEMENT SUPERSEDED. This Credit Agreement shall supersede the Existing Credit Agreement in its entirety, except as provided in this §28. On the Effective Date, the rights and obligations of the parties under the Existing Credit Agreement and the "Notes" as defined therein shall be subsumed within and be governed by this Credit Agreement and the Notes; provided, however, that each of the "Revolving Credit Loans", "Swingline Loans", "Letters of Credit" and the "Term Loan" (as each such term is defined in the Existing Credit Agreement) outstanding under the Existing Credit Agreement on the Effective Date shall, for purposes of this Credit Agreement, be included as Revolving Credit Loans, Swingline Loans, Letters of Credit and the Term Loan hereunder in accordance with the provisions hereof.

§29. TREATMENT OF CERTAIN CONFIDENTIAL INFORMATION.

§29.1. <u>Confidentiality.</u> Each of the Lenders, the Issuing Lender and the Agents agrees, on behalf of itself and each of its affiliates, directors, officers, employees and representatives, to use reasonable precautions to keep confidential, in accordance with their customary procedures for handling confidential information of the same nature and in accordance with safe and sound banking practices, any non-public information

supplied to it by any of the Borrowers pursuant to this Credit Agreement that is identified by such Borrower as being confidential at the time the same is delivered to such Lender, the Issuing Lender or such Administrative Agent, provided that nothing herein shall limit the disclosure of any such information (a) after such information shall have become public other than through a violation of this §29, or becomes available to any of the Lenders, the Issuing Lender or the Agents on a nonconfidential basis from a source other than such Borrower, (b) to the extent required by statute, rule, regulation or judicial process, (c) to counsel for any of the Lenders, the Issuing Lender or the Agents, (d) to bank examiners or any other regulatory authority having jurisdiction over any Lender, the Issuing Lender or either Agent, or to auditors or accountants, (e) to either Agent, the Issuing Lender, any Lender or any Financial Affiliate, (f) in connection with any litigation to which any one or more of the Lenders, the Issuing Lender or either Agents or any Financial Affiliate is a party, or in connection with the enforcement of rights or remedies hereunder or under any other Loan Document, (g) to an affiliate of any Lender or the Issuing Lender or a Subsidiary or affiliate of either Agent, (h) to any actual or prospective assignee or participant or any actual or prospective counterparty (or its advisors) to any swap or derivative transactions referenced to credit or other risks or events arising under this Credit Agreement or any other Loan Document so long as such assignee, participant or counterparty, as the case may be, agrees to be bound by the provisions of this §29.1 or (i) with the consent of the Borrowers. Moreover, each of the Agents, the Issuing Lender, the Lenders and any Financial Affiliate is hereby expressly permitted by the Borrowers to refer to any of the Borrowers and the Excluded Subsidiaries in connection with any advertising, promotion or marketing undertaken by such Agent, the Issuing Lender, such Lender or such Financial Affiliate and, for such purpose, the such Agent, the Issuing Lender, such Lender or such Financial Affiliate may utilize any trade name, trademark, logo or other distinctive symbol associated with any of the Borrowers or any of their Subsidiaries or any of their businesses.

§29.2. Prior Notification. Unless specifically prohibited by applicable law or court order, each of the Lenders, the Issuing Lender and the Agents shall, prior to disclosure thereof, notify the Borrowers of any request for disclosure of any such non-public information by any governmental agency or representative thereof (other than any such request in connection with an examination of the financial condition of such Lender, the Issuing Lender or such Agent by such governmental agency) or pursuant to legal process.

§29.3. Other. In no event shall any Lender, the Issuing Lender or either Agent be obligated or required to return any materials furnished to it or any Financial Affiliate by the Borrowers. The obligations of each Lender under this §29 shall supersede and replace the obligations of such Lender under any confidentiality letter in respect of this financing signed and delivered by such Lender to the Borrowers prior to the date hereof and shall be binding upon any assignee of, or purchaser of any participation in, any interest in any of the Loans or Reimbursement Obligations from any Lender.

IN WITNESS WHEREOF, the undersigned have duly executed this Second Amended and Restated Revolving Credit and Term Loan Agreement as a sealed instrument as of the date first set forth above.

BORROWERS: CASELLA WASTE SYSTEMS, INC. ALL CYCLE WASTE, INC. ALTERNATE ENERGY, INC. ATLANTIC COAST FIBERS, INC. B. AND C. SANITATION CORPORATION BLASDELL DEVELOPMENT GROUP, INC. BRISTOL WASTE MANAGEMENT, INC. CASELLA TRANSPORTATION, INC. CASELLA WASTE MANAGEMENT OF MASSACHUSETTS, INC. CASELLA WASTE MANAGEMENT OF N.Y., INC. CASELLA WASTE MANAGEMENT OF PENNSYLVANIA, INC. CASELLA WASTE MANAGEMENT, INC. DATA DESTRUCTION SERVICES, INC. FAIRFIELD COUNTY RECYCLING, INC. FCR CAMDEN, INC. FCR FLORIDA, INC. FCR GREENSBORO, INC. FCR GREENVILLE, INC. FCR MORRIS, INC. FCR PLASTICS, INC. FCR REDEMPTION, INC. FCR TENNESSEE, INC. FCR VIRGINIA, INC. FCR, INC. FOREST ACQUISITIONS, INC. GRASSLANDS INC. HAKES C & D DISPOSAL, INC. HIRAM HOLLOW REGENERATION CORP. K-C INTERNATIONAL, LTD. KTI BIO FUELS, INC. KTI ENERGY OF VIRGINIA, INC. KTI ENVIRONMENTAL GROUP, INC. KTI NEW JERSEY FIBERS, INC. KTI OPERATIONS INC. KTI RECYCLING OF NEW ENGLAND, INC. KTI RECYCLING OF NEW JERSEY, INC. KTI SPECIALTY WASTE SERVICES, INC. KTI, INC.

[SIGNATURES CONTINUED ON FOLLOWING PAGE]

MECKLENBURG COUNTY RECYCLING, INC. NATURAL ENVIRONMENTAL, INC. NEW ENGLAND WASTE SERVICES OF MASSACHUSETTS, INC. NEW ENGLAND WASTE SERVICES OF ME, INC. NEW ENGLAND WASTE SERVICES OF N.Y., INC. NEW ENGLAND WASTE SERVICES OF VERMONT, INC. NEW ENGLAND WASTE SERVICES, INC. NEWBURY WASTE MANAGEMENT, INC. NORTH COUNTRY ENVIRONMENTAL SERVICES, INC. NORTHERN PROPERTIES CORPORATION OF PLATTSBURGH NORTHERN SANITATION, INC. PERC, INC. PINE TREE WASTE, INC. R.A. BRONSON INC. RESOURCE RECOVERY OF CAPE COD, INC. RESOURCE RECOVERY SYSTEMS OF SARASOTA, INC. RESOURCE RECOVERY SYSTEMS, INC. RESOURCE TRANSFER SERVICES, INC. RESOURCE WASTE SYSTEMS, INC. SCHULTZ LANDFILL, INC. SUNDERLAND WASTE MANAGEMENT, INC. U.S. FIBER, INC. WASTE-STREAM INC. WESTFIELD DISPOSAL SERVICE, INC. WINTERS BROTHERS, INC.

By: <u>/s/ Richard A. Norris</u> Name: Richard A. Norris Title: Vice President and Treasurer

CASELLA NH INVESTORS CO., LLC

By: KTI, Inc., its sole member

By: <u>/s/ Richard A. Norris</u> Name: Richard A. Norris Title: Vice President and Treasurer

CASELLA NH POWER CO., LLC

By: KTI, Inc., its sole member

By: <u>/s/ Richard A. Norris</u> Name: Richard A. Norris Title: Vice President and Treasurer

CASELLA RTG INVESTORS CO., LLC

By: Casella Waste Systems, Inc., its sole member

By: /s/ Richard A. Norris

Name: Richard A. Norris Title: Chief Financial Officer and Treasurer

THE HYLAND FACILITY ASSOCIATES

By:	/s/	Richard A. Norris
Nai	me:	Richard A. Norris
Tit	le:	Duly Authorized Agent

MAINE ENERGY RECOVERY COMPANY, LIMITED PARTNERSHIP

By: KTI Environmental Group, Inc., general partner

By: <u>/s/ Richard A. Norris</u> Name: Richard A. Norris Title: Vice President and Treasurer

NEW ENGLAND LANDFILL SOLUTIONS, LLC

By: Rochester Environmental Park, LLC

By: <u>/s/ Richard A. Norris</u> Name: Richard A. Norris Title: Vice President and Treasurer

PERC MANAGEMENT COMPANY LIMITED PARTNERSHIP

By: PERC Inc., general partner

By: <u>/s/ Richard A. Norris</u> Name: Richard A. Norris Title: Vice President and Treasurer

ROCHESTER ENVIRONMENTAL PARK, LLC

By:

r: <u>/s/ Richard A. Norris</u> Name: Richard A. Norris Title: Vice President and Treasurer

FLEET NATIONAL BANK, individually and as Administrative Agent

By: /s/ David C. Brecht Name: Title:

David C. Brecht Vice President

BANK OF AMERICA, N.A., individually and as Syndication Agent

By: /s/ Steven R. Arentsen Name: Title:

Steven R. Arentsen Senior Vice President

COMERICA BANK

By: <u>/s/ D. Scott Hagwell</u> Name:

Name: Title: D. Scott Hagwell CBO

LASALLE BANK NATIONAL ASSOCIATION

By: /s/ Brian Peterson Name: Title:

Brian Peterson First Vice President

BANK NORTH, N.A.

By: /s/ E. Kirke Hart Name: E. Kirke Hart Title: Regional vice President

MERRILL LYNCH CAPITAL, a division of Merrill Lynch Business Financial Services, Inc.

By: /s/ Sheila C. Weimer Name: Title:

Sheila C. Weimer Vice President

CITIZENS BANK

By: <u>/s/ Daniel Bernard</u> Name:

Name: Title: Daniel Bernard Vice President

AMENDMENT NO. 2 TO SECOND AMENDED AND RESTATED REVOLVING CREDIT AND TERM LOAN AGREEMENT

This AMENDMENT NO. 2 TO SECOND AMENDED AND RESTATED REVOLVING CREDIT AND TERM LOAN AGREEMENT (this "<u>Amendment</u>") is made and entered into as of the 26th day of August, 2003, by and among CASELLA WASTE SYSTEMS, INC., a Delaware corporation (the "<u>Parent</u>"), its Subsidiaries (other than Excluded Subsidiaries) listed on <u>Schedule 1</u> to the Credit Agreement defined below (together with the Parent, collectively the "<u>Borrowers</u>"), such banks or other financial institutions which may become a party thereto (the "<u>Lenders</u>"), FLEET NATIONAL BANK as Administrative Agent for the Lenders (the "<u>Administrative Agent</u>"), and BANK OF AMERICA, N.A. as Syndication Agent.

WHEREAS, the Borrowers, the Lenders and the Agents are parties to a Second Amended and Restated Revolving Credit and Term Loan Agreement dated as of January 24, 2003, (as the same may be amended and in effect from time to time, the "<u>Credit Agreement</u>"), pursuant to which the Lenders have extended credit to the Borrowers on the terms set forth therein;

WHEREAS, pursuant to the Credit Agreement, the original Term Loan Lenders have made a term loan to the Borrowers in the original principal amount of \$150,000,000 (the "Original Term Loan");

WHEREAS, the Borrowers wish to repay the Original Term Loan in full and to amend the Credit Agreement to allow for a replacement term loan in the aggregate principal amount of \$150,000,000 and having a lower interest rate than the Original Term Loan (the "<u>Replacement Term Loan</u>"), the proceeds of which shall be used to prepay the Original Term Loan;

WHEREAS, each of the Borrowers has requested that the Required Lenders and the Administrative Agent make certain amendments to the Credit Agreement to permit the aforementioned transactions, and the Required Lenders and the Administrative Agent are agreeable thereto upon the terms and conditions described herein;

NOW, THEREFORE, in consideration of the foregoing, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereby agree as follows:

§1. <u>Replacement Term Loan</u>.

(a) Subject to the terms and conditions set forth herein, each of the undersigned financial institutions identified on <u>Schedule A</u> hereto as a "Replacement Lender" (individually, a "<u>Replacement Lender</u>" and collectively, the "<u>Replacement Lenders</u>") agrees to lend to the Borrowers on the Second Amendment Effective Date (defined below) that percentage set forth opposite its name on <u>Schedule A</u> hereto of a term loan, which shall be in the aggregate principal amount of \$150,000,000. The proceeds of the Replacement Term Loan shall be applied to prepay in full the outstanding principal balance of the Original Term Loan.

- (b) Upon payment of the Original Term Loan (the "August 2003 Prepayment"), the Replacement Term Loan shall be thereafter the "Term Loan" referred to and as defined in the Credit Agreement, shall constitute an Obligation of the Borrowers, shall amortize on the schedule and be subject to the mandatory and voluntary prepayment provisions as set forth in §4 of the Credit Agreement for the Term Loan, and the outstanding principal amount thereof shall be due and payable on the Term Loan Maturity Date. The Replacement Term Loan shall bear interest from and after the Second Amendment Effective Date as set forth in §4.6 of the Credit Agreement, with the Applicable Margin being determined by reference to the pricing grid set forth in §2(a) of this Amendment. The Borrowers jointly and severally promise to pay to the Administrative Agent for the account of the Replacement Lenders the principal of and interest on the Replacement Term Loan in accordance with the provisions of §4 of the Credit Agreement (as amended by this Amendment).
- (c) The Replacement Lenders, the Revolving Credit Lenders, the Borrowers and the Agents hereby acknowledge that immediately following the August 2003 Prepayment (i) the Credit Agreement shall be amended in accordance with the provisions of §§2 through 7 below, (ii) all references in the Credit Agreement and the other Loan Documents to the words "Term Loan" shall be deemed to be references to or to relate to the Replacement Term Loan advanced pursuant to this §1, (iii) the Replacement Term Loan shall be governed by and subject to all of the provisions, terms and conditions set forth in the Credit Agreement and the other Loan Documents in every respect as though such Replacement Term Loan had been originally referred to therein as the "Term Loan", and (iv) thereafter the Replacement Term Loan shall be referred to as the Term Loan. In furtherance of and not in limitation of the foregoing, and notwithstanding anything to the contrary contained herein, the amendments effected by the provisions of §§ 2 through 7 below shall only be applicable to the Replacement Term Loan and shall not be applicable to the Original Term Loan.
- (d) Each Replacement Lender not a party to the Credit Agreement prior to the Second Amendment Effective Date (each a "<u>New Term Lender</u>"), by execution of this Amendment, hereby agrees to be bound by and shall be entitled to the benefits of, all of the terms, conditions and provisions of the Credit Agreement as if such New Term Lender had been one of the lending institutions originally executing the Credit Agreement as a "Term Loan Lender"; <u>provided</u> that nothing herein shall be construed as making the New Term Lender liable to the Borrowers or the other Lenders in respect of any acts or omissions of any party to the Credit Agreement or in respect of any other event occurring prior to the Second Amendment Effective Date. The New Term Lender (a) represents and warrants that (i) it is duly and legally authorized to enter into this Amendment, (ii) the execution, delivery and performance of this Amendment does not conflict with any provision of law or of the charter or bylaws of the New Term Lender, or of any agreement binding on the New Term Lender, (iii) all acts, conditions and things required to be done and performed and to have occurred prior to the

execution, delivery and performance of this Amendment, and to render the same the legal, valid and binding obligation of the New Term Lender, enforceable against it in accordance with its terms, have been done and performed and have occurred in due and strict compliance with all applicable laws; (b) confirms that it has received a copy of the Credit Agreement and the Loan Documents, together with copies of the most recent financial statements delivered pursuant to §§6.4 and 7.4 thereof and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Amendment; (c) agrees that it will, independently and without reliance upon the Lenders or the Administrative Agent and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Credit Agreement; (d) represents and warrants that it is eligible to become a party to the Credit Agreement under the terms and conditions of the Credit Agreement as amended hereby; (e) appoints and authorizes the Administrative Agent to take such action as Administrative Agent on its behalf and to exercise such powers under the Credit Agreement and the other Loan Documents as are delegated to the Administrative Agent by the terms thereof, together with such powers as are reasonably incidental thereto; and (f) agrees that it will perform in accordance with their terms all of the obligations which by the terms of the Credit Agreement are required to be performed by it as a Term Loan Lender.

- (e) The Borrowers shall, within five (5) days after a request by a Replacement Lender to the Administrative Agent, deliver to each such requesting Replacement Lender a Term Note evidencing such Replacement Lender's Replacement Term Loan, as renamed pursuant to §1(c) hereof.
- §2. <u>Amendment to §1.1 of the Credit Agreement</u>. Section 1.1 of the Credit Agreement is hereby amended as follows:

(a) The definition of "<u>Applicable Rate</u>" set forth in such §1.1 is amended by deleting the table set forth in such definition and replacing it with the table set forth below.

			Revolving Credit Loans:		Term Loans:		
			Applicable	Applicable	Applicable	Applicable	
		Pricing	Rate for Base Rate	Rate for Eurodollar	Rate for Base Rate	Rate for Eurodollar	
I	evel	Ratio	Loans	Rate Loans	Loans	Rate Loans	Commitment Fee
-							
	Ι	Less than 2.75:1.0	Base Rate <u>plus</u>	Eurodollar Rate <u>plus</u>	Base Rate <u>plus</u>	Eurodollar Rate <u>plus</u>	0.375%
			0.25% per annum	2.25% per annum	0.75% per annum	2.75% per annum	

Π	Greater than or equal to 2.75:1.0 and less than 3.25:1.0	Base Rate <u>plus</u> 0.50% per annum	Eurodollar Rate <u>plus</u> 2.50% per annum	Base Rate <u>plus</u> 0.75% per annum	Eurodollar Rate <u>plus</u> 2.75% per annum	0.500%
III	Greater than or equal to 3.25:1.0 and less than 3.75:1.0	Base Rate <u>plus</u> 0.75% per annum	Eurodollar Rate <u>plus</u> 2.75% per annum	Base Rate <u>plus</u> 0.75% per annum	Eurodollar Rate <u>plus</u> 2.75% per annum	0.500%
IV	Greater than or equal to 3.75:1.0 and less than 4.25:1.0	Base Rate <u>plus</u> 1.00% per annum	Eurodollar Rate <u>plus</u> 3.00% per annum	Base Rate <u>plus</u> 1.00% per annum	Eurodollar Rate <u>plus</u> 3.00% per annum	0.500%
V	Greater than or equal to 4.25:1.0	Base Rate <u>plus</u> 1.25% per annum	Eurodollar Rate <u>plus</u> 3.25% per annum	Base Rate <u>plus</u> 1.00% per annum	Eurodollar Rate <u>plus</u> 3.00% per annum	0.500%

(b) The definitions of "Term Loan" and "Term Loan Lenders" set forth in such §1.1 are deleted in their entirety and substituted in lieu thereof are the following new definitions:

"<u>Term Loan</u>. (a) Prior to the Second Amendment Effective Date, the term loan made by the Term Loan Lenders to the Borrowers on the Effective Date in the original principal amount of \$150,000,000 pursuant to \$4, and (b) as of the Second Amendment Effective Date, the term loan made by the Term Loan Lenders (which Term Loan Lenders were originally referred to in such Second Amendment as the Replacement Lenders) to the Borrowers on the Second Amendment Effective Date in the original principal amount of \$150,000,000 pursuant to \$10,000,000 pursuant to \$10,

"<u>Term Loan Lenders</u>. (a) Prior to the Second Amendment Effective Date, the lenders holding a portion of the Term Loan made to the Borrowers on the Effective Date as set forth on <u>Schedule 2</u> hereto together with any other Person who became an assignee of any rights and obligations of a Term Loan Lender pursuant to §19 of the Credit Agreement prior to such date, and (b) from and after the Second

Amendment Effective Date, the lenders (which Term Loan Lenders were originally referred to in such Second Amendment as the Replacement Lenders) who hold a portion of the Term Loan made to the Borrowers on the Second Amendment Effective Date (the proceeds of which were used to prepay the Term Loan advanced on the Effective Date) as set forth on <u>Schedule 2</u> hereto together with any other Person who becomes an assignee of any rights and obligations of a Term Loan Lender pursuant to §19 of the Credit Agreement."

(c) The following new definitions are added to such §1.1 in the appropriate alphabetical order:

"Second Amendment. That certain Second Amendment to Second Amended and Restated Credit and Term Loan Agreement, dated as of August , 2003, among the Borrowers, the Administrative Agent and the Lenders."

"Second Amendment Effective Date. The date on which each of the conditions precedent set forth in §8 of the Second Amendment have been satisfied."

§3. <u>Amendment to §4.2 of the Credit Agreement</u>. Section 4.2 of the Credit Agreement is hereby amended by deleting the text of such §4.2 in its entirety and substituting in lieu thereof the following text:

§4.2 Evidence of Debt; Term Notes.

§4.2.1 Loan Accounts. Each Term Loan Lender shall maintain in accordance with its usual practice an account or accounts evidencing indebtedness of the Borrowers to such Term Loan Lender resulting from such Term Loan Lender's Term Loan Percentage of the Term Loan Lender from time to time, including the amounts of principal and interest payable and paid to such Term Loan Lender from time to time under this Credit Agreement. The Administrative Agent shall maintain accounts in which it shall record (a) the amount of the Term Loan made hereunder, the Type thereof and each Interest Period applicable thereto, (b) the amount of any principal or interest due and payable or to become due and payable from the Borrowers to each Term Loan Lender hereunder and (c) both the amount of any sum received by the Administrative Agent hereunder for the account of the Term Loan Lender's share thereof (if any). The entries made by each Term Loan Lender in the accounts maintained pursuant to this §4.2.1 (or any Term Note Record referred to below) shall, to the extent permitted by applicable law, be <u>prima facie</u> evidence of the existence and amounts of the obligations of the Borrowers to repay (with applicable interest) the Term Loan made in accordance with the terms of this Credit Agreement.

§4.2.2. <u>Term Notes</u>. The Borrowers agree that, upon the request of any Term Loan Lender, the amount of the Term Loan advanced or to be advanced by such Term Loan Lender shall be evidenced by a promissory

note of the Borrowers in substantially the form of Exhibit A-3 hereto (each a "Term Note"), dated the Second Amendment Effective Date (or such other date on which a Term Loan Lender may become a party hereto in accordance with §19 of the Credit Agreement) and completed with appropriate insertions. The Term Note shall be payable to the order of such Term Loan Lender in a principal amount equal to the Term Loan made by such Term Loan Lender on the Second Amendment Effective Date (or, in the case of Term Notes issued after the Second Amendment Effective Date, be in a stated principal amount equal to the outstanding principal amount of the Term Loan of such Term Loan Lender on the date of the issuance thereof). The Borrowers irrevocably authorize each Term Loan Lender with a Term Note to make or cause to be made a notation on such Term Loan and, at or about the time of the receipt of any payment of principal on such Lender's Term Note, an appropriate notation on such Term Loan Lender's Term Note Record reflecting the receipt of such payment.

§4. <u>Amendment to §4.5 of the Credit Agreement</u>. Section 4.5 of the Credit Agreement is hereby amended by inserting the following sentence and table at the end of such section:

If the Borrowers prepay the Term Loan in whole or in part, the Borrowers shall pay a premium with respect to each such prepayment in an amount determined in accordance with the percentages set forth in the following table opposite the period during which such prepayment is made:

Period	Prepayment Premium	
Second Amendment Effective Date through first anniversary thereof	1% of amount prepaid	
Day after first anniversary of Second Amendment Effective Date through the second anniversary of the Second Amendment Effective Date	0.5% of amount prepaid	
Thereafter	-0-	

§5. <u>Amendment to §6.15 of the Credit Agreement</u>. Section 6.15 of the Credit Agreement is hereby amended by deleting the text of such §6.15 in its entirety and substituting in lieu thereof the following table:

6.15. <u>Use of Proceeds</u>. The proceeds of the Loans shall be used (a) to refinance the existing Indebtedness of the Borrowers under the Existing Credit Agreement, (b) to fund Permitted Acquisitions, (c) for Capital Expenditures and (d)

for working capital and other general corporate purposes, <u>provided</u> that the proceeds of the Term Loan (advanced on the Second Amendment Effective Date) shall be used solely to repay the existing indebtedness of the Borrowers with respect to the Term Loan (advanced on the Effective Date). No proceeds of the Loans are to be used, and no portion of any Letter of Credit is to be obtained, in any way that will violate Regulations U or X of the Board of Governors of the Federal Reserve System. The Borrowers will obtain Letters of Credit solely for general corporate purposes.

§6. <u>Amendment to §19(a) of the Credit Agreement</u>. Section 19(a) of the Credit Agreement is hereby amended by inserting, after the first two words "the Borrowers" in clause (A) of such section, the following parenthetical: "(if requested by the assignee in the case of an assignment of a portion of the Term Loan)".

§7. <u>Schedule 2</u>. Upon the effectiveness of this Amendment, the Administrative Agent shall revise <u>Schedule 2</u> of the Credit Agreement to reflect each Replacement Lender's Term Loan Percentage, dollar amount of its portion of the Term Loan and lending office as of the date of effectiveness.

§8. <u>Conditions to Effectiveness</u>. This Amendment shall become effective upon the satisfaction of each of the following conditions precedent:

(a) the Administrative Agent shall have received a counterpart signature page to this Amendment duly executed and delivered by each of the Borrowers, each Replacement Lender and the Required Lenders;

(b) the Administrative Agent shall be satisfied that the proceeds of the Replacement Term Loan shall be immediately applied to prepay the outstanding balance of the Original Term Loan, and the Borrowers shall pay any breakage fees owed in connection with such prepayment to the exiting Term Loan Lenders; and

(c) the Administrative Agent shall have received evidence reasonably satisfactory to the Administrative Agent that all corporate or other action necessary for the valid execution, delivery and performance by the Borrowers of this Amendment, the borrowing of the Replacement Term Loan and the transactions contemplated hereby shall have been duly and effectively taken.

§9. <u>Representations and Warranties</u>. Each of the Borrowers represents and warrants as follows:

(a) The execution, delivery and performance of each of this Amendment and the Credit Agreement, as amended as of the date hereof and the transactions contemplated hereby and thereby are within the corporate power and authority of such Borrower and have been or will be authorized by proper corporate proceedings, and do not (a) require any consent or approval of the stockholders of such Borrower, (b) contravene any provision of the charter documents or by-laws of such Borrower or any law, rule or regulation applicable to such Borrower, or (c) contravene any provision of, or constitute an event of default or event which, but for the requirement that time elapse or notice be given, or both, would constitute an event

of default under, any other material agreement, instrument or undertaking binding on such Borrower.

(b) This Amendment and the Credit Agreement as amended as of the date hereof and all of the terms and provisions hereof and thereof are the legal, valid and binding obligations of such Borrower enforceable in accordance with their respective terms except as limited by bankruptcy, insolvency, reorganization, moratorium or other laws affecting the enforcement of creditors' rights generally, and except as the remedy of specific performance or of injunctive relief is subject to the discretion of the court before which any proceeding therefor may be brought.

(c) The execution, delivery and performance of this Amendment and the Credit Agreement as amended as of the date hereof and the transactions contemplated hereby and thereby do not require any approval or consent of, or filing or registration with, any governmental or other agency or authority, or any other party.

(d) Each of the representations and warranties of the Borrowers contained in the Credit Agreement or in any document or instrument delivered pursuant to or in connection with the Credit Agreement shall be true as of the date as of which they were made and are true and correct in all material respects as of the date hereof with the same effect as if made on and as of the date hereof (except to the extent of changes resulting from transactions contemplated or permitted by this Credit Agreement and changes occurring in the ordinary course of business which singly or in the aggregate are not materially adverse, and to the extent that such representations and warranties relate expressly to an earlier date).

(e) After giving effect to this Amendment, no Default or Event of Default under the Credit Agreement has occurred and is continuing.

§10. Ratification, etc. Except as expressly amended hereby, the Credit Agreement, the other Loan Documents and all documents, instruments and agreements related thereto are hereby ratified and confirmed in all respects and shall continue in full force and effect. Each Borrower hereby affirms all of its Obligations under the Credit Agreement and under each of the other Loan Documents to which it is a party and hereby affirms its absolute and unconditional promise to pay to the Lenders the Loans and all other amounts due under the Credit Agreement (as amended hereby) and the other Loan Documents. Each Borrower hereby confirms that the Obligations (including those in respect of the Replacement Term Loan) are and remain secured pursuant to the Security Documents and pursuant to all other instruments and documents executed and delivered by such Borrower as security for the Obligations. This Amendment and the Credit Agreement shall hereafter be read and construed together as a single document, and all references in the Credit Agreement or any related agreement to the Credit Agreement shall hereafter refer to the Credit Agreement as amended by this Amendment.

§11. <u>GOVERNING LAW</u>. THIS AMENDMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE COMMONWEALTH OF MASSACHUSETTS.

§12. Delivery By Facsimile Or Other Electronic Transmission. This Amendment, to the extent signed and delivered by means of a facsimile machine or other electronic transmission in which the actual signature is evident, shall be treated in all manner and respects as an original agreement or instrument and shall be considered to have the same binding legal effect as if it were the original signed version thereof delivered in person. At the request of any party hereto, each other party hereto or thereto shall re-execute original forms hereof and deliver them to all other parties. No party hereto shall raise the use of a facsimile machine or other electronic transmission in which the actual signature is evident to deliver a signature or the fact that any signature or agreement or instrument was transmitted or communicated through the use of a facsimile machine or other electronic transmission in which the actual signature is evident as a defense to the formation of a contract and each party forever waives such defense.

§13. <u>Counterparts</u>. This Amendment may be executed in any number of counterparts and by different parties hereto on separate counterparts, each of which when so executed and delivered shall be an original, but all of which counterparts taken together shall be deemed to constitute one and the same instrument.

§14. <u>Miscellaneous</u>. The captions in this Amendment are for convenience of reference only and shall not define or limit the provisions hereof. The Borrowers agree to pay to the Administrative Agent, on demand by the Administrative Agent, all reasonable out-of-pocket costs and expenses incurred or sustained by the Administrative Agent in connection with the preparation of this Amendment, including reasonable legal fees.

[Remainder of page intentionally blank]

FLEET NATIONAL BANK, individually and as Administrative Agent

By: /s/ Maria de Faria e Maia

Name: Maria de Faria e Maia Title: Managing Director

BANK OF AMERICA, N.A., individually and as Syndication Agent

By: <u>/s/ Jonathon M. Phillips</u> Name: Jonathon M. Phillips Title: V.P.

BANK NORTH N.A.

By: <u>/s/ E. Kirke Hart</u> Name: E. Kirk Hart Title: Regional Vice President

CITIZENS BANK

By: /s/ Cindy Chen

Name: Cindy Chen Title: Vice president

[SIGNATURES CONTINUED ON FOLLOWING PAGE]

COMERICA BANK

By: <u>/s/ D. Scott Hagwell</u> Name: D. Scott Hagwell Title: Corporate Banking Officer

LASALLE BANK NATIONAL ASSOCIATION

By: /s/ Brian D. Peterson Name: Brian D. Peterson Title: First Vice President

MERRILL LYNCH CAPITAL, a division of Merrill Lynch Business Financial Services, Inc.

By: /s/ Sheila C. Weimer Name: Sheila C. Weimer Title: Vice President

SENIOR DEBT PORTFOLIO

By: Boston Management and Research as Investment Advisor

By:

EATON VANCE SENIOR INCOME TRUST BY: EATON VANCE MANAGEMENT AS INVESTMENT ADVISOR

By:

EATON VANCE INSTITUTION SENIOR LOAN FUND BY: EATON VANCE MANAGEMENT AS INVESTMENT ADVISOR

By:

OXFORD STRATEGIC INCOME FUND BY: EATON VANCE MANAGEMENT AS INVESTMENT ADVISOR

By:

EATON VANCE CDO III, LTD. BY: EATON VANCE MANAGEMENT AS INVESTMENT ADVISOR

By:

EATON VANCE CDO IV LTD, BY: EATON VANCE MANAGEMENT AS INVESTMENT ADVISOR

By:

COSTANTINUS EATON VANCE CDO V, LTD. BY: EATON VANCE MANAGEMENT AS INVESTMENT ADVISOR

EATON VANCE CDO VI LTD, BY: EATON VANCE MANAGEMENT AS INVESTMENT ADVISOR

By:

GRAYSON & CO

BY: BOSTON MANAGEMENT AND RESEARCH AS INVESTMENT ADVISOR

By:

BIG SKY SENIOR LOAN FUND, LTD. BY: EATON VANCE MANAGEMENT AS INVESTMENT ADVISOR

EATON VANCE VT FLOATING-RATE INCOME FUND BY: EATON VANCE MANAGEMENT AS INVESTMENT ADVISOR

By:

EATON VANCE LIMITED DURATION INCOME FUND BY: EATON VANCE MANAGEMENT AS INVESTMENT ADVISOR

By:

TOLLI & CO. BY: EATON VANCE MANAGEMENT AS INVESTMENT ADVISOR

Carlyle High Yield Partners, L.P.

By:

Carlyle High Yield Partners II, Ltd.

By:

Carlyle High Yield Partners III, Ltd.

By:

Carlyle High Yield Partners IV, Ltd.

By:

Sankary Advisors, LLC as Collateral Manager for Great Point CLO 1999-1 LTD., as Term Lender

By:

/s/ Stacy Braatz Name: Stacy Braatz Title: Secretary

Sankary Advisors, LLC as Collateral Manager for Great Point CLO 1999-1 LTD., as Term Lender

By: /s/ Stacy Braatz

Name: Stacy Braatz Title: Secretary

Sankary Advisors, LLC as Collateral Manager for Great Point CLO 1999-1 LTD., as Term Lender

By: /s/ Stacy Braatz

Name: Stacy Braatz Title: Secretary

Sankary Advisors, LLC as Collateral Manager for Great Point CLO 1999-1 LTD., as Term Lender

By: /s/ Stacy Braatz

Name: Stacy Braatz Title: Secretary

THE TRAVELERS INSURANCE COMPANY

By: /s/ Denise T. Duffee Name: DENISE T. DUFFEE Title: INVESTMENT OFFICER

COLISEUM FUNDING LTD.

By Travelers Asset Management International Company, LLC

By: /s/ Denise T. Duffee Name: DENISE T. DUFFEE Title: INVESTMENT OFFICER

COLUMBUS LOAN FUNDING LTD.

By Travelers Asset Management International Company, LLC

By: /s/ Denise T. Duffee Name: DENISE T. DUFFEE Title: INVESTMENT OFFICER

CITIGROUP INVESTMENTS CORPORATE LOAN FUND INC.

By Travelers Asset Management International Company, LLC

By: <u>/s/ Denise T. Duffee</u> Name: DENISE T. DUFFEE Title: INVESTMENT OFFICER

CITICORP INSURANCE AND INVESTMENT TRUST

By Travelers Asset Management International Company, LLC

By: <u>/s/ Denise T. Duffee</u> Name: DENISE T. DUFFEE Title: INVESTMENT OFFICER

Hanover Square CLO Ltd.

By: Blackstone Debt Advisors L.P. As Collateral Manager

By: /s/ Dean Criares

Name: Dean Criares Title: Managing Director

Fidelity Advisor Series II: Fidelity Advisor Floating Rate High Income Fund (161)

By: /s/ Frank X. Knox Name: Frank X. Knox Title: Assistant Treasurer

BANK ONE, NA

By:

/s/ Clifton Chang Name: Clifton Chang Title: Managing Director

EMERALD ORCHARD LIMITED

By: /s/ Stacey Malek Name: STACEY MALEK Title: ATTORNEY IN FACT

ELT LTD.

By: /s/ Ann E. Morris Name: ANN E. MORRIS Title: AUTHORIZED AGENT

KZH HIGHLAND-2 LLC

By:

/s/ Hi Hua Name: HI HUA Title: AUTHORIZED AGENT

BORROWERS: CASELLA WASTE SYSTEMS, INC. ALL CYCLE WASTE, INC. ALTERNATE ENERGY, INC. ATLANTIC COAST FIBERS, INC. B. AND C. SANITATION CORPORATION BLASDELL DEVELOPMENT GROUP, INC. BRISTOL WASTE MANAGEMENT, INC. CASELLA TRANSPORTATION, INC. CASELLA WASTE MANAGEMENT OF MASSACHUSETTS, INC. CASELLA WASTE MANAGEMENT OF N.Y., INC. CASELLA WASTE MANAGEMENT OF PENNSYLVANIA, INC. CASELLA WASTE MANAGEMENT, INC. DATA DESTRUCTION SERVICES, INC. FAIRFIELD COUNTY RECYCLING, INC. FCR CAMDEN, INC. FCR FLORIDA, INC. FCR GREENSBORO, INC. FCR GREENVILLE, INC. FCR MORRIS, INC. FCR REDEMPTION, INC. FCR TENNESSEE, INC. FCR, INC. FOREST ACQUISITIONS, INC. GRASSLANDS INC. HAKES C & D DISPOSAL, INC. HARDWICK LANDFILL, INC. HIRAM HOLLOW REGENERATION CORP. K-C INTERNATIONAL, LTD. KTI BIO FUELS, INC. KTI ENVIRONMENTAL GROUP, INC. KTI NEW JERSEY FIBERS, INC. KTI OPERATIONS INC. KTI RECYCLING OF NEW ENGLAND, INC. KTI SPECIALTY WASTE SERVICES, INC. KTI, INC.

By: /s/ Richard A. Norris Name: Richard A. Norris Title: Vice President and Treasurer

[SIGNATURES CONTINUED ON FOLLOWING PAGE]

MECKLENBURG COUNTY RECYCLING, INC. NATURAL ENVIRONMENTAL, INC. NEW ENGLAND WASTE SERVICES OF MASSACHUSETTS, INC. NEW ENGLAND WASTE SERVICES OF ME, INC. NEW ENGLAND WASTE SERVICES OF N.Y., INC. NEW ENGLAND WASTE SERVICES OF VERMONT, INC. NEW ENGLAND WASTE SERVICES, INC. NEWBURY WASTE MANAGEMENT, INC. NORTH COUNTRY ENVIRONMENTAL SERVICES, INC. NORTHERN PROPERTIES CORPORATION OF PLATTSBURGH NORTHERN SANITATION, INC. PERC, INC. PINE TREE WASTE, INC. R.A. BRONSON INC. RESOURCE RECOVERY OF CAPE COD, INC. RESOURCE RECOVERY SYSTEMS OF SARASOTA, INC. RESOURCE RECOVERY SYSTEMS, INC. RESOURCE TRANSFER SERVICES, INC. RESOURCE WASTE SYSTEMS, INC. SCHULTZ LANDFILL, INC. SUNDERLAND WASTE MANAGEMENT, INC. U.S. FIBER, INC. WASTE-STREAM INC. WESTFIELD DISPOSAL SERVICE, INC. WINTERS BROTHERS, INC.

 By:
 /s/ Richard A. Norris

 Name:
 Richard A. Norris

 Title:
 Vice President and Treasurer

[SIGNATURES CONTINUED ON FOLLOWING PAGE]

CASELLA NH INVESTORS CO., LLC

By: KTI, Inc., its sole member

By: /s/ Richard A. Norris Name: Richard A. Norris Title: Vice President and Treasurer

CASELLA NH POWER CO., LLC

By: KTI, Inc., its sole member

By: /s/ Richard A. Norris Name: Richard A. Norris Title: Vice President and Treasurer

CASELLA RTG INVESTORS CO., LLC

By: Casella Waste Systems, Inc., its sole member

By: <u>/s/ Richard A. Norris</u> Name: Richard A. Norris Title: Chief Financial Officer and Treasurer

THE HYLAND FACILITY ASSOCIATES

By: <u>/s/ Richard A. Norris</u> Name: Richard A. Norris Title: Duly Authorized Agent

[SIGNATURES CONTINUED ON FOLLOWING PAGE]

MAINE ENERGY RECOVERY COMPANY, LIMITED PARTNERSHIP

By: KTI Environmental Group, Inc., general partner

By: <u>/s/ Richard A. Norris</u> Name: Richard A. Norris Title: Vice President and Treasurer

PERC MANAGEMENT COMPANY LIMITED PARTNERSHIP

By: PERC, Inc., general partner

By: /s/ Richard A. Norris Name: Richard A. Norris Title: Vice President and Treasurer

ROCHESTER ENVIRONMENTAL PARK, LLC

By: /s/ Richard A. Norris Name: Richard A. Norris Title: Duly Authorized Agent

CERTIFICATIONS

I, John W. Casella, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Casella Waste Systems, Inc.;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e) for the registrant and we have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) [Paragraph omitted in accordance with SEC transition instructions contained in SEC Release 34-47986]
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: September 12, 2003

By: /s/ John W. Casella John W. Casella Chief Executive Officer

1

CERTIFICATIONS

I, Richard A. Norris, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Casella Waste Systems, Inc.;
- 2. Based on my knowledge, this quarterly report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e) for the registrant and we have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) [Paragraph omitted in accordance with SEC transition instructions contained in SEC Release 34-47986]
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: September 12, 2003

By: /s/ Richard A. Norris Richard A. Norris Chief Financial Officer

1

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350,

AS ADOPTED PURSUANT TO

SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report on Form 10-Q of Casella Waste Systems, Inc. (the "Company) for the period ended July 31, 2003 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned, John W. Casella, Chief Executive Officer of the Company, hereby certifies, pursuant to 18 U.S.C. Section 1350, that:

1

(1) the Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: September 12, 2003

By: /s/ John W. Casella John W. Casella Chief Executive Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350,

AS ADOPTED PURSUANT TO

SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the quarterly report on Form 10-Q of Casella Waste Systems, Inc. (the "Company") for the period ended July 31, 2003 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), the undersigned, Richard A. Norris, Chief Financial Officer of the Company, hereby certifies, pursuant to 18 U.S.C. Section 1350, that:

1

(1) the Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: September 12, 2003

By: /s/ Richard A. Norris Richard A Norris Chief Financial Officer